Minimizing Construction Disputes

Grace E. Younis

School of Built Environment University of Salford, Salford, UK

Submitted in Partial Fulfilment of the Requirements of the Degree of Doctor of Philosophy, February 2010

Table of Contents

Ac	knowledgement	iii
Lis	st of Figures	v
Lis	st of Tables	vi
Lis	st of Tables	vi
Ab	ostract	vii
Lis	st of Abbreviations	viii
CE	HAPTER 1: INTRODUCTION	1
1.1	1 Introduction	1
1.2	2 Research Question	2
1.3	3 Research Aim	2
1.4	4 Research Objectives	2
1.5	5 Research Methodology	3
1.6	6 Research Contribution	3
1.7	7 Organisation of Thesis	4
CE	IAPTER 2: LITERATURE REVIEW	6
2.1	1. Introduction	6
2.2	2. Disputes	6
2.2.	1. Definition of Dispute	7
2.2.	2. Conflicts as opposed to Disputes	8
2.2.	3. Dispute Causes and Factors	11
2.2.	4. Claims Management	12
2.2.	5. Dispute Avoidance and Resolution	13
2.3	3. Risks	16
2.3.	1. Definition of Risk	17
2.3.2	2. Risk Categories and Risk Types	18
2.3.	3. Trends in Risk Allocation	19
2.3.4	4. Risks in Procurement Practices	19
2.3.3	5. Risk Management	20
2.4	Procurement	21
2.4.	I. Procurement Selection Methods	21
2.4.2	2. Alternative Non-traditional Procurement Methods	23
2.4.3	5. Contractor Selection Discussion	25
2.4	I. The Behavioural Aspect	27
2.4	LADTED 2- DESEADOU METHODOLOGY	35
	IAPTER 5: RESEARCH METHODOLOGY	
3.1	L. Introduction	
3. 2	Leatification of the methodology	
5.5	b. Justification of the methodology	
5.4		40
3.5	Summary	45
CH	IAPTER 4: EXAMINATION OF CONSTRUCTION PROJECTS	46
4.1	. Introduction	46
4.2	Lases Examined	46

4.3.	Data Analysis	53
4.4.	Summary	57
CHAP	TER 5: INTERVIEWS	58
5.1.	Introduction	
5.2.	Interview Questions and Answers	60
5.3.	Analysis of Replies	72
5.4.	Summary	74
СНАР	TER 6: DISPUTE CASE STUDIES	76
6.1.	Introduction	76
6.2.	Analysis Approach	76
6.3.	Dispute Influencing Areas	96
6.4.	Summary	96
CHAP	FER 7: THE FIDIC FORM OF CONTRACT	97
7.1	Introduction	97
7.2	Conditions of Contract	97
7.3	Dispute Influencing Clauses	99
7.4	Summary	108
CHAP	TER 8: CONCLUSION ON DISPUTE INLFUENCING AREAS	109
8.1. D	ispute Influencing Areas	109
8.2.	Analysis of Findings	109
8.3.	Summary	126
CHAP	TER 9: RECOMMENDATION	127
9.1.	Provisional proposals as derived from the Conclusions	127
9.2.	Provisional Proposals	132
9.3.	Recommendations	145
9.4.	Summary	149
CHAP	TER 10: SUMMARY, LIMITATIONS AND FUTURE RESEARCH.	150
10.1.	Summary	150
10.2.	Research Criteria	152
10.3.	Meeting the Aim and Objectives	153
10.4.	Research Limitations	153
10.5.	Future Research	154
Refere	nces	156
Appen	dix A: Chronological Listing of Research on Sources of Dispute (Fenn	l,
2006)	Ling D. Channels right intig of Other Descends on Service of Discus	171
Appendix B: Chronological Listing of Other Research on Sources of Dispute175		
Appendix U: Unronological Listing of Kisk Categorization		
Appendix D: Extracts of the Particular Conditions on the 4 Projects		
Appendix E: A Driel Overview of the Dispute Cases		
Appendix F: Dispute Factors identified in the Dispute Cases		

Acknowledgement

In countless ways I have received love and support from a great number of people to whom I am indebted.

My supervisor Mr. Gerard Wood who was at all times only a phone call away. Through your wise and effective guidance, you made any problem disappear. I would like to thank the Merit Programme for giving me the opportunity to work with such a 'great' supervisor.

The Merit family headed by Dr. Vian Ahmed who has been very kind and prompt in attending to every need and making me feel at home during my visits. The Merit Programme is a success because of you people. Prof. Ghassan Aouad for your follow-up. Your perseverance and success in becoming the Pro-Vice Chancellor and President of the CIOB makes us Lebanese very proud. My panel, Dr. Paul Chynoweth and Prof. Chris Gorse, your comments although considered tough at times have definitely improved the quality of the work.

My local advisor Prof. Asem Abdul-Malak to whom I would express gratitude for being my advisor throughout my academic journey. Your well known meticulousness contributed to the neatness of the data analysis and shaped the breakdown of the work.

My parents for always believing in me. It is your encouragement that made me always endeavor and succeed on the most challenging pursuits. Your continuous unequaled love and support is a blessing. I hope I have given back a fraction of what I have received.

My uncle and Godfather Mr. George Younis, the Chairman of D.G. Jones & Partners, for your support throughout the years. I hope to remain up to your expectations.

My brother Henry for always being there and for the wonderful moments we shared. I feel both lucky and proud to have you as my brother.

My friend Grace for being a great listener and providing me in your calmness with the necessary serenity at all times.

My friend Nada for providing the best breaks in Dubai. Even through a long distance call, the laughs we had when I would be calling to share my distress were truly uplifting.

My friend Maz for the most intriguing and challenging chats. Whether it was construction law issue or a philosophical view, your virtual support made all the difference.

My colleagues Janine, Jizel, Abbas, Jad and Noel for your friendship. Their continuous care and concern was amazing. I will miss working with you guys. Also, the late Mr. Hassib Wehbe who would have been pleased to see this work completed.

My director, Mr. Joe Psaroulla for providing me with the necessary research material and for being more than just a boss. I have to equally thank my uncle, Mr. Sami Younis, the director of the claims department at D.G. Jones & Partners for sharing your experience in disputes and your patience in allocating time and providing the necessary data. Also, I

would like to thank Mr. Talal Ammouni for sharing details of disputes from his project. At this point I would like to thank all 24 interviewees who have generously and patiently allocated time and accepted to have the interviews.

My contractor, Mr. Maher Merehbi, the CEO of A.C.C., for providing research data and resources as well as for the valuable insights as a Contractor that helped shape the recommendations.

Dr. Nael Bunni for generously dedicating time to review my work. Meeting with you in Dubai was one of the highlights in this PHD journey.

Dr. Nayla Comair-Obeid, Mr. Thomas Kernaghan, Mr. William Wharton for offering their expert opinions.

Lastly, my special thanks go to my darling fiancé, Mr. Sarkis Seraydarian, for taking the time to proofread the work. And for your encouragement making the last steps of the journey when I was almost out of strength and patience very enjoyable. You arrived just in time.

List of Figures

Figure 1 Root Causes and Proximate Causes by Kumaraswamy (1997)	9
Figure 2 Concepts and Causal linkage of conflicts and disputes (Kumarasawamy	, 1998) 10
Figure 3 Stages of a Conflict (Robbins, 1994)	
Figure 4 Psychological Cycle of a Conflict (Jong and Seung, 2003)	
Figure 5 People Branch of the Heirarchy (Diekmann and Girard, 1994)	31
Figure 6 Level of Influence of the Factors (Cheung et. al., 2000)	
Figure 7Conceptual Flowchart of Dispute Evolution and Resolution	34
Figure 8 Research Fieldwork Types	40
Figure 9 Case Study Method (Yin, 2003)	
Figure 10 Research Methodology	44
Figure 11 Types of Projects	53
Figure 12 Type of Employer (Public vs. Private)	53
Figure 13 Forms of Contract	54
Figure 14 Dispute Resolution Method	54
Figure 15 Amount of Variations	55
Figure 16 Final Contract Value vs. Original Contract Value	55
Figure 17 Percentage of added variations to original scope of work	56
Figure 18 Final Contract Duration vs. Original Contract Duration	56
Figure 19 Time Overrun (Final Duration/Original Duration)	56
Figure 20 Claim Value vs. Final Settlement	
Figure 21 Dispute Influencing Areas Emerging from Interviews	73
Figure 22 Sample Chronology of Events for Dispute Case	80
Figure 23 Sample Analysis Sheet for Dispute Case	81
Figure 24 Risks Eventuating to Dispute Factors	
Figure 25 Dispute Influencing Areas witnessed in the Dispute Cases	95
Figure 26 Dispute Escalation Influencing Areas	
Figure 27 Summary of Research Methodology	

List of Tables

Table 1 Recommendation for changes in the construction industry analysis (Wood, 2001).	7
Table 2 Shen's (1997) Preventive and Remedial Actions	21
Table 3 The Styles of Handling Interpersonal Conflict (Rahim, 1983)	.29
Table 4 Network of Basic Assumptions Characterizing The Subjective-Objective Debate	
within Social Sciences (Burell and Morgan, 1979; Morgan and Smircich, 1980)	37
Table 5 Elements of the four categories of scientific paradigms (Guba and Lincoln, 1994;	
Perry et al., 1999; Healy and Perry, 2000)	38
Table 6 Spectrum of Risks identified by Bunni (2003)	77
Table 7 Risks identified in Contract by Zack (1996)	78
Table 8 Generic Risks (Bunni's, 2003) vs. Contract Risks (Zack, 1996)	79
Table 9 Risks as categorized by Zack (1996) identified in the dispute cases	84
Table 10 Risk Type Occurence	87
Table 11 Dispute Factors identified in the Dispute Cases	90
Table 12 Dispute Factor Occurence	93

Abstract

The continuing incidence of costly disputes in the construction industry has led to a common interest of researchers in different countries to identify the generic aspects of conflicts, claims, disputes and their resolution. This thesis undertakes an extensive review of literature in the field of construction disputes examining the current understanding of the causes of disputes, as identified by other researchers in the field, and attempts made to minimize them. An analysis of the literature helps identify important themes for particular investigation: procurement methods, risk allocation, claims management and dispute resolution methods.

A preliminary examination of 20 projects in Lebanon confirmed the existence and revealed the extent of disputes on Lebanese projects. Twenty-four semi-structured interviews with practitioners actively involved in construction projects in Lebanon at the project management level are conducted, from which a set of dispute influencing areas emerge. Fifty cases of disputes occurring on four live case study projects in Lebanon are also analysed to examine the risk allocation and occurrence, the behavioural attitudes of key stakeholders, and the factors which lead to disputes between the parties. The findings demonstrate the relationship between those risks which are addressed in the contract and their interaction (when they eventuate) with the behavioural traits of the project participants involved. Furthermore, the dispute factors encountered in these fifty cases are categorized into dispute influencing areas to establish any correlation with the areas raised in the twenty-four interviews. Following comparison of the evidence gained from the literature, the interviews and the case studies, a set of provisional recommendations to minimize disputes is proposed and organized under three themes: a pre-contract award workshop; the drafting of general and particular conditions of contract; and the potential for improvement based behavioural on compliance of project participants. The validity of the provisional recommendations is tested by the reviews of five experts in the field of construction disputes, in accordance with which the recommendations are amended.

List of Abbreviations

Submittal Approved As Noted AAN Amicable Dispute Resolution ADR Analytic Hierarchy Process AHP Submittal Approved As Noted Resubmit ANR Submittal Approved APP Bills of Quantities BOQ Case Based Reasoning CBR Computer Based Training CBT Concurrent Engineering CE Chartered Institute of Arbitrators CIArb Chartered Institute of Builders CIOB Case-Based Procurement Advisory System **CPAS** Cost Plus Incentive Fee **CPIF** Construction Project Success Survey **CPSS** Dispute Adjudication Board DAB Dispute Adjudication Board DAB Dispute Resolution Board DEB Degree of Interaction DOI Dispute Resolution Advisor system DRA Decision Support System DSS **Engineering Contractor** EC Electricite Du Liban EDL Extension of Time) EOT Engineering, Procurement and Construction EPC Engineering, Procurement, Construction, Installation EPCI EPCM Engineering, Procurement and Construction Management Employer Representative ER Estimating using Risk Analysis ERA

FBP	Final and Binding Power
FIR	Field Inspection Request
MOU	Memorandum of Understanding
NA	Submittal Not Approved
NEDO	National Economic Development Office
NPD	Notification of Possible Delay
NPV	Notification of Possible Variation
ОСР	Overall Contractors Performance
PFI	Private Finance Initiative
PM	Project Manager
PMI	Project Management Institute
PPP	Public/Private Partnership
PPSM	Project Procurement System Selection Model
PRAM	Project Risk Analysis and Management technique
RICS	Royal Institute of Chartered Surveyors
RFI	Request for Information

CHAPTER 1: INTRODUCTION

1.1 Introduction

The continuing incidence of costly disputes in the construction industry has led to a common interest amongst researchers in different countries to identify the generic aspects of conflicts, claims, disputes and their resolution (Kumaraswamy, 1998). Construction of new facilities necessitates two parties (hereafter referred to as the Employer and the Contractor) to enter into an agreement. Each party normally expects to receive benefits and perform obligations. The Employer aims to achieve a quality project on time at a fair price. The contractor aims to deliver a quality project on time at a fair price. However, when unanticipated changes are required, often the consequences thereof are not clearly and timeously communicated by each party to the other. As a result, misunderstanding occurs and this leads to a claim and in some cases a dispute (Epling, 1987).

Unlike other types of industries where the development and manufacture of a product can be standardised and tested before being purchased, the nature of projects in the construction industry is extremely diverse. Every project is unique. Even where identical buildings are under construction, the site conditions in each differ and introduce new challenges. Moreover, it is a multi-party process where numerous specialised parties are involved due to the range of skills required. Thus maintaining a teamwork atmosphere and controlling potential conflicts is important. In addition, construction projects normally span for a long period between a decision to invest and completion of works. This leads to instability of supply and demand and a high sensitivity to economic fluctuation (Wood, 2001).

To better understand the causes of disputes in construction industry, a literature review is useful to examine research done in the areas of conflicts, claims, disputes, project success criteria, risks involved in construction projects, procurement techniques and dispute resolution strategies.

1.2 Research Question

Disputes have a direct economic impact on the construction industry. The problem of disputes is international in nature and disputes continue to occur. The literature reveals abundant research pertaining to different aspects of the problem and proposes preventive and remedial measures at the different stages of the construction project. Still the construction industry suffers from cost overruns due to disputes that jeopardize the success of the project procurement and construction. There is a need recognized by several different authors to identify the generic causes of disputes. Questions to be examined are:

- 1. What is the impact of risk allocation in contributing to the incidence of disputes on construction projects?
- 2. What is the significance of project management / contract administration in helping to mitigate claims and minimize construction disputes?
- 3. How does the behavioural attitude of the parties involved in projects affect dispute avoidance, management and/or escalation?

1.3 Research Aim

The aims of this research are to examine the causes of common disputes in the Lebanese construction industry; to identify possible relationships within and between the risk allocation strategies adopted during the procurement of the construction works, contract administration and the behavioural attitude of the parties; and to propose recommendations for improving practice.

1.4 Research Objectives

The research objectives are set as follows:

- Identify and map the interrelated factors causing disputes based on literature and previous research.
- Examine common practices in contract administration and claims management in Lebanon mainly focusing on the procurement trends, forms of Contract used, and risk allocation strategies.

- Gather and analyse data on the nature, incidence and frequency of disputes in the Lebanese construction industry
- Use the processed data to address the importance of sound contract conditions administered by experienced and knowledgeable practitioners and the likely impact on the minimization of disputes.
- Make educative recommendations for academics and practitioners

1.5 Research Methodology

The research methodology used to attain the objectives mentioned above can be summarized as follows:

- Perform an exhaustive literature review in the area of disputes that will include examining areas of risks, conflicts, claims, procurement methods, dispute resolution methods and the impact of behavioural factors.
- Carry out a pilot study on 20 major projects in Lebanon during the recent years to study the claims and the resulting time and cost overruns.
- Conduct semi-structured interviews with 24 participants where questions are raised to discuss different aspects that lead to dispute.
- Examine 50 dispute cases in-depth on 4 projects to identify the risks that have eventuated and the dispute factors that have emerged.
- Use the results from the case studies and the interviews will be used to reveal the common practice in drafting of contracts, the behavioral attitude of practitioners in contract administration along together with their knowledge of the area of dispute prevention and minimization.
- Analyze data collected by way of comparison of findings.
- Propose recommendation for minimizing disputes and gather feedback from 5 experts on their likely success.

1.6 Research Contribution

This study intends to make the following contributions to knowledge:

• It provides data on the common practices in procurement and dispute resolution in Lebanon as well as the nature, incidence and frequency of disputes.

• Results can be used as an educative tool to inform practitioners of specific recommendations for actions which are likely to help minimizing disputes on construction projects.

1.7 Organisation of Thesis

The thesis is divided into 9 chapters, the first of which is the introduction and the last is the summary along with the research limitation and the future research. The remaining 7 are divided as follows:

- Chapter 2 reviews the literature related to dispute causes including the study of risks, conflicts, claims, procurement methods, and behavioural aspects.
- Chapter 3 explores research methodology. It starts with examining how qualitative research developed. The full spectrum from subjectivist to objectivist approaches is presented and the ontology, epistemology and methodology of each of the four scientific paradigms: positivism, critical theory, constructivism and realism are described. The chapter then examines guidelines for case studies, interviews, analysis procedure where selection criteria and limitations of the case study approach are pointed out. Based on those guidelines, the research methodology stages are presented.
- Chapter 4 consists of a preliminary study conducted on 20 construction projects in Lebanon. Analysis of data collected in the preliminary study is undertaken to examine the extent of disputes and the disputed matters.
- Chapter 5 carries out interviews with 24 practitioners in the industry equally divided between Engineers and Contractors to provide an in-depth understanding of the dispute factors.
- Chapter 6 examines 50 dispute cases on 4 projects. Each of the 50 disputes is studied through listing the chronology of events that gave rise to the dispute. It then analyses those events to identify the resulting risks that eventuated, the behavioural attitude of participants and the dispute factors.
- Chapter 7 provides further context through a discussion of the contract clauses (under the FIDIC Red Book 4th Edition) identified in the interviews and case studies.

- Chapter 8 compares the findings/data collected from the literature, the interviews and the case studies in order to draw conclusions for each of the sixteen dispute influencing areas identified.
- Chapter 9 proposes recommendations to reduce the negative impact of disputes based on the conclusions drawn in Chapter 8. The veracity of these recommendations is tested through the critical assessment of three (is it 3 or 5???) expert opinions.
- Chapter 10 presents a summary of the thesis revisiting research criteria and the aims and objectives. It also describes the research limitations and proposes related future research areas.

CHAPTER 2: LITERATURE REVIEW

2.1. Introduction

This chapter examines the literature related to disputes to differentiate between disputes and conflicts, understand the dispute causes and factors and propose remedial and avoidance methods. This has led to the study of the interrelated areas of risk, procurement practices, and the behavioural aspect of participants to understand their impact on dispute emergence.

2.2. Disputes

There have been numerous attempts towards dispute avoidance and minimization and as such there is considerable literature on the subject. In the UK this includes reports through initiatives made by a number of public and private bodies and research carried out by individual authors. From the Simon Report in 1944 to the Latham Report in 1994, several reports included research into construction disputes, as follows:

- The Banwell Report in 1964: The Banwell committee addressed matters with great brevity. It received119 responses to the questionnaire and focused mainly on payment problems and the use of common form contracts.
- The Tavistock Institute in 1966 focused on the impact that human relationships had on the problems that were being encountered and on the importance of the client at the heart of the building process. However, it was theoretical in nature and not followed up by further research.
- The NEDO Report in 1975 considered relationships between the public sector and the construction industry. The contractual concerns examined were: inadequate preparation of documents, use of inappropriate contract forms, excessive variations, underpayment and delay in settling claims. NEDO sent over 300 questionnaires to public sector organisations relating to more than 2,000 contracts and then conducted 50 case studies.
- Building towards 2001 Report in 1989 was produced by the Centre for Strategic Studies in Construction. It was based on discussion groups drawn from the construction industry. The recommendations were generalised focusing on

contractual obligations linking all the parties together and the allocation of specific obligations to each member of the project

- The Latham Report in 1994 was based on the interpretations of the cross section of the industry through a series of discussions and debates. The findings reveal the insufficiency of trust and resources throughout the industry and further sheds light on the fact that many of the industry problems could be solved by either.

A summary of recommendations given in these and other reports by Wood (2001) is shown in Table 1.

	Culture	Process	Organisation
Simon (1944)		Tendering process	
Banwell (1964)		Contractual changes	
Tavistock (1966)	Co-operation	Client centred	Change pattern of
	Trust	Tendering process	relationships
NEDO (1975)		Tendering process	
		Contractual changes	
Building Towards 2001	Co-operation	Client centred	Recognize distinct
(1989)		Contractual changes	phases of concept and
		Research and	delivery
		development	Change organisation
Latham (1994)	Co-operation	Client centred	Integrate design and
	Trust	Contractual changes	construction
	Fair dealing	Tendering process	Partnering
		Research and	
		development	
Atkin & Flanagan	Co-operation	Value management	Partnering
(1995)	Transparency & trust	Better briefing	
	Problem avoidance	Risk analysis	
National Power (1995)	Co-operation	Contractual changes	
Egan (1998)	Co-operation	Client centred	Partnering & supply
	Quality driven	Integrated project	chain management
	Commitment to people	process	Long term perspective
		Research &	Benchmarking and
		development	performance
		Staff development	measurement
			Lean thinking
			Standardisation

Table 1 Recommendation for changes in the construction industry analysis (Wood, 2001)

2.2.1. Definition of Dispute

The definition of dispute is a matter 'in dispute'. Some authors refer to disputes as a simple disagreement, other refer to disputes as the consequence of rejecting a claim (Kumarasawamy, 1997). According to Ren *et al.* (2001) disputes result from the poor resolution of claims. The authors attribute the increased amount of disputes to social,

industrial and project factors. Diekmann and Girard (1995) define disputes in general terms as "any contract question or controversy that must be settled beyond the jobsite management staff".

From a legal point of view conflict is considered to be behavioural whereas disputes are considered to be justiciable. The legality of disputes i.e. whether there is a dispute in the adjudication context, has been treated differently by different schools of thought. Where one party considers that the existence of a dispute is relevant to whether the contract provides for it, another school of thought considers that a claim made and not admitted is sufficient for the crystallization of a dispute. A third party shows reluctance from allowing disputes to be a tool by one party to commence dispute resolution prematurely (Lowe and Leiringer, 2006).

Put in simple terms, the dispute is considered to be as defined in rule 1 of the ICE Arbitration Procedure: 'when a claim or assertion made by one party is rejected by the other party and that rejection is not accepted' (Eggleston, 1993; Kumaraswamy, 1997; Bunni, 2005). Accordingly there has to be a claim, a rejection and a non-acceptance of the rejection. It is not considered to exist on the basis of a claim alone (Bunni, 2005).

2.2.2.Conflicts as opposed to Disputes

Disputes are often precursors of conflicts. As such several studies have aimed at the clarification of terminology between conflicts and disputes. A listing of the research in this area is shown in what follows:

Conflicts are unavoidable in the organizational life, have positive aspects in commercial risk taking, and are resolved by non binding resolution methods. Disputes on the other hand are avoidable, and may be resolved by binding or nonbinding resolution methods (Fenn *et a.l*, 1997; Kumaraswamy, 1997). Kumaraswamy (1997) identifies the common root causes and the proximate causes and confirms the need of further studies to isolate the real root causes of avoidable claims and disputes. A list of the root causes and the proximate causes is shown in Figure 1.

Figure 2 shows an indication of how constructive conflicts can be channelled into improvements while less constructive conflicts may lead to disputes; conflicting

interpretations of contractual documents on instructions could also result in claims

(Kumarasawamy, 1998).



Figure 1 Root Causes and Proximate Causes by Kumaraswamy (1997)

Early conflict theory including that of Follet (1925) viewed conflict as a negative thing that should be avoided through conflict management. The recommended method of conflict management was integration (win-win situation) where each side refocuses its effort so that neither side loses (Lowe and Leiringer, 2006).



Figure 2 Concepts and Causal linkage of conflicts and disputes (Kumarasawamy, 1998)

Disputes are seen to develop when conflicts are not managed. The sooner the destructive conflict is resolved the higher the percentage of resolution success and the lower the cost (Harmon, 2003).

As such there is a common consensus among the authors discussed above that conflicts can be constructive or destructive. Accordingly, constructive conflicts should be encouraged whereas destructive conflicts that lead to disputes should be avoided.

2.2.3.Dispute Causes and Factors

To be able to prevent disputes one should be able to identify/predict the causes and factors. Fenn (2006) conducted an exhaustive study of previous research into or on causes of disputes. A chronological listing of his findings is shown in Appendix A. Similarly a chronological listing of other research on sources of dispute is examined in Appendix B. The listing includes empirical studies conducted by authors in USA, UK, Australia, Canada, Taiwan, Saudi Arabia, Hong Kong, Taiwan, China and Nigeria along with other theoretical studies. The depth and extent of research conducted by those authors differs. The most extensive research is that conducted by Kumaraswamy (1998) and Fenn (1997). It can be noticed that the naming of the categorisation in identifying the problem differs between: areas of dispute, factors in development of disputes, common contributing factors in claims, sources of disputes, major sources of disputes, heads of claims, primary causes of claims. In some cases, similarities in the specified areas are noticed. For example poor communication is identified as a factor in the development of disputes by Rhys Jones (1994); it is also identified as primary cause of claim by Bristow and Vasilopoulous (1995). Also, it is noticed that some like Diekmann et al. (1994) made general categorisation of people, process and project. Others like Watts and Scrivener (1994) defined 290 sources of disputes.

Fenn (1997) concludes that there is a need for research that would investigate the causes of general disputes. Kumaraswamy (1998) again emphasized the need for a deeper analysis of the causal linkage between conflicts, claims and disputes. Identifying common causes and consequences of unresolved conflicts and claims would allow for more effective dispute avoidance as well as more efficient resolution of 'unavoided and unavoidable disputes' (Kumaraswamy, 1998). In spite of abundant research in the area, the continuing emergence of costly disputes corroborates that further studies are needed to identify the causes of these disputes. The following section will examine research made in claims management.

2.2.4.Claims Management

Claims are defined in the Canadian Dictionary as 'an assertion of the right to remedy, relief or property' (Semple *et al.*, 1994; Kumaraswamy, 1997). Claims are raised usually for the assertions for extra money or time 'based on the contract itself, a breach of contract, a breach of some other common law duty, a quasi-contractual assertion for reasonable (quantum merit) compensation, or an *ex-gratia* settlement request'. Due to the designer's explicable inability to provide for all project's eventualities, changes will be made to the project and where they involve additional work this will necessitate an assessment of the time and cost resulting (Harris and Scott, 2001). As such some construction claims are unavoidable and even necessary to contractually accommodate unforeseen changes (Kumaraswamy, 1997).

Semple *et al.* (1994) describe claims as the right of any party to the contract to request for compensation of damages incurred. The authors further suggest the following preventive measures that can help minimize risk and mitigate the causes of claims:

- to allow reasonable time for completing the design, drawings and specifications
- to adopt value engineering and implement constructability
- to set an efficient mechanism for processing and evaluating change orders
- to use critical path method scheduling, cost control and productivity analysis

Claim management is applied at the preconstruction phase through using the standard construction contracts, risk theory and project procurement (Ren *et al.*, 2001).

In an attempt to provide better claim management practices Vidogah and Ndekugri (1998) conclude that there is insufficient emphasis on the importance of claims management practice and related information systems.

Other authors again stressed the need of a structured instrument that would allow monitoring of construction claim process (Kululanga *et al.*, 2001). They proposed the following:

- Claim identification that implies timely and accurate detection of factors that give rise to claims

- Claim notification through informing the other party of the potential increase of time and cost in a non-adversarial manner.
- Claim examination through establishing the legal and factual grounds on which the claim is based.
- Claim documentation through collection of sufficient evidence to defend the case
- Claim presentation through demonstrating the resulting harm by way of legal and factual basis and an estimated recovery
- Claim negotiation through expert skills for proving rights in negotiation
- Use of total quality management to prevent claims

Other authors presented a model developed into an automated decision tool which encompasses the general claim model along with an on-line help that educates the user to the significance of each stage through providing a library of relevant claim cases and court rulings (Abdul-Malak *et al.*, 2002).

The literature examined proposes claim management practices that would help minimise disputes. These vary between applying better contract administration, stressing the significance of behavioural attitudes, and better claims documentation and presentation.

2.2.5.Dispute Avoidance and Resolution

General factors affecting the efficiency in avoiding disputes are set by Henderson (1991) as: clarity of original bid documents, ability of the Contractor to plan and execute the job, recognition by the Employer that changed conditions exist and ability of the owner to respond in a timely manner. Henderson (1991) ascertains that the best way to cope with the risk of disputes is through avoiding them. Proposed preventive measures for avoiding disputes proposed include:

- use value engineering and peer review
- have bid documents checked for constructability, clarity and completeness
- avoid too many or too complex addenda
- evaluate job cost during the design process using a professional estimator
- provide and use adequate CPM scheduling and update requirements

- provide adequate tracing mechanisms for requests for information, substitution requests and change order proposals
- review the A/E's specifications whether they represent your project requirements
- allow a reasonable time for designing the project and for bidding
- require that the contractor's bid documents be placed in escrow
- promote open and factual communication

1

Based on experience the author proposes minimizing disputes through using negotiation as a tool to identify the changes on the job and resolve them at the site personnel level and where necessary by top managers (Henderson, 1991). If resolved through negotiations this results in a change order or a modification. Otherwise it will become a dispute that is resolved through a dispute resolution method such as arbitration, mediation or litigation (Arditi and Patel, 1989).

Betant *et al.* (1995) again proposes general project requirements similar to that of Henderson (1991) that if fulfilled will help in reducing disputes in major projects. Unlike Betant *et al.* (1995) whose work addressed requirements during contract preparation, Henderson (1991) focussed on recommendations applicable in the procurement contract and executions of the works:

- Checking the contractor's tender sum for possible errors or underestimation of certain items that might be ambiguous.
- Drafting clear tender documents minimizing errors and ambiguities preferably using General Conditions of Contract that have sufficient interpretation.
- Clearly identifying Risk areas and discussing the allocation of risk provisions with the Contractor.
- Minimizing on the number of change orders resulting from design changes.
- Where changes are inevitable, these changes shall be discussed with the Contractor and the cost impact of which agreed to before order is issued.
- Interface with other Contractors should be avoided.
- Minimizing disputes through negotiating reasonable claims in good faith.

Spalj (2005) examines dispute avoidance for the Contractor's side based on his experience in the construction industry. He states that disputes become easy to resolve through anticipating, preparing and laying the grounds for resolution before

they occur. The author proposes dispute avoidance measures advised to Contractors at the different stages of construction.

- Before bidding and negotiating the contract, the author proposes investigating subsurface conditions, project site accessibility, weather and keeping records of all investigations.
- Before signing the contract the Contractor is advised to read the contract carefully and pay particular attention to clauses that pertain to: incorporation by reference provisions, flow-down clauses, differing site condition clauses, indemnification clauses, no damage for delay clauses and change clauses.
- During the course of the works the contractor should impose a discipline on the project management team to maintain proper job documentation that would serve as evidence when a dispute arises.

As such the literature examined sheds the light on the importance of preparation of clear contract documents. The contract documents are critical for two main reasons, they define the obligations of each party and they are the documents based on which the Contractor prices. Wherever there is little doubt to the language the Contractor might use that to aggressively interpret that to his benefit more so in competitively bid projects where the contractor was a low bidder. The short period for pricing will further intensify this problem. The Contractor might make hurried assumptions while pricing, and if they are contrary to the designer's intent, the contractor will raise a dispute as the designer's interpretation might cause more time and money (Spittler *et al.*, 1992). In current practice the Contractor is often given a mass of documents that include information and data that may or may not be well coordinated and organized. The Contractors are thus expected to process these documents checking and analyzing the information and the data and raising any discrepancies or missing information in a relatively short period of time where the tender is expected to be 'intelligent but profitable' (Zack, 1996).

Jannadia *et al.* (2000) investigate techniques that can be incorporated in preparing construction contracts and results showed that there was a common desire among the parties to draft dispute resolution clauses provided they are better educated about the importance of 'fair risk allocation'.

Effective project management might be more successful than resorting to claim experts. Jannadia *et al.* (2000) conclude based on previous studies that waiting until the end of the project to resolve disputes makes the procedure more time and cost consuming. Epling (1987) proposes gathering the information that is necessary for assessing the effect of the delay or change while it is fresh. Hence, it is very important to monitor any time or cost overrun through planning, estimating, budgeting and scheduling. This will not stop major disputes from arising but will allow minimizing them (Epling, 1987).

Mitropoulos and Howell (2001) recommend preventing opportunistic behaviour. According to the authors in some cases perception of the other party as being opportunistic is more problematic than opportunistic behaviour itself. Vaaland (2004) on the other hand discusses a process for minimising conflict through enhancing the understanding of the other party's perception, stimulating openness, reducing relational uncertainty, and analysing problematic issues before escalating the tension. Risk allocation strategies, procurement practices and the behavioural aspect of project participants is further examined in the following sections.

2.3. Risks

Two categories of cost identified by Jergeas and Hartman (1996) that are intangible ie add no material value to the project works but result in additional cost are risk premiums that the contractor allows for in bidding and cost for dispute resolution. Therefore there is an interest to reducing these costs early on in the project.

Risk has been the center of attention in the construction industry because of its impact on both time and cost overrun. It is involved in all three phases of the project development cycle: the conceptual phase, the planning and design phase and the procurement and construction phase (Uher and Toakley, 1999; Chan and Kumaraswamy, 1997). The conceptual phase entails the highest degree of uncertainty where it directly impacts the final cost, and where risk management is recognised to have high significance (Uher and Toakley, 1999) This level of uncertainty decreases and is clarified as the project is further developed and detailed (Mak and Picken, 2000).

2.3.1. Definition of Risk

Many attempts have been made at providing a clear definition of the term risk as follows:

- "the possibility that human actions or events lead to consequences that have an impact on what humans value" (Renn, 1998).
- a variable in the process of a construction project whose variation affects the final cost, duration and quality of the project (Bufaied, 1987)
- "the probability of occurrence of some uncertain, unpredictable and even undesirable event(s) that would change the prospects for the profitability on a given investment." (Kartam and Kartam, 2001).
- a combination that occurs when threat and vulnerability overlap (Akintoye and MacLeod, 1997).
- Unforeseen factors that would adversely affect the successful completion of the project in terms of time, cost and quality (Kartam and Kartam, 2001).

Risk is calculated as the probability or frequency of the occurrence of a defined event multiplied by the consequences of the occurrence of that event. Construction projects are sensitive to an extremely large matrix of hazards and risks due to some of the inherent characteristics of the construction project (Bunni, 2005). Risk assessment is the process of defining the components of risk at stake, calculating the probability of (un)wanted consequences and aggregating both components (Renn, 1998). Many attempts have been made to identify the risk associated with construction projects. These are presented in a chronological order in Appendix C.

Changes, claims and litigation are a consequence of the manifestation of risks. This explains the similarity in some cases between causes of risk described in this section and causes of disputes in section 2.1.3. Accordingly proper management of risk will contribute to avoiding potential disputes or at least resolving them more easily (Jergeas and Hartman, 1994). Even when all types of risks are identified, assessing and allocating the risk to the parties as may best suit the project remains a major concern. Previous literature has ascertained that the risk should be transferred to the party that has the competence and expertise for best assessing, managing, controlling and minimizing this risk (Kartam and Kartam, 2001; Pickavance, 2000; Akintoye and MacLeod, 1997). Risk allocation may be achieved through any one or a combination of risk retention, risk transfer, risk reduction and risk avoidance (Akintoye and MacLeod, 1997). During contract procurement, there are conflicting interests between the Owner and the Contractor. The Contractor by nature aims at getting paid as high as possible incurring the least amount of risk possible. The owner on the other hand tends to pay as low as possible while transferring as much risk as possible to the other party. (Pickavance, 2000).

2.3.2.Risk Categories and Risk Types

Bunni (2003) refers to the Grove report that sets four criteria for allocation of risks:

- The fault standard: cost and time impacts of risk caused (or not avoided) through the faults of a party should be borne by that party.
- The foreseeability standard: He who is best able to foresee the risk is allocated that risk.
- The management standard: He who is best able to control and manage the risk is allocated that risk.
- The incentive standard: risks should be placed on that party most in need of incentive (presumably already with the ability) to prevent and control them.

Willingness of parties to bear risk is affected by their general preference for risk/return trade-off, perception of the risk involved, ability to bear the consequences, ability to mitigate and the need to obtain work. Clear perception of the risk involved is a must; otherwise the risk/return trade-off will not apply correctly. As the owner assigns most of the risk to the Contractor, this privilege is associated with an increase in contract price. Experienced owners will find it more cost effective to handle part of this risk. Pricing should be based on identification of the risk categories that can be quantified with sufficient certainty. In the absence of adequate time, contract price is set based on management experience and 'gut feeling' (Ward *et al.*, 1991).

2.3.3. Trends in Risk Allocation

Several studies have been conducted to determine the trends in risk allocation. A survey conducted by Ahmed et al. (1999) in Hong Kong reveals that the Contractor has more readiness to take risk than expected by the Owner.

Employers tend to be risk averse and attempt to transfer at least some of the risks to the Contractor by using disclaimer clauses. By doing so the Employers feel they are limiting their liability if risks that result in cost overrun occur. However, the General Contractor would reduce his exposure by apportioning the risk to the subcontractor who in turn will be adding premiums to cover the same. Hence, it is the Employer who will be paying for the risk premium (Jergeas and Hartman, 1996).

Yeo (1991) notes that there is a tendency to avoid contingency allocation in budget submissions as these contingencies are considered as "fats". As such no allowance is made in anticipation of risk or error in estimation. These "fats" are normally avoided not to raise the project budget too high. But then again this increases the risk of facing a crisis situation during the execution of the work if any of these risks rise. Where the contingency are set correctly in the budget, this will allow for proper comparison at tender analysis and a better allocation for project value.

2.3.4.Risks in Procurement Practices

Jergeas and Hartman (1996) quote the American consulting Engineers Council and the Associated General Contractors of America in the "Owner's Guide to Saving Money by Risk Allocation"

"What threatens the stability and financial security of the construction industry is not design but the problems of distribution of risk inherent in the construction process among the owner, the construction contractor, and the architect and engineer...The industry can not be healthy unless the risk are forthrightly recognized and acknowledged, and the various contracting parties assume under the contract, without ambiguity, their respective parts of the risk."

A comprehensive risk analysis should be completed at contract procurement stage to clearly define a risk allocation strategy. This coupled with sharing of information about risk among different parties builds trust and allows for more efficient construction management. In the cases where there is no transparent comprehensive risk analysis, the parties will not have a clear perception of the level of risk involved. This will have an impact on the contract value. Where the Contractor is knowledgeable of details of the risk involved, the corresponding uncertainty decreases and the Contractor's price for this risk decreases. Also, if the Contractor assumes a risk that was inaccurately judged or underestimated, he/she will try to recover the cost from the other party through claims. Then again, where a risk eventuates and the party bearing the risk realizes that this risk was evident to the other party and was not properly pointed out during contract negotiations an adverse relationship emerges (Ward *et al.*, 1991). Zaghloul and Hartman (2003) stress the relationship between trust and contracting methods and its contribution to effective project management and contract administration. As such, Jergeas and Hartman (1996) proposed adding a new contract clause that makes the Contractor's risk premium visible to the Owner.

2.3.5. Risk Management

Given the significance of risk as described above and its impact on the ability to complete the project successfully, construction practitioners have raised awareness to the importance of risk management (Akintoye and MacLeod, 1997).

Risk management can be defined as "minimizing, controlling, and sharing risks and not merely passing them off onto another party". Risk management can be achieved through retention, transfer, mitigation, and prevention of risk or any combination thereof (Kartam and Kartam, 2001).

Two kinds of management actions defined in the literature by Shen (1997) are preventive actions and remedial actions as shown in Table 2.

Kartam and Kartam (2001) conducted a similar survey in Kuwait. Results showed that Contractors in Kuwait consider judgement and subjective probability using the experience gained from similar project undertaken in the past as very efficient preventive action. Quantitative risk analysis techniques were not considered to be highly effective preventive action for reducing risks. This reflects insufficient knowledge of risk analysis techniques and difficulty of finding probability distribution of risk in practice required in these techniques.

Preventive Actions	Remedial Actions
a. make more accurate time estimation through quantitative risk analysis technique.	a. increase manpower and/or equipment
b. make proper time estimation and produce a proper programme with subjective judgement.	b. change the sequence of work by overlapping activities
c. make proper time estimation and produce a proper programme by referring to previous and ongoing similar projects.	c. provide close supervision to subordinates for minimizing abortive works
d. produce a proper schedule by getting updated project information.	d. increase the working hours
e. plan alternative methods/options as stand-by.	e. change the construction method
f. consciously adjust for bias and add a risk premium to time estimationg. transfer or share risk to/with other parties.	f. coordinate closely with subcontractor.

Table 2 Shen's (1997) Preventive and Remedial Actions

2.4. Procurement

There are different procurement options available such as: sequential traditional, accelerated traditional, design/build, turnkey, management contracting, and construction management (Cheung *et al.*, 2001). Other new procurement strategies replacing or supplementing traditional approach include Concurrent Engineering, Engineering, Procurement and Construction (EPC), Private Finance Initiative (PFI), Public/Private Partnership (PPP), Relational Contracting approaches such as alliancing and partnering (Love *et al.*, 1998; Bing *et al.*, 2005; Mahmoud-Joueini *et al.*, 2004; Palaneeswaran *et al.*, 2003).

Also, there are various methods for valuation of work done with different risk allocating strategies including Lump Sum/Fixed Price, Remeasured, Cost Plus Fixed Fee, Fixed Price Incentive contract and the Cost Plus Incentive Fee (Berends, 2000).

2.4.1.Procurement Selection Methods

Available theoretical models for procurement selection include discriminate analysis approach, multivariate analysis, decision support system, knowledge-based systems, procurement rating systems, procurement path decision charts, the multiattribute approach, the analytical hierarchical process, and the multicriteria/ multiscreening model (Ng. *et al.*, 2002, Luu *et al.*, 2003).

- Studies made to examine the logic of the **multi-attribute utility approach** identified pitfalls in its application to procurement route selection which include: selection of priority variables, possibly inappropriate association of procurement routes with differing coefficients for priority variables due to the assumption of complete contracting; and the insensitivity to project attributes of the utility coefficients used to link routes to outcomes (Chang and Ive, 2002).
- Alhazmi and McCaffer (2000) propose the **project procurement system selection model** (PPSM) which has the potential to assist the client in procurement system selection. It is an integration of Parker's judging alternative technique of value engineering and analytic hierarchy process (AHP) and consists of feasibility ranking evaluation by comparison, weighed evaluation, and AHP.
- The objective-subjective procurement method makes use of the concept of multi-attribute utility technology with the development of utility factors table and the owner's preferences and the characteristics of the project to achieve objective procurement selection. The eight selection criteria for this method are speed, certainty, flexibility, quality level, complexity, risk avoidance, price competition, and point of responsibility (Cheung *et al.*, 2001),
- Due to the lack of consolidated knowledge about some of the specific merits of all potential procurement alternatives, Kumaraswamy and Dissanayaka (2001) identify **potential success criteria**. The study presents observations derived from a pilot exercise to assemble critical modules of the proposed model of the decision support system for optimizing procurement protocols and parallel managerial sub-systems.
- The lack of fuzzy selection criteria is considered to be one possible reason for not using these methods frequently in practice. The **fuzziness degrees of linguistic variables** used as procurement selection criteria is examined through an empirical study conducted in Australia by Ng. *et al.* (2002). The criteria studied are speed complexity, flexibility, responsibility, quality level, risk allocation and price competition (Ng *et al.*, 2002).

Luu *et al.* (2003) again notice that the common procurement selection approaches fail to give an indication of the suitability of the procurement method selected. Their work proposes a case based reasoning approach (CBR) and the case-based procurement advisory system (CPAS) for construction.

There is no reference though of applications of those approaches and their success thereto.

2.4.2.Alternative Non-traditional Procurement Methods

For many years, rigid formal contracts have been adopted in construction. However, the recent changing roles of the parties in today's complex projects have necessitated a new set of contractual arrangements designed to promote more collaborative relationships (Cheung *et al.*, 2006).

2.4.2.1.Design-build

The design-build system has been developed to address the problems of the traditional system and cope with the growth in both the private and public systems (Chan and Yu, 2005). Design-build procurement have evolved over the years with different contractual arrangements: designer-led, builder-led, joint venture, in-house design build (Tenah, 2000).

Fifteen primary project characteristics listed in the importance of ranking in the selection model of the design build system were identified as: well defined scope, shared understanding of scope, adequate owner staffing, owner's construction sophistication, established budget, established completion date, availability of design/builders, willingness to forgo design input, owner's risk aversion, standard design specifications, size of project, technological advances, current state of the market and alternative financing options (Songer and Molenaar, 1997; Molenaar and Songer, 1998).

2.4.2.2.Partnering

Research in the construction industry on partnering has demonstrated the criticality of the owner-contractor relationship. Where there is no trust between the parties, successful project conditions may be jeopardized (Drexler and Larson, 2000).

In spite of being keen on the dedicated involvement in relational contracting, Palaneeswaran *et al.* (2003) warn against shortfalls in the contract and advise maintaining contractual safeguards.

The organizational and cultural factors and economic realities of supply chain relationships are two main barriers to achieving genuine win-win procurement solutions through a qualitative study where interviews were made with senior professionals operating at the policy level.

Facilitative mediation with partnering is proposed as an alternative process that entails holding partnering conference where each of the parties shares his concerns, perception of conflict, and previous experience. At the end of this intervention an agreement is signed by all members to the project team. This intervention will allow for continuous sessions to discuss difficulties that arise (Harmon, 2003).

2.4.2.3.0ther

The literature includes work on other non-traditional procurement methods:

- Concurrent Engineering: Love *et al.* (1998) propose Concurrent
 Engineering (CE) as a holistic approach to the design development and
 procurement of a product. Multi-disciplinary project team is required
 whereby participants are brought together during the design to determine
 how downstream issues may be affected by design decisions. The Project
 manager in this case will be responsible for the initiating managing and
 maintaining coherence during the design development process.
- Incentive/Disincentive Contracts: Arditi and Yasamis (1998) studied a sample of Illinois BOT highways contracts that include I/D provisions to reveal milestones, the way they are executed, and the kind of work practices the contractor uses and the manner in which the contracting parties perceive I/D contracts' effectiveness.
- Engineering Contractor: Berends (2000) investigates the role of the Engineering Contractor (EC) for the Engineering, Procurement and Construction Management (EPCM) of capital investment projects. Where project development is carried out by the EC, the owner may negotiate the

EPCM contract with the same EC or put the work out for tendering. Cost Plus Incentive Fee (CPIF) are proposed by the author to have the best cost risk allocation scenario where the owner bears the cost risk consequences and gives incentive for cost risk management to the EC.

- EPC: Mahmoud-Jouini *et al.* (2003) study the time factor in the case of Engineering, Procurement and Construction (EPC) contracts. Based on an analysis of six projects, different phases in a project including preparation, learning, ongoing and backup, each of which with its own speed representing the global speed profile concept are analyzed. Their work identifies three types of contrasting planned profiles and four speed effective profiles.
- PFI/PPP: Bing *et al.* (2005) examine the risk in procurement and its allocation by the Private Finance Initiative (PFI) in the UK which is a form of Public/Private Partnership (PPP) Procurement. A three-level meta-classification is proposed, where risk is classified as macro (origins beyond the system boundaries of projects), meso (directly concerned with the nature of each project) and micro (associated with the relationships between the parties involved).
- Relational contracts: Cheung *et al.* (2006) examined the different types of contracts and classified them as traditional and relational. They examine the application of relational contracts by establishing a long term relationship of communication and trust to minimise adversarial tendencies. This study suggests that the main contract and the domestic subcontract are more relational than the nominated subcontract and the direct labour contract

2.4.3.Contractor Selection Discussion

The correct choice of the Contractor is a critical decision that will affect the expected performance on the project to be procured. Contractor selection is commonly done in two stages: eligible bidders are short listed through a prequalification model and then the best bid is selected among those bidders (Kumaraswamy, 1996).

Palaneeswaram and Kumaraswamy (2001) studied general contractor selection practices for various public clients in different countries including Hong Kong, USA, Australia, Sri Lanka, Singapore and Canada. The authors accordingly developed a model for contractor prequalification. The model provides a structured framework that can be used as a base for developing an intelligent client advisory decision support system that can be customized for different projects (Kumaraswamy, 1996). As for the bid selection, the lump-sum lowest bid method is commonly used especially in public project that awards the contract to the lowest bid price. Crowley and Hancher (1995) explain that since 30 years ago there has been a conflict of views between policy-makers and procurement practitioners. Policy-makers see competitive procurements as a very effective and efficient method that serves the public's interest. Procurement practitioners on the other hand find competitive procurement to be risky, as it exposes the Employer to accepting the wrong bid or accepting the wrong firm. Accepting the low bid that has been mistakenly been underestimated might lead to the "winner's curse". As such an error to the bid this would make the award unfair to the winning bidder, to the other bidders that might have made a more accurate estimate and to the Employer that will have to struggle to contain the running cost of the underfunded project. With the wrong firm the Employer is exposed to a firm that might study the bid closely and look for mistakes, ambiguities and possible change order and claims that the bidder can use after being awarded the contract to recapture monies and offset their original low bid.

To overcome this problem, selection of the 'average bid' was proposed as an alternative in Italy and Taiwan. In Singapore, tendering price advantage is assigned according to the "Construction Quality Assessment System" where a premium list is developed by the Construction Industry Development Board (Kumaraswamy, 1996). From1984, the US Congress recognized the need for improved procurement procedures, the Best-Value procurement which selects the contractor with the offer most advantageous was devised where the price along with other factors such as technical and managerial merit, financial health and past performance are considered in evaluating offers. Disadvantages of this method include: added more time and effort needed in preparing the bid, evaluation process becoming complicated, and increased danger of bid protests (Gransberg and Ellicot, 1997). Wang *et al.* (2005) present an electronically

26
facilitated unit-price-based model for evaluating competitive bids through examining the lowest bid based on the unit prices of cost items of the project.

Wang and Yang (2005) have devised an electronically facilitated bidding model. The new method proposed examines the bidder's quantities and unit rates submitted and where found unreasonable adjusts them to be set as the contract quantities and unit rates. Also, the Contractor is requested to specify three equals with different trademarks for the significant products within a significant period after the contract is awarded.

The new management strategies are surfacing to replace or supplement traditional approaches but when these are not implemented this negatively affects the client's objectives. The dominant building procurement system in many parts of the world is still the typical/pure traditional building procurement system and the main cost control mechanism is the bill of quantities which is prepared from incomplete design because of time constraints (Rwelamila and Meyer, 1999). This is particularly true of the Middle East construction industry including Lebanon.

2.4. The Behavioural Aspect

The last factor examined in this chapter is the behavioural aspect among project participants. Maintaining a cooperative environment becomes a difficult task because conflicts are inherent in construction projects (Zack, 1995; Fenn, 1997). The conflict can be described as the progression of four related stages as shown in Figure 3 below (Robbins, 1994):

- Antecedent conditions: The first step in a conflict incidence is the presence of conditions that allows for conflict to occur.
- **Cognition and personalization of conflict:** If these antecedent conditions are present, they generate frustration and hence, conflict. The issue of conflict could be one of the following: scarce resources, collective procedures, policies or action, and role behaviour of individuals.

Behaviour manifestation: This is the point where the conflict is out in the open. Expression of the conflict can be subtle indirect and highly controlled or can take the form of aggressive, violent and uncontrollable struggle. The five principle interpersonal conflict handling behaviours are described in Table 3 below. Aftermath of conflict: More often, destructive outcomes of conflict are recognized over the potential benefits. These outcomes include physical or psychological injury, increased hostility and misperception, hardened antagonistic position and emotional exhaustion. Among the potentially positive outcomes are the development of a sense of solidarity among members of groups engaged in conflict; the emergence of creative ideas; the formulation of new policies, procedures and services, reformation and renewal of programs; and heightened enthusiasm and purpose among the conflicting parties.



Figure 3 Stages of a Conflict (Robbins, 1994)

Rahim (1983) based his study on the conceptual scheme first presented by Blake and Mouton (1964) to classify the modes for handling interpersonal conflicts into five types: problem-solving, smoothing, forcing, withdrawal, and sharing. He differentiated styles of handling interpersonal conflict on two basic dimensions, concern for self and concern for others as shown in Table 3.

- **Integrating (collaborating)**: This style is characterized by confrontation and problem solving where the open and direct communication allows for the problem solving.

Concern for Self High Low Concern for Others Low Dominating Compromising Avoiding

Table 3 The Styles of Handling Interpersonal Conflict (Rahim, 1983)

- **Dominating (competing\forcing)**: the individual with high concern of self and low concern for the other party is identified with a win-lose orientation forcing behaviour to win one's position
- Avoiding (withdrawal): the unconcerned attitude toward the self, issues or parties results in postponing an issue and withdrawing from a threatening situation.
- **Compromising (negotiating)**: this is the mixed motive style where both parties give up something to make a mutually acceptable decision splitting the difference or seeking other quick middle-ground positions (Rahim *et al.*, 2000).

Conflict styles are learned during childhood and are reinforced and modified as we experience conflict during our lives. The tendency is to use our learned style of conflict behaviour in stressful situations. In less-stressful situations, people have the ability to move between the styles. Movement between the styles can be helpful because each style has its place. For example, in a minor conflict where there will be no continuing relationship, the best course of action may be to minimize one's investment in the conflict by accommodating the desires of the other. For more meaningful conflicts, however, it is generally realized that outcomes produced by collaboration or compromise are superior to those produced by other methods. Dealing with conflict by arguing, fighting, appealing to the courts or other party, or voting are win-lose methods of resolving conflict (Brandt and Murphy, 2000).



Figure 4 Psychological Cycle of a Conflict (Jong and Seung, 2003)

The psychological cycle shown in Figure 4 includes three stages: (1) the conflict escalation, (2) climax is reached and a stalemate continues, and (3) de-escalation. During the conflict escalation, four transformations are identified: conflict issues increase, changes from criticism of behaviour to a focus on personalities, use of stronger tactics to win, and more people engaged in the conflict. When the conflict reaches the peak i.e. the stalemate, the parties start reconsidering collaboration as a way to get what they want. During the de-escalation stage the parties move towards a settlement. The critical issue in this case is that the conflict residues that remain describe ill feelings in the conflicting parties' minds. The only method to avoid these conflict residues that might result in more desperate conflicts later on is through resolving the conflict by addressing mutual interests and relationships.

Of the three factors technical, contractual and behavioural on developments of disputes, people criteria (opportunistic behaviour) have proved in more than one study to have the most effect (Mitropoulos and Howell, 2001; Molenaar *et al.*, 2000). Diekmann and Girard (1995) studied the people, project and process criteria effect on emanating disputes. The results showed that the people criteria had the most effect followed by the

process criteria. People are the foundation of every construction project: they must deal with ever changing conditions, must manage the



Figure 5 People Branch of the Heirarchy (Diekmann and Girard, 1994)

process, and most importantly must negotiate and deal with disagreements and disputes that are bound to arise. As shown in Figure 5 above, people criteria are arranged into three branches: the owner, the contractor and the business relationship that exists between the two.

The fact that team building can be used as a project management tool to reduce the occurrence of these conflicts was examined for each case as follows (Gardiner and Simmons, 1998):

- Conflict due to task interdependency: The construction industry calls for high level of interdependence among the different parties involved on a project

including clients, users, designers and contractors, where hundreds of tasks have to be undertaken and integrated to achieve the finished product. Accordingly, interorganisational team building and building of trust can reduce the conflict that may result if the cooperative spirit is not maintained.

- Conflict due to differentiation: Construction projects are traditionally environments where there is differentiation between different teams that have been brought together for the purpose of completing the project. Given the fact that organizational differentiation is a source of conflict, the effort made at the beginning of the project to bring the different teams together to understand and become familiar with the other participants' perception is important.
- Conflict due to differing values, interests and objectives: Since the different teams working in the project might have conflicting goals, it is essential for the project manager to set a shared common goal that is in the best interest of the project.
- Conflict due to communication obstacle: Due to the tight schedule, this might result in less collaboration and unreasonable demands as each party is unaware of the requirements of the other party. This again necessitates team building.
- Conflict due to tension: Anxieties may result from inconsistent demands among different parties. It is therefore essential to provide for social interactions that help alleviate theses anxieties.
- Conflict due to personality traits: As there is evidence that relationships involve mixed motives. Managing these motives necessitates high behavioural flexibility among project team members. This can only be achieved through formal human resource management to select project team members. Objectives and qualities of team members are examined in the following section.

The study emphasizes the significance of team building and partnering. Organizational development at the beginning of the project has proved to achieve significant improvement in a relatively short period.

As there are constructive and destructive conflict, efforts should be made because the sooner the destructive conflict is resolved, the higher the percentage of resolution success and the lower the cost (Harmon, 2003). One way of conflict minimization is through enhancing the understanding of the other party's perception, stimulating openness, reducing relational uncertainty, and analyzing problematic issues before escalating the tension (Vaaland, 2004).

The three level influence diagram by Cheung *et al.* (2000) is shown in Figure 6. The factors (substantive influences) in the inner circle are mainly design changes, involvement of claim advisor, incentive to settle and project personnel relationships have the most influence on dispute resolution. Factor two (facilitative influences) comprise mainly contractual use of ADR and involvement of senior management level promotes dispute resolution. Factor three (indirect influences) in the outermost layer mainly claim consciousness of the contractor and change in tender price index cause lingering of disputes. It resembles a three level influencing diagram where the innermost circle has the most critical influence in the outcome of dispute resolution process.



Figure 6 Level of Influence of the Factors (Cheung et. al., 2000)



Figure 7Conceptual Flowchart of Dispute Evolution and Resolution

According to Zack (1995), if the qualified people are assigned to both parties to a contract, they will begin to know, understand, respect and trust each other. The work experience will build solid relationship and thus the effectiveness in negotiating settlements will increase and the time spent negotiating settlements will decrease.

2.4. Summary

The literature reveals abundant research studying the interrelationship between different factors that influence the emergence of disputes and proposes preventive and remedial measures at the different stages of the construction project to minimize them. A study of disputes has led to the study of risks, conflicts, claims, and procurement methods. Figure 7 offers a conceptual flowchart that describes the trajectory of dispute evolution and resolution based on the literature review. The literature proves efforts carried out in different countries addressing categorization of risks and disputes, risk allocation, claim management, preventive and remedial measures that could help reduce eventuation of risks and the emergence of disputes, as well as the behavioral traits of the project participants. However, the construction industry continues to suffer from cost overruns due to disputes and there remains a need, recognized by many authors, to identify the generic causes of disputes.

It is worth noting that no research was found addressing disputes, risks, procurement practices and behavioral aspects of participants in the Lebanese context. The following chapters will try to examine the nature and causes of disputes in the Lebanese construction industry and to identify possible relationships within and between the risk allocation strategies adopted during the procurement stage and the behavioral attitude of the parties. This will be carried out based on the methodology described in Chapter 3 below.

CHAPTER 3: RESEARCH METHODOLOGY

3.1. Introduction

Scientific research methods began to take form in the eighteenth century. The earliest systematic discussions were overwhelmingly focused on the experimental method in science. This scientific methodology reflected an optimistic belief in "a determinate non-contradictory, self identical, and coherent world" that exists independently of the researcher's perception (Pollnet, 1987; Weinberg, 2002). Until the 1960s and the 1970s research was influenced by the abstracted empiricism based on the use of quantitative methods (Morgan and Smircich, 1980). After the 1980s, qualitative research has become one of the big growth areas (Morgan and Smircich, 1980; Travers, 2001).

This chapter describes the research methodology adopted. It outlines the steps carried out in the following chapters in terms of fieldwork, analysis of dated, and deriving a conclusion and a recommendation

3.2. Scientific Paradigms

Distinction between qualitative and quantitative approaches is at best approximated for both types as umbrella categories that cover many different actual methods (Gummesson, 2005; Long *et al.*, 2000; Wilson and Natale, 2001; Hanson and Grimmer, 2007). This can be better understood by looking at the full spectrum from subjectivist to objectivist approaches used in the contemporary social sciences by Burell and Morgan (1979) shown in Table 4 (Morgan and Smircich, 1980).The adequacy of the methodology whether qualitative or quantitative depends on the nature of knowledge under study and the method through which that knowledge can be obtained (Gummesson, 2000).

Two major approaches to theory development are the deductive theory testing and the inductive theory building (Bonoma, 1985; Parkhe, 1993; Romano, 1989; Perry, 1998). The difference between them is defined by the scientific paradigms used. A scientific paradigm is defined by three elements: ontology, epistemology and methodology where ontology is the "reality" that researchers investigate, epistemology is the relationship between the realities and the researcher and methodology is the technique used by the researcher to investigate that reality (Guba and Lincoln, 1994; Perry *et al.*,

	Subjectivist Approaches to Social Sciences					Objectivist Approaches to Social Sciences
Core Ontological assumptions	reality as a projection of human imagination	reality as a social construction	reality as a realm of symbolic discourse	reality as a contextual field of information	reality as a concrete process	reality as a concrete structure
Assumptions About Human Nature	man as a pure spirit, consciousness, being	man as a social constructor; the symbol creator	man as an actor; the symbol user	man as an information processor	man as an adaptor	man as a responder
Basic Epistemological stance	To obtain phenomenological insight, revelation	to understand how social reality is created	to understand pattern of symbolic discourse	to map contexts	to study systems, process, change	to construct a positivist sciences

Table 4 Network of Basic Assumptions Characterizing The Subjective-Objective Debate within Social Sciences (Burell and Morgan, 1979; Morgan and Smircich, 1980)

1997; Healy and Perry, 2000). The deductive approach represents the positivist paradigm and the inductive approach represents the phenomological paradigm (Easterby-Smith *et al.*, 1991; Perry, 1998). Phenomological paradigm can be further broken down into three: critical theory, constructivism and realism (Guba and Lincoln, 1994). Going back to the continuum in Table 4, hard positivism is at one end and constructivism is at the other (Carson *et al.*, 2001; Jean Lee, 1992; Healy and Perry, 2000; Kidd, 2002; Guba and Lincoln, 2000).

A brief overview of each of the four paradigms shown in Table 5 is included in what follows:

Positivism is based on the assumption that it is possible to describe the world objectively from a vantage point (Travers, 2001) ie data does not change because it is being observed (Guba and Lincoln, 1994; Healy and Perry, 2000). It is based on quantitative research (Healy and Perry, 2000).

Critical Theory is one of the qualitative research options available. It emphasizes social realities. Critical theory researches critique and transform social, political, cultural, economic, ethnic, and gender values (Guba and Lincoln, 1994; Healy and Perry, 2000).

Constructivism has relativist ontology and considers that each person has his or her own reality. At the epistemological level, the achievement of objectivity is rejected;

	Positivism	Critical theory	Constructivism	Realism
Ontology	Reality is real and apprehensible	Virtual reality shaped by economic, ethnic, political, cultural and gender values crystallised over time	Multiple local and specific "constructed" realities.	Reality is "real" but only imperfectly and probabilistically apprehensible
Epistemology	Objectivist: findings true	Subjectivist: value findings	Subjectivist: created findings	Modified objectivist: findings probably true
Common Methodologies	Experimental/ surveys: verification hypotheses, quantitative methods	Dialogic/dialectic al: researcher is "transformative intellectual" changes the social world which participants live	A Hermeneutical/ dialectical: wholesearcher is a "passionate withinparticipant" within the world being investigated	Case studies/convergent interviewing: triangulation, interpretation of research issues by qualitative and by some quantitative methods such as structural equation modelling

Table 5 Elements of the four categories of scientific paradigms (Guba and Lincoln, 1994; Perry etal.., 1999; Healy and Perry, 2000)

instead individuals are expected to understand particular viewpoints (Morgan and Smircich, 1980; Hanson and Grimmer, 2007).

Realism looks behind appearances to discover laws or mechanisms. Realistic researchers believe that the "real" world exists but it is only "imperfectly apprehensible" (Guba and Lincoln, 1994; Healy and Perry, 2000).

Other scientific paradigms include **interpretivism** and **poststructuralism**. For interpritivists there is no benefit with working with large data sets. They rather use very short decontextualized extracts from interviewees. Poststruralism on the other hand is a radical philosophical movement that seeks to challenge the assumption that it is possible to obtain valid knowledge about the world (Travers, 2001).

3.3. Justification of the methodology

Realism has been adopted as the most suitable paradigm for the purpose of this thesis. Realism admits that there is an external reality. However, observation of that

reality is achieved through the limitations of the researcher's mental capacity (Tsoukas, 1989; Perry, 1998).

Six criteria presented by Healy and Perry (1998) that further confirm the appropriateness of realism methodology for the purpose of this research are:

- Ontological appropriateness: The research deals with complex social phenomena involving reflective people.
- Open fuzzy boundary systems: Social phenomena are fragile and the causal impacts are dependent on their environment. The research aims at developing a "family of answers" to cover several contingent possibilities (Pawson and Tiley, 1997; Healy and Perry, 1998).
- Epistemology: realists are value aware. They are neither value laden like positivists nor value free like constructivists (Guba and Lincoln, 1994; Manning, 1997; Healy and Perry, 1998). Realism relies on multiple perceptions which is achieved through triangulation of several data sources and the researchers interpretation of those triangulations (Healy and Perry, 1998).
- Methodological trustworthiness: Methodological trustworthiness in realism is similar to reliability in the positivism. It is the extent through which the researcher can be audited by developing a case study database and the use of quotations in the report (Lincoln and Guba, 1985; Healy and Perry, 1998).
- Analytic generalization: According to realists, theory building has to be built and confirmed or disconfirmed, before its generalisability to a population is tested (Healy and Perry, 2000).
- Construct validity: It is an assessment of the appropriateness of measures (Healy and Perry, 2000).

Although realism is considered to be more of an inductive rather than a deductive approach, case study research includes deduction based on prior theory. Formulation of the research problem is influenced by the literature review or the researcher's preconceptions; hence starting from scratch with a theoretical hypothesis is neither practical nor preferred (Perry, 1998).

3.4.Description of Methodology

The methodology adopted for this research work consists of the following stages:

- A thorough literature review is conducted for previous studies of disputes. This has led to examining areas of risks, procurement practices, and the behavioural aspect of project participants. The literature review revealed interrelationship of the four areas that was shown in Figure 8. There was abundant examination by several researchers on dispute factors and risk factors. However, no research on this subject was found to be conducted in Lebanon.
- Field work was carried out at three different fronts as shown in Figure 8:
 - Preliminary Examination of Projects: 20 projects are examined to draw an overview of the dispute conditions in facts and figures in the Lebanese construction context. Data related to 20 different projects is studied through a general examination of the scale of claim, time and cost overrun, the causes of disputes and the dispute resolution methods employed. This would give an indication of the dispute causes and extent in the Lebanese context.



Figure 8 Research Fieldwork Types

• Interviews: The literature examined and the observations made in the 20 projects instigated questions that were raised in 24 interviews conducted with senior practitioners acting as Project Managers. This provided

feedback for practitioners on the causes of disputes and the common practice in handling disputes.

The interviews were conducted in the following effective interview spirit proposed by Gerson and Horowitz (2003):

- To guide respondents through the main theoretical concerns in an orderly fashion within a limited period of time.
- To have a theoretically informed and user-friendly interview schedule or an effective interview guide.
- To collect information in a manageable form for later analysis.
- To set the direction of the interview for the critical factors/outcomes that the interview needs to explore.
- To allow for "discovering the unexpected and uncovering the unknown."

The interviews provided a deeper insight of the practitioners views on areas causing disputes. It also, gives an overview of the level of knowledge in contracts and in project management.

Guidelines to interviewing presented by experienced researchers state that interviews depend on the feedback received from the interviewees commitment. Convincing others to dedicate time and contribute to a project requires a strong belief of the interviewer in the value of the study and a persistent approach. Conducting the interview itself requires analytical skills and intense concentration. The best interviews become a conversation between two parties trying to examine the variables and their effects on the research problem. The interviewer should at all times listen carefully and supportively and refrain from drawing judgemental conclusions (Gerson and Horowitz, 2003).

 Dispute Cases: The interviews were followed up by scrutinizing 4 ongoing projects where 50 disputes cases were investigated. This allowed for an in-depth objective examination of the causes of disputes and the common practice in the construction industry beyond any subjectivity that might have been introduced by the interviewees above.

Case study is a research strategy that examines the dynamics present in a particular setting. Analyzing data is the most difficult and the least described stage. Within case analysis is the first stage of data analysis and it is very important in examining the particulars of each case to become familiar with its characteristics. Yin (2003) describes the stages of developing a theory from case studies in Figure 9. The researcher should stop adding cases when theoretical saturation is reached ie where incremental learning is minimal. However, in practice pragmatic constraints such as time and budget limit the number of cases examined (Eisenhardt, 1989).

The multiple case study or the collective case study is adopted for the purpose of this study (Denzin and Lincoln, 1994). The dispute cases are described in Chapter 6.



Figure 9 Case Study Method (Yin, 2003)

It should be noted that the information richness of the cases and the observational/analytic capabilities of the researcher are of more significance to the validity and meaningfulness of the case studies than the sample size (Patton, 2002). Patton (2002) lists 15 strategies of "purposeful sampling" of which mixed purposeful sampling was adopted. Also, the sample was chosen with little bias in a way to bestow confidence in the findings reflecting larger trends rather than idiosyncrasies of restricted groups as recommended by Gerson and Horowitz (2003). A case study database was created where the data collected, including the case study notes and the case study documents, was stored in a dedicated area to allow for retrieval of the necessary information efficiently along the different stages.

- Conclusion: The dispute influencing areas are examined in light of the findings in each of the literature review, the interviews, and the case studies. The three sources are compared to reach a common consensus regarding the impact/influence of each of the dispute influencing areas. The use of multiple sources of evidence allows for converging lines of inquiry which reinforces the construct validity. This offsets the additional time and resources needed in providing these multiple sources. Efforts were made to maintain a chain of evidence to describe the basis of the conclusions reached. These two practices are important in preserving the reliability of the research conducted (Yin, 2003).
- **Provisional Recommendation:** The conclusion is used to formulate a provisional recommendation for minimizing the incidence of construction disputes. This provisional recommendation is further examined through feedback received from five expert opinions. The five experts are prominent members of chartered institutes such as the RICS, EIOB, and CIArb who have practices contract administrators in Europe and with current experience in Lebanon/the Middle East.



Figure 10 Research Methodology

It should be noted that the four tests to judge the quality of research design are as follows:

- 1. Construct validity: where the operational methods such as the sources of evidence and data collection are studied.
- 2. Internal validity: establishing causal relationships in data analysis by way of pattern-matching, explanation building, addressing rival explanations, and using logic models.
- 3. External validity: establish the limits within which the case study can be generalized.
- 4. Reliability: demonstrate that the case study operation can be repeated by using case study protocol and developing case study databases.

These four criteria will be adopted as the benchmark to check the quality of this research work. As such compliance with those four tests will be revisited in Chapter 10.

3.5.Summary

This Chapter starts with examining how qualitative research developed. The full spectrum from subjectivist to objectivist approach is presented along with the ontology, epistemology and methodology of each of the four scientific paradigms. The choice of the realism approach for this research is explained. The research methodology stages are then presented. It entails 3 fieldworks studies examining 20 projects to explore the nature and extent of disputes in Lebanon, interviewing 24 practitioners in the industry to discuss their experience in causes of disputes and scrutinizing 50 case studies to track the factors influencing disputes.

Findings from 3 sources (literature, interview, case study) are compared and recommendation is formulated accordingly. This recommendation is further refined based on comments received from 5 expert opinions on the veracity of the proposed recommendation.

CHAPTER 4: EXAMINATION OF CONSTRUCTION PROJECTS

4.1.Introduction

Since no research was reported in the literature to address disputes on construction projects. It was necessary to examine a sample of projects to substantiate the need for a study on dispute minimization in the Lebanese context.

A study of 20 projects in the Lebanese construction industry was undertaken to form an overview on causes of disputes along with time and cost overrun.

The 20 cases to be examined can be categorised as a combination purposeful sampling method described by Patton (2002) where two different sampling methods are applied by way of triangulation. These two methods are the:

- Maximum variation sampling where the cases are heterogeneous representing different parties to the contract and different procurement practices.

- Criterion sampling where all the cases examined are Lebanese construction projects.

The basic data gathered in these case studies consists of: project type, original contract value, value of variations, claimed value, final settlement value, original contract duration, final contract duration, type of contract, form of contract, dispute settlement procedure.

4.2.Cases Examined

Claims that have been submitted on these projects have been examined to identify the underlying causes. A short description of these claims of these projects is described below:

<u>Case 1</u>

The contract was signed for the execution of a private hotel. During the execution the Contractor claimed to have suffered from the following: "lack of complete or fully coordinated design information, delay in issuing drawings, direction, clarifications, instructions, order or approvals, the extent and timing of instruction variations, suspension of the works, delay in nomination of specialist subcontractors, delay in consenting to sub-contracting parts of the works, invalid instructions, and failure of the Engineer to correctly value the works and certify payments." The Contractor claimed accordingly and requested an extension of time and compensation. The Contractor considering the Engineer's determination unfair requested an Engineer's Decision. The Contractor then gave notice for arbitration where he found that the Engineer's Decision was not rightful. A meeting was held after completion of the works and an amicable settlement was reached.

<u>Case 2</u>

The contract was signed for the execution of a diaphragm wall for a private building complex. During execution of the works an addendum was signed adding the mat foundation to the scope. However, the **permit was delayed** due to the political situation resulting in a suspension of two years. The Contractor claimed for extension of time and compensation. Upon resumption of the works another addendum was signed and an amicable settlement was reached regarding the extension and the compensation. During the execution of the works, the Contractor submitted several claims for **new regulations** for dumping material that resulted in extra cost, **escalation of prices**, and an **error in design calculations**. The Engineer agreed in principle to these claims. In the case of price escalation although the contract did not allow for it the Engineer presented the Contractor's case to the Employer proposing to compensate for the unexpected cost incurred. The Employer accepted this proposal. An assessment for these claims was prepared and presented to the Employer. The Contractor did not approve to the Engineer's assessments and all claims were put in one basket where an amicable settlement was reached at a higher management meeting.

<u>Case 3</u>

The contract was signed for the execution of a private residential building. Several design modifications were requested during the execution of the works that delayed the works. However, the Contractor also suffered delay due to failure of concrete strength tests. The Contractor submitted a claim for extension of time and compensation due to **design modifications**. The Contractor also submitted a claim for **escalation in steel rates** although the contract did not allow for price escalation. The Engineer upon receiving the

Employer's approval certified payment for 50% of the price adjustment claim for escalation of steel rates. An assessment of the EOT and compensation was made. The Contractor disputed the same. However, an amicable settlement was reached after completion of the works at a higher management meeting.

<u>Case 4</u>

The contract was signed for public highway works that included a bridge. The Contractor claimed for compensation due to **increase in custom duties**, **price escalation of fuel**, **crushed material and cement**, **additional cost due to new design requirements**, **design modifications**. He also claimed compensation for an extension of time that was previously granted conditional to waiver of compensation. The Contractor had signed conditional waiver on receiving this extension. These claims were rejected by the Engineer. The Contractor proceeded with arbitration as the claim value constituted 42% of the original contract value.

Case 5

An agreement was signed for the execution of a major public facility. The Contractor was granted an extension of time due to the **nonavailability of work areas**. The Contractor claimed for incurred cost during the delayed period due to site overhead, inability to execute other projects during this period, interest rates, extended warranties and head office overhead. The case was resolved through an amicable settlement.

Case 6

The Contract was signed for the execution of a public university. The Contractor was granted an extension of time due to **delayed possession of the site**. The Contractor claimed for **escalation of prices of quarry material and cost of steel** during the extension. This claim is expected to be resolved through an amicable settlement. However, to date no compensation has yet been made.

Case 7

An agreement was signed for a public university building where the Contractor suffered from delay due to the following: late handover of the site due to illegal occupants,

addition to the scope of work, delay in the completion of infrastructure works by other contractors and change in design specification. Five extensions of time were granted. However, the decisions were issued on condition that the Contractor desists from any claim for compensation. At the last extension of time the Contractor refused to sign this condition and submitted a claim for compensation for costs incurred during the total extended period. An amicable settlement was reached.

Case 8

An agreement was signed for a private residential building. During tender negotiations the Employer decided to **add the scope of the works** and accordingly added the value of these works. The Employer later on decided to **change the concept of the design** dramatically and add basements. The Contractor had to suspend the works. Upon issuance of the new design drawings the works were resumed. However, the municipality stopped the works due to **permit noncompliance** with the executed new design. The Contractor further requested an extension of time and compensation. The disputed claim was resolved amicably.

<u>Case 9</u>

The consultant was appointed to a public hospital project 7 months after the start date of the project. This led to a **delay in the processing of submittals and approvals** of shop drawings and materials, abortive shop drawings, and delay in procurement of long lead item. Upon the appointment of the consultant the inadequacy of the electrical and mechanical design which was finalized 10 years before awarding the contract resulted in **redesign** and abortive shop drawings, in addition to the disruption of the regular progress of works and associated costs. This dispute was resolved through arbitration.

<u>Case 10</u>

An agreement was reached regarding a private residential project. The Contract allowed for deletion of parts of the works without compensating loss of profit. The Employer shielded himself under this clause on numerous occasions. The **modification of design** had resulted in delay of project completion. An extension of time was granted but the issue got resolved through arbitration.

<u>Case 11</u>

The Contract for a private residential building was signed. During execution of the works **design modifications** due to structural redesign along with the ongoing unstable political situation in the country resulted in a **shortage of manpower**. This caused delay to the works. The Contractor claimed for an extension of time and compensation. The Engineer's determination was not accepted by the Contractor but an agreement was reached regarding extension of time with the Employer conditional to the Contractor's acceptance of further design modifications to be transmitted on a preset schedule. **The war** of July 2006 led to the halt of the works and the demobilization of labor. The Contractor disputed the Engineer's assessment for an extension of time due to the war. Further delay was incurred by the Contractor due to further **design modifications and late reply to submittals**. The Engineer's assessment was late. The Contractor requested an Engineer's Decision. The Engineer issued an Engineer's Decision which was disputed by the Contractor. The Contractor issued a notice to proceed with arbitration. The dispute was resolved through amicable settlement.

<u>Case 12</u>

The Contract was signed for the execution of the electromechanical works of a public facility. The Contractor suffered from the prolongation of the project duration due to the **nonavailability of work areas**. The claimed value was based on the incurred cost during the delayed period due to site overhead, inability to execute other projects during this period, interest rates, extended warranties and head office overhead. The Contractor submitted the claim three times over five years each time presenting more substantiation as requested by the Engineer. An amicable settlement was reached.

<u>Case 13</u>

An agreement was signed for public road works. The Contractor suffered losses due to **late expropriation** of the land, **deletion of tremendous scope of the work** (63%), providing the Contractor with **wrong benchmark and reference** points (which led to redesigning the works), and **wrong design criteria** (which led to defects upon construction). Also, the Contractor claimed that he was put under a tremendous pressure to sign an MOU which was unfair and which was breached later on by the Employer. The

Contractor stated that the Engineer in this case was acting as an Employer's Representative and was not being impartial. The Contractor found the Engineer's Decision to be unfair and proceeded with arbitration.

<u>Case 14</u>

The Contract for the execution of a public water supply project was delayed due to failure by the Employer to pay the Advance Payment, delays in issuing the Order to Commence, Re-design of approximately 50% of the project, failure by the Employer to expropriate and grant possession of all sites, delays resulting from the legislation relating to the closure of all sources of quarries in Lebanon and delays due to exceptionally adverse climatic conditions. The Contractor presented a claim for extension of time and compensation showing the impact of the above mentioned delay factors. A settlement has not been reached to date for this public project.

<u>Case 15</u>

The Contract was signed for a public irrigation scheme public project. The Contractor gave notices during the execution of the works and claimed for the delay incurred due to different delay factors. The Contractor did not agree to the Engineer's determinations. All claims were then compiled at the end of the project for an extension of time and compensation claim for: price escalation of quarry material, steel and fuel due to changes in legislation, additional abortive engineering works due to redesign, disruption and loss of productivity, delays in settlement of the advance payment, price fluctuation due to Euro/U.S Dollar exchange rate. A settlement has not been reached to date.

<u>Case 16</u>

Agreement was reached for building a private shopping complex. The Contractor for this project was awarded the contract after his bid value was reduced 25%. This reduced the overhead and profit allocated by the Contractor. The contract duration was very tight. **Design modifications** were requested as the contract design was finalized 10 years prior to contract award and therefore it was no more suitable for the purpose of this shopping complex. The Contractor was granted an extension of time with no compensation as there was delay on the part of the Contractor running in parallel. This had placed the Contractor in a difficult financial situation as he could no more afford to provide extra

resources. The Contractor has been promised that his request for compensation would be considered towards the end of the project. This approach however, indirectly caused project delay.

<u>Case 17</u>

The Contract was signed for the building of a residential compound. The Contractor proceeded with the works and many **design modifications** were issued. Also, part of the scope was deleted. The unit rates for the additional works were disputed. According to the Contractor payments were withheld for unjustified reasons. All this resulted in cost implications that caused cashflow problems that delayed the works. The dispute is being resolved through arbitration.

<u>Case 18</u>

This project was awarded for building a university facility. However, tender was released in a short period where the design was not fully developed and coordinated. This led to underestimated BOQ quantities. **Variation to the design** induced during the execution of the works delayed the progress of work. This has led the Contractor to request Engineer's Decision for fair entitlement. The Decision was disputed by the Contractor. The latter proceeded with arbitration during execution of the works which resulted in tension builtup and further delay in the progress of works. The dispute is ongoing.

<u>Case 19</u>

The claims arising out of this residential project related to **design**, **price escalation due to increase in cost of steel**, **mixed aggregates**, **and fuel**. The Contractor claimed for an extension of time and compensation. The dispute is expected to be resolved in an amicable settlement.

<u>Case 20</u>

An agreement was signed for a public waterworks facility. The Contractor claimed an extension of time and compensation for additional cost incurred due to **late expropriation and design modifications**. The project had been completed in 2005. An amicable settlement has not been achieved yet.

4.3.Data Analysis

The data collected from these projects have been presented below in graphs to give an indication of the time and cost overruns on these projects. It should be noted that some of the data presented below needs to be further verified.



Figure 12 Type of Employer (Public vs. Private)



Figure 13 Forms of Contract

Figure 11 shows the industry or sector the project originates from; the three biggest sectors being: residential, public facilities, and universities. It is worth noting that this breakdown is not representative of the Lebanese construction industry. It only aims at categorizing the projects under study. Figure 12 shows that projects under study are almost equally divided between public and private sectors. Figure 13 shows that except for one, the remaining 19 projects adopt the FIDIC Red Book Fourth Edition. Figure 14 below shows that two third of the projects examined are resolved through amicable settlement.



Figure 14 Dispute Resolution Method



Figure 15 Amount of Variations



Figure 16 Final Contract Value vs. Original Contract Value

The causes of disputes as shown in the description of case studies above included: design modification, price escalation, failure to grant possession, delay in issuing order to commence, failure to pay, late expropriation, late reply to submittals, shortage of manpower, and permit noncompliance. However, the modification to design is noticeably a common cause of claim in almost all projects. This is also confirmed in Figure 17 which shows in the 20 projects an average of 45% variations to the original amount. Also, looking at this figure it is noticeable that the amount of variations exceeds 50% in four projects of which 2 exceed by 150%.



Figure 17 Percentage of added variations to original scope of work



Figure 18 Final Contract Duration vs. Original Contract Duration



Figure 19 Time Overrun (Final Duration/Original Duration)



Figure 20 Claim Value vs. Final Settlement

Figures 18 and 19 similarly show the time overrun which exceeds 100% in all of them i.e. the duration of all projects at least doubles with an average of 226% extension to the original contract duration. These figures reflect a high value of time and cost overrun in this pilot study. Although those might be mainly the direct result of the increase in scope of variations or other risk eventuating, the claims examined with respect to the final settlement value show an alarming amount of disputes. Figure 20 compares the final settlement value vs. the original claim value. This shows that the settlement/ award value is on average equal to 20% of the claimed value. It can be deduced from this figure that disputes in the value of 80% of the project value on average existed on these projects. The existence of disputes in the sample of projects examined confirms the need for a study to help minimizing disputes.

4.4.Summary

In this chapter, a field study was conducted on 20 construction projects in Lebanon to examine the disputed claims that were leading to disputes. It gave an indication of the types of contracts utilized on these projects mainly the FIDIC Red Book 4th Edition and the dispute resolution methods adopted. Also, analysis of the data collected reflected a high occurrence of major time and cost overruns as well as disputes on all the projects.

CHAPTER 5: INTERVIEWS

5.1. Introduction

Following the preliminary examination of 20 projects in Lebanon which confirmed the prevalence of disputes in the construction industry, interviews have been conducted with practitioners to explore their perception of dispute factors. As they share their experience in the field, this would give an insight on the "people" factor referred to in the literature through examining their knowledge and impressions.

The questions were set to allow for discussing conflicts and disputes to cover the dispute impacting aspects as had been reported in previous research and as perceived from the causes of disputes on the 20 projects examined in Chapter 4. As such the questions addressed the difficult construction project phases is procurement stage, contract execution and dispute resolution. This allowed the interviewees the freedom to expand on any of the areas that impacted disputes.

The interviews have been conducted with 24 professionals in the industry divided equally between contractors and consultants. The 24 interviewees belonging to different companies represent a cross section of the leading contracting and consulting companies in Lebanon. They include 6 consulting companies and 9 contracting companies. Employers were not interviewed because of the difficulty in reaching them noting that many of them were reported to be individual investors with limited experience in the construction industry.

The questions focused on minimizing construction disputes throughout the project lifecycle i.e. at the procurement stage, during execution of the works and at dispute resolution stage (which might sometimes occur during execution of the works). These questions were derived from the literature review.

The interviews were not recorded to allow the interviewees to discuss openly sensitive disputed matters. Any reference to names or project specifics is kept confidential as promised to the interviewees. The interviews were semi-structured in nature. In many cases in answering one question the interviewees addressed remaining questions. Accordingly, although all interviews covered the main questions originally set, the flow of the discussion dictated the order in addressing these questions. These interviewees were semi-structured in nature and carried the qualitative trait. As such they differed from a questionnaire where the answers are quantitatively compared. Also, as the questions were general, some interviewees elaborated on some questions more than others.

The findings can be classified into two types:

- Direct questions such as: "Who is the most influential party in dispute resolution? whether the Engineer is impartial? Is the DAB introduced in the new FIDIC 1999 a better substitute to the Engineer's Decision?"
- Discussions where each interviewee elaborated on his/her views and looked at the issue from different angles. Some interviewees touched on certain issues that others did not discuss as if the interviewees were adding different pieces of information that would help draw the full picture. These types of discussions were difficult to compare.

Gerson and Horowitz (2003) describe in-depth qualitative interviews as follows (a check against the comment confirms that this has been witnessed in the 24 interviews carried out):

"The best interviews become a conversation between two engaged people, both of which are searching to unravel the mysteries and meanings of a life.

Inevitably, some interviews will provide more useful information than others. No single interview, however, revealing, can offer more than limited insight into general social forces and processes. Only by comparing a series of interviews can the significance of any one of them be fully understood. And in the long run, each interview will add to the final story... Some participants are able to offer great detail and insight, while others find it difficult to recollect past circumstances or contemplate future possibilities... Where new interviews are more likely to confirm earlier insights than to spark new discoveries there is a good chance that theoretical saturation has been reached....If all goes well, these categories will be quite different than the ones that seemed obvious before the study began".

Meeting the writers' recommendation for good qualitative interviewing practices confirmed that the interviews were conducted in a healthy qualitative method and where the results compliment each other rather than allow for direct comparison the full picture would be drawn accordingly. Similar to an interview survey conducted by Dozzi *et al.* (1996), the study sample here is small and does not lend itself to statistical analysis. As such responses were analyzed and evaluated through drawing inferences and observing some trends and commonalities.

5.2. Interview Questions and Answers

The seven questions addressed in the interviews are listed below. The replies received were defined under different sub-headings to better identify the problem areas that were raised by the interviewees.

Question 1

Does risk allocation have an effect on dispute arising on a project? What is the effect of unfair/unclear risk allocation? How do you expect the contractor to control a major risk when it arises? How should the tenderer be selected? What about unrealistic tender pricing from the contractor's side as opposed to unrealistic time/cost/quality targets from the Employer's side?

Risk Allocation Best Practice/Common Practice

Almost all interviewees were aware and agreed to the common principle that risk should be allocated to the party that is best able to handle it. Otherwise the Employer would be paying for it upfront whether the risk occurs during project execution or not. When asked about the common practice in risk allocation at the procurement stage in Lebanon, all interviewees stated that the contracts in general are being unbalanced where most of the risk is shifted to the contractor.

Replies to the question 'who is responsible for specifying these conditions?' came as follows:

 Sixteen of the interviewees said the Employer is not educated and so the Engineer would put these conditions that suit the Employer to protect his/her interest by minimizing the chances of cost overrun. Eight of the interviewees (mainly Engineers) said the Employer would ask for these conditions, especially if he/she has a restricted budget and can not afford cost overruns.

Two main reasons behind the tendency to allocate risk to the Contractor identified from the replies were:

- The Employer desires to minimize headaches and cost overruns (some interviewees added that the Employer might have limited budget and be obliged to have these conditions)
- The Engineer allocated the risks based on previous experience to safeguard the Employer's rights and keep the contract in his/her favor. This will also allow the Employer to have an upper hand. Two contractors and three consultants said such practices an attempt to avoid abuse by the contractor based on previous experience.

The question was asked whether the contract can still be considered to be fair inspite of the unbalanced risk allocation. The answer was that since the Contractor has read it, verified it, accepted it and signed it then it is considered to be a fair contract.

Employer's Influence in risk allocation

According to the interviewees, Employers allocate all the risks to Contractor mainly for the following:

- The Employer has the upper hand because if the risk arises he might elect to pay for it as a gesture of will. However, the contractor has no rights to such compensation.
- They do not want to keep the budget open

The above is applicable to both the public and private sectors. However, the Government would be concerned with maintaining a healthy market and would look into a Contractor's major possible/potential losses. Private Employers do not share this concern. Some interviewees added that supply and demand have a major impact. Contractors would not have to sign/accept these Contracts where there is high demand in construction. The Employer by going to the very low bids incurs the risk of:

- The Contractor using cheaper alternatives to the specifications.
- The Contractor searching for potential claim areas that would compensate the losses.
- The Contractor failing to complete the project.

Contractor's allowance for risk premium/ Contractor's ability to sustain risk

In practice both the Engineers and the Contractors stated that contractors do not allow for risk or allow for a nominal (2-3%) risk that would not fall short of covering the risks that are allocated to the Contractor should they surface. The reason behind this was explained as follows:

- Maintaining a competitive edge
- Adopting a marketing strategy where the project would add to the Contractor's profile
- Shortage of qualified professionals in pricing to identify the risk and allow for the necessary markup and proper estimation of the resources necessary for the proper execution of the works
- Short periods for tender submission
- Identifying potential claim areas that would provide for profit sources during execution of the works

Interviewees added that this will not allow the contractor to afford the risks when they surface. They agreed that if the Contractor is not allowed for a risk premium in his offer where the risk eventuates a dispute is bound to occur. Even if the Contractor is not entitled to it, he will still claim for it if it has a major impact on the cash flow and affects his financial stability. Two contractors of the same contracting firm stated that their firm would refuse to sign under these conditions and this is normally set in their qualifications.

Contract Documents

Most interviewees stressed the importance of having a complete design before tender and drafting clear contract documents with accurate BOQ quantities and
specifications that are written for the specific needs of the project rather than copied and pasted from other projects. The importance of having a complete design was considered to be more detrimental in lump sum projects. Examples were given by some interviewees where cost and time overrun were incurred due to missing design information where the Engineer elaborated on the design during the execution of the works or where major redesign was made during execution of the works due to it being outdated. Also, clear achievable design/specifications issued were raised as an important factor that could minimize ambiguities and disputes resulting from contradicting explanations of the contract requirements.

Tender Evaluation

Most of the interviewees emphasized the importance of prequalifications in screening the tenderers. Having set the criteria for the participating contractors the two main criteria raised were the bid price and the project duration. In examining the bid price, some interviewees added that the qualifications included in the tender are to be examined to ensure proper comparison of bids. Also, almost all interviewees raised their concern regarding the importance of closely examining the lowest bid where it is found to be out of range.

Tender Period

The insufficiency of bid period allowed was raised by 2 consultants and 5 contractors. However one consultant argued that the bid period could be sufficient when was the main contractor prices the Concrete and the Architectural works only and the remaining parts are broken down to packages to be priced by the subcontractors. Other Contractors considered having experienced and adequate number of staff for pricing as more important than the bid period itself.

Question 2

What is the form of contract most commonly used? Why is the FIDIC Red Book 4th Editions more commonly used that the most recent one? What are the hazards of introducing particular conditions? Are they normally coherent or unbalanced or affect the clarity? If the answer is in the affirmative then what aspects are normally unbalanced? Could using a different form of procurement serve the purpose of the project better?

Fairness of Contract

FIDIC 4th Edition Red Book was confirmed to be commonly used due to familiarity. Some stated that FIDIC 4th Edition is popular because it is being adopted on projects by the lead consultant companies in the country. One interviewee added that the saying "The devil that you know is better than the devil that you don't know" could partly explain why practitioners are resistant to adopting new forms.

Types of Contract

The types of contracts used as per the reported popularity are as follows:

- Lump Sum: Very commonly used on projects. Reasons for using this type were explained to be:
 - Employer's need to cut down on the risk of cost overrun
 - Engineer's desire to cut down on the load of work in issuing lump sum payments rather than remeasured
 - Very prone to disputes especially in cases where tender quantities were not correct and there were major design changes
- Remeasured: Again commonly used especially in the private sector
- Cost Plus: There seemed to be a common agreement between Engineers and Contractors that this type of contract would reduce disputes especially those related to assessments and payments. It is not as common as the lump sum and the remeasured because there is a fear of seeing the Contractor abusing it as well as higher possibilities of cost overrun.
- Design build: Engineers and Contractors agreed that it reduces disputes especially resolving design issues. However, the Employers in Lebanon do not adopt it because it would give the Employer little control during the progress of the works.

• Partnering: Only few of the interviewees were familiar with the concept of partnering. They considered the culture not to allow for it.

Particular Conditions/ their influence on contract consistency

According to interviewees particulars are necessary to reflect specifics of a project. However, if not dealt with correctly, they might:

- Create an ambiguity where new conditions are added that do not match with general ones or even a loop hole where clauses are deleted and not replaced.
- Make conditions of contract unbalanced by shifting risks.

Modifications made to the general conditions and which normally become problem areas include:

- Price escalation. This is one major risk that rose lately where the government is looking for compensation and the syndicate of contractors is working on it.
- Procedure for notification of possible variation.
- Design liability.
- Employer's prior approval to any Engineer's time and cost determination
- Order of precedence of Contract documents. One interviewee stated that on one project precedence was given to the Engineer's discretion who in turn gave preference to the document that best covers Employer's benefit.
- Deletion of works to be assigned to another contractor without compensation of profit

Question 3

Who is/are the key person/persons on a project that can contribute to resolving a dispute? Is the role of the construction manager and/or the project manager critical (in other words would lack of competence in this case affect the project)? How do you best describe the role of the Employer's representative? Would the behavioral attitudes of these persons contribute to the amount of disputes on the project?

Key Persons on the Project

When asked who of the 3 people (the Project Manager, the Engineer, or the Contractor) has the most influence on minimizing disputes, out of the 24:

- 11 answered the Engineer should hold the threads, maintain a cooperative spirit, be fair and make recommendations to the Employer
- 5 answered the Contractor has control over the actual progress of works, makes critical decisions and minimizes disputes especially when allowed for good risk and profit margins.
- 4 answered the Employer Representative (ER) has the most influence. If the Engineer is competent but does not have a well experienced ER that would collaborate on critical issues then there is little that the Engineer can do. ER would have the final decisions on Engineer's recommendations as the Engineer would have no right to waive any of the ER's rights
- 2 answered the three of them should be qualified
- 2 answered the Engineer and the Contractor

Proactive attitude in resolving problematic issues

All interviewees strongly agreed to the fact that the behavioral attitude has a major effect on the progress/success of the works (time and cost overruns). The following was repeatedly heard in this regard:

- Personal conflicts might occur between any of the three parties
- The Engineer should have a personality that could weigh difficult situations and resolve them
- There is 'chemistry' between the people working on the same project. By that the interviewees meant the instinctual reaction of participants that would influence them liking or disliking each other. Sometimes one person that might be a troublemaker on one project, might fit perfectly on another project with a different team.
- Behavioral attitudes might have a bigger effect in Lebanon because the people by nature are more emotionally driven.

Employer's Representative

- Most interviewees (21) stated that the Employer's Representative should only interfere on strategic issues related to time and cost. His interference should be motivated by the significance of the variation. He may pinpoint mistakes or issues that the PM might be overloaded to notice. However, his important role is in examining recommendations by the Engineer and taking the decision that is in the best interest of the project.
- The remaining three (mainly Contractors) stated that ER has no role under the FIDIC and is not needed on a construction project.

Direct Dispute Factors

During the execution of the works disputes were attributed to a variety of reasons:

- Engineers considered that the Contractor suffered from bad planning and lack of coordination with subcontractors. This leads to time and cost overrun which the contractor tries to cover through claims.
- Contractors stated that an Engineer who does not have the necessary site expertise which would allow him to have reasonable constructible instructions instead of insisting on every detail in specification would drive towards increased number of disputes.
- Disagreement on time and cost entitlement assessment was raised as a problem area by both Engineers and Contractors.

Question 4

What are the interviewees' views on: the importance of giving notice, proper claim documentation and presentation, the Engineer's prompt attendance to the claim? Would it help the project to resolve disputes early on? Would you be willing to compromise on some of the disputed issues to resolve them?

Timely response to contractual matters (giving notice, making determinations...)

All agreed that it is very important to give a notice of a time or cost overrun so as to allow the Engineer to take alternative decisions where necessary. Also, it is very important for the Engineer to make timely determinations. Where the effect is ongoing, interim determinations should be made so that Contractor can make financial decisions accordingly.

One Contractor said that the Engineer might be late in answering because he tries to delay the negative answer. One consultant said the reason for late reply was the shortage of steel required to handle the necessary workload in a timely manner.

As for the time constraint for the Engineer's assessment introduced in the 1999 suite of the FIDIC contracts:

- 91% stated that it would be fairer to the Contractor to receive his determination within a specified time period.
- 9% (2 Engineers) warned that this might have a negative drawback. If the time frame set is not closely examined to be reasonable & where the Engineer might not have enough staff this might induce the Engineer to make premature determinations.

Contractor's willingness to compromise

Both Engineers and Consultants stated the following:

- The Contractor might compromise to resolve disputes depending on the value of the dispute and its impact on his cash flow.
- Where the Contractor knows that the disrupted cost implications are significant in value and that he is fairly entitled to it, he will be willing to proceed with arbitration.

Question 5

If the Engineer makes a determination regarding a disputed issue, what factors contribute to making a different Engineer's Decision? What are your views on DAB in the FIDIC 1999 Red Book as an alternative?

Engineer's Decision

Most consultants stated that the Engineer would only revert from/alter his decision if this is in contradiction with the original determination. Most of the contractors on the other hand stated that the Engineer rarely changes the original determination. Some added that the Decision is influenced by the Employer. Interviewees from both sides said it would be a better practice to have the Engineer's Decision made by a different entity in the Engineer's office as opposed to the Engineer's representative on site.

Half of the interviewees stated that the Engineer's Decision provides a chance for the Engineer to be impartial and avoid the Employer's influence on determinations.

DAB in FIDIC

Only 2 interviewees reported using the new 1999 FIDIC on 2 projects but the DAB clause was removed in both.

When questioned about their views on adopting in the 1996 Supplement o the 1992 Red Book Fourth Edition and later in the 1999 FIDIC suite of Contract and whether that would help reduce disputes different replies were received. There were divided as follows:

Engineer's views on DAB

- 4 of the Engineers were not familiar with the 1999 FIDIC
- 5 believed that it would be another buffer that would help resolve disputes and would help maintain a more fair and impartial environment. It allows for mediation attempt that the Engineer is not allowed for.
- 3 believed that the Engineer is more familiar with the project and is able to better assess the situation. If he is given the power under the contract he is best qualified to resolve disputes. Also, there was a concern about the availability of professionals that would qualify as DAB members.
- One Engineer added that: "The DAB might not be able to meet the expected results in terms of dispute resolution. Arbitration at some point in time was

considered to be the best approach to dispute resolution but in many cases it failed to meet the aspired results."

Contractors views on DAB

- 3 of the Contractors were not familiar with the 1999 FIDIC
- 7 agreed that the 1999 FIDIC is a better option for the Contractor as it would provide impartiality where the Employer can no longer have direct influence on the decision.
- 2 expressed concern about the new DAB as it might introduce more individuals to give their opinion on the dispute and this might further complicate the dispute.

One Contractor added that one hazard of the 1999 FIDIC is that "the Engineer would no longer have his original role-of being the fair 'judge' between the Contractor and the Employer. In fact the Engineer who has a very critical role on the project might automatically turn into an Employer's representative on all issue and the DAB will be the judge."

<u>Question 6</u>

After the Engineer's decision is issued, if one of the parties is not content with the decision, this party should give a notice of intention to proceed with arbitration within 84 days? Would this affect the ongoing relationship between the parties? How is the dispute dealt with beyond this point?

Arbitration, adjudication, mediation

All interviewees agreed that proceeding with arbitration would create adversarial relationships between different parties. Therefore, it is better not to proceed with it till the end of the project due to two reasons:

- An amicable settlement might be reached
- It will require a lot of effort/resources from the people working on the project. Thus, it is not advisable during the execution of the works.

Few of the interviewees added that the Contractor might need to proceed with arbitration if:

- The number of disputes increases
- The value of the disputes impacts his financial stability

Negotiation Basket and Amicable Settlement

Interviewees stated that there is a common trend to amicably resolve disputes after completion of the works.

The negotiation basket approach used to reach an amicable settlement was reported to be to the Employer's benefit as it:

- Withholds payment of all claimed amounts till the end
- Allows the Employer to exert pressure on the Contractor and in some cases propose an amicable settlement whereby one party waives his right to liquidated damages against the other party's entitlement to claims.

Delaying dispute resolution till the end might avoid adversarial relations. However:

- Late payment might cause financial problems to the Contractor.
- There will be a trend of submitting more claims to have a stronger case at the end, thus encouraging more disputes.

Question 7

Has the ongoing political situation in Lebanon had an effect on the causes of dispute and/or on dispute resolution?

Regarding the effect of political situation in the country:

- All interviewees agreed that the war does not have a direct effect on the enforceability of the contract. It does however have indirect effect and causes:
 - Dispute due to delay impact where internal political conflicts/turmoil occur
 - Dispute over the contractor's entitlement to compensation due to Employer's risk

 Employee turnover and in some cases loss of experienced personnel during execution of the works

5.3. Analysis of Replies

The replies obtained were consistent with similar studies conducted by Dozzi *et al.* (1996) and Hartman (2003), and Shleifer (1990) which had derived the following observations:

- Contractors save claims until the project is complete or almost complete because they do not wish to compromise/ jeopardize their relationship with the owner. This was also the opinion, though not as strong, for relationships with the consultants.
- Construction contracts apportion risks unfairly to the contractor and to subcontractors. They do not apportion risk unfairly to the owner or the consultant.
- Exculpatory clauses increase the likelihood of a contract dispute.
- Consultants who act as contract administrators on behalf of their clients often lack objectivity in making decisions about contract issues and their interpretation.
- More efficient risk management will reduce the final cost of construction to the owner.
- Contractors should be screened and prequalified before being allowed to bid on a contract.

Furthermore, two main observations can be drawn from these replies:

- Although the subject areas raised by interviews reflect their understanding of the different dispute factors, not all were equally knowledgeable in certain contractual aspects. This is evident from the DAB question where around one third of the respondents were unfamiliar with the new FIDIC 1999. The statement made by three respondents (contractors) that the Employer's representative had no role under FIDIC was not correct. 70% of the interviewees expressed lack of knowledge of new procurement methods such as partnering. Little knowledge was reported in dispute resolution techniques devised abroad and the concept of team building.

Contract Administration	Dispute Resolution Mechanism		Assessment of Time and Cost Entitlement		Timely Response to Contractual Matters		Contractor's Expertise in	Management	
Behaviour	Key Persons Assigned &	Adequacy of Staff Allocated		Proactive Attitude in Resolving	Problematic Issues	Employer's Representative	(scope of work)		
Risk	Risk Allocation (Best Practice/ Common Practice)	Employer's	Allocation	Contractor's Allowance for Risk	Premium	Forms/Types of Contracts Adopted		Particular Conditions	
Tender	Contract Documents		Tender Period		Tender Evaluation				

Figure 21 Dispute Influencing Areas Emerging from Interviews

73

The seven questions raised in the semi-structured interviews helped identify 15 dispute influencing areas that emerged from the discussions. These are shown in Figure 21 below. They have been categorized under four themes: Tender, Risk, Behavior and Contract Administration. These are identified as dispute influencing areas because each has its contribution in creating or augmenting the incidence of disputes. By way of example, an ambiguity in the contract documents would give rise to a dispute, lack of chemistry among participants or even the lack of proactive attitude would worsen the situation. Thus the deficiency in the dispute resolution mechanism would prevent efficient resolution. Moreover, if a Contractor risk eventuates in an unbalanced contract where proper risk allocation is not accounted for and the Employer doesn't interfere to resolve the same, then a dispute is likely to result due to the financial difficulties.

5.4. Summary

The interviews of 24 practitioners in the industry equally divided between Engineers and Contractors were carried out as the second level of fieldwork. It reflected the understanding and comments of practitioners of the common practices. The respondents highlighted through their replies the significance of 15 dispute influencing areas. These were categorized under four themes: tender, risk, behaviour, and contract administration.

The interviewees expressed an understanding of healthy practices in procurement that would allow for minimizing disputes in line with what was recommended in the literature examined. Examples of such healthy practice includes allocation of risk to the party that could best handle it, the importance of having complete design, the importance of closely examining the lowest bidder, the impact of manipulating the particular conditions, the impact of behavioural factor of participants, the importance of giving notice. The interviewees also expressed differing views on matters such as who was responsible for risk allocation in contract formation, the advantage of replacing the Engineer's Decision by a Dispute Adjudication Board (DAB). The DAB mechanism was not reported as per the Interviewees experience on Lebanese projects. Although the interviewees expressed awareness of the contractual implications on disputes, one third of the interviewees who are assigned to administer contracts, had limited knowledge related to more recent forms of FIDIC and new methods of procurement adopted in other countries.

CHAPTER 6: DISPUTE CASE STUDIES

6.1. Introduction

Following the two fieldwork analysis carried out in previous chapters to examine disputes in Lebanon through analyzing project data and through interviewing practitioners, this chapter analyses disputes through an in-depth analysis of 50 case studies on four projects. As such the case study approach adopted is the multiple case design with embedded units of analysis (50 dispute cases). The data related to the dispute cases was gathered through examining project documentation including correspondences, claims and Engineer's Decisions.

6.2.Analysis Approach

The Case Study approach adopted is the multiple-case design with embedded units of analysis. The Cases are four ongoing projects that consist of shopping complex, residential tower, university building, and hotel with the following values: US\$15M, US\$75M, US\$52M and US\$95M. All four projects use the FIDIC form of contract. However, extracts of the Particular Conditions modifying the General Conditions are shown in Appendix D to show how risk was allocated and shifted in many cases to the Contractor. The extracts of clauses 2.1, 2.7, 5.2, 70.1, and 70.2 confirm what was said in the interviews regarding common practice in modifying the risk allocation in FIDIC form of contract. The FIDIC form of contract is further described in Chapter 7.

After the chronology of events for each of the 50 cases was listed, an analysis of the events was necessary. Through the analysis both the risks eventuating and the dispute factors would be examined. However, this necessitated devising a uniform framework for analysis. For this reason, the interrelationship between the risk, dispute factors, contract conditions and behavioral attitudes was to be examined. This was done as follows:

- Risks that could eventuate on a project were identified using Bunni's (2003) generic spectrum of risks. These are listed in Table 6.
- Zack (1996) had in his work identified the list of risks that are addressed in the conditions of contract (noting that different forms of contract might address them differently). The risks identified by Zack are included in Table 7.

Examining Bunni's risks against Zack's risks, it is noticed in Table 8 that Zack's risk addresses those specified by Bunni except for those relating to behavioral attitudes. This can be verified by the mere fact that the contract although does attend to time frames for submittals and replies does not make provisions for the cases where the parties have developed adversarial relationships. The contract however makes provisions for dispute resolution mechanisms where disputes arise due to the same.

Spectrum	of Risks
Riot and civil commotion, Arson, Strike, Malicious acts	Brief and remuneration
War, nuclear reactions etc	Financial stability
Acts of God (Excessive Rainfall, Flood and inundation, Wind and storm, Hurricane, tornado and whirlwind, Extremes of temperature, Cyclone, Earthquake)	Inefficiency and delays
External stability of Government	Extended duration of construction
Internal stability of Government	Programming the work
Inappropriate choice of design	Negligence and lack of care
Negligence and lack of care	Incompetence
Lack of knowledge and checking	Inadequate site supervision
Adequacy of site investigation	Inadequate site management
Adequacy of surveys and inspections	State of the art, codes and knowledge
Unforeseen Site Conditions (Topography and surface water runoff, Adverse geological and underground characteristics, Underground obstructions)	Defective design
Variation from contract documents	Risks associated with dispute resolution
Choice of site	Failure to take account of foreseen problems
Transit to site	Use of untested and proven techniques
Defective workmanship and material	Technical Complexity and new methods
Defective design, workmanship and quality control	Removal of Support
Mechanical and electrical breakdown	Dangerous substances and items during constructions and commissioning
Defective Temporary works and their design, Corrosion, Collapse, Collapse of temporary works	Taxes and the Stability of the Legal System, Red tape
Human error	Acceptability of projects by locals
Failure to comply with insurer's conditions and requirements	Inadequate performance of equipment
Choice of contractor or subcontractor	Lack of safety precautions
Theft and Burglary, Illegal activities, Faud and infidelity, Impact	Lack of communication
Adequacy of finance and related aspects	Owner Choice of Professional Team

Table 6 Spectrum of Risks identified by Bunni (2003)

Force Majeur
Impracticality/impossibility
Latent site conditions
Quantity variations
Site access
Weather
Defective work
Labor forces
Subcontractor, supplier failure
Contract Termination
Cost Escalation
Failure to pay
Project funding
Taxes
Acceleration
Delays and disruptions
Early use of facility
Suspension of work
Untimely responses
Changes
Contractor-furnished equipment/materials
Coordination
Defective contract documents
Interpretation of requirements
Means and methods of construction
Permits and licenses
Productivity
Site safety
Work Quality

Table 7 Risks identified in Contract by Zack (1996)

As such analysis was carried out where the following are examined for each of the 50 case studies: Zack's risks, and dispute factors including behavioral attitudes. For this reason an analysis sheet as shown in Figure 23 was devised where dispute factors are listed at the right side vertical column. Zack's risks are listed in the lower horizontal rows. Behavioral description attitudes examined through the analysis are listed in the last row.

-	wher Choice of Professional Team	ol	1	1	T	-1		- 1'-	T	<u>}</u>	1							. 1			T							<u> </u>	1	Т	┯	\top	٦
-	.sek of communication		1	+-	+		+		+					-	-+		+-		-+	-+-				+						+		╋	1
-	sack of safety precentions				-	-+-	╡	╀	+		-+-		\vdash				+				+			+						+	+	+	{
-	uauequate periormance of equipment		+	+	+	+	+		+-	<u> </u>				-	-+			-	+	-+									_	+	Ļ		1
-		-	+	_	_	_	-	+	+-	ļ		_		-	_				_	+		+		-	+			ļ	_		1	\perp	1
-		7	+		_	_		_			_	_		-	\downarrow		<u> </u>	-	_	_					1			Ĺ	×		\perp	\bot	1
-	exes and the Stability of the Legal System, Red ape																												×				
	angerous substances and items during constructions and commissioning	a I																										×					
-	temoval of Support	a l				1		\top					-+	-	+		+-		+	+	+			+	+	+		×	1	\dagger	+	Η	
	sborhen wen bne vitselgmo O leoinaise		1		1	1					+-		1		\uparrow	-	+		-					+	1	+		×		1	+	\square	
-	lse of untested and proven techniques	1	<u> </u>	1	+	+	+	1	\square		-	1	+	-†	-†-		1	-	+	+	+	<u> </u>		+				×	1	+	+	Ħ	
-	allure to take account of foreseen problems		1	+	╈	+-	╉	┼─			-	-		+	+	+-	┼──	-	-+-	+	+			-+				*		+	╀	Н	I
-	בוצע ב שבמכו מנהמ ואוני מו באחונה נקבטותה סט	,			+	┿	+	╞	$\left \right $		+	\rightarrow	+	╉	-+-	-+		-	-+-	+	+			+-	+	+				+	+-	++	
-		<u></u>			+	+		+	\square		+	-+	+	+	_	+	-	+	+	+	+			┢	+	-				╀	+	Н	
-		1			+		_	1				_	+		-		<u> </u>	\downarrow	+	+-	+			-	<u> </u>	_				\downarrow	+	Ц	
_	State of the art codes and knowledge	5				_	1					_	_					\downarrow							×		×			_		Щ	
	nadequate si te management	1																_						×									
?`_	nadequate site supervision	1																		×	<								1				
	escence	I		T			Τ	T	Π						Τ					_	-			Τ					1		\square		
r -	Vegigence and lack of care	ł		1	T	-										1				×	-			1						1			
	Programming the work		1	†	\uparrow	\uparrow	1	1	╞─┨		+-	-	+	1	\uparrow	\top	×	-	1	1	+			1	1	\uparrow			1	t	$\uparrow \uparrow$	\square	
ĭ-	nousure of new of construction		1	+	+	+			-		+	-+	+	+	+	+	×		+	-	+					+		•		-	+-+		
<u>,</u>	דופונובונכע שנו מפועא צ		<u> </u>		╋	-†	+	+		· · · ·	+	+	+	+	+-	×		+	+	+	+			+						+-	+ - 1	H	
-					┿	+-		┢				-+-	+	+	_	+	<u> </u>		╉	+	+			+-		-+-	+			-	\vdash	H	
	itigen einen ein	1				-						+	Ļ	1	1	+-		-	+-		+			-						-	+		
5 –		Ц 	<u> </u>		+		-						,	×) >	×	<	L				\vdash						\rightarrow				+ -	\square	
: 	Adequacy of finance and related aspects	'	ļ		-	1	1				1	-	× ;	<	<	<	L			_	1									L			
	infectivy, Impact	ŗ									,																		1				
) –	Theft and Burglary Illegal activities Faud and	-				1	\vdash		\square		_	\rightarrow			-	╞	ļ		_	_						_				L		\square	
; _	Choice of contractor or subcontractor)							×	×	L				\perp			\perp						1.		_							
	sus usinba	1						×																									
<u>)</u> -	bue to comply with insuer's conditions and	1	 		+						<u> </u>	\rightarrow				4_		_	1.					1		+					\downarrow	Ц	
<u>-</u>	Human error	I_					_	×			<u> </u>	_			_				_						×				L	L		×	
5_	Defective Temporary works and their dealgn. Corroaion, Collapse, Collapse of temporary works							×																									
	Mechanical and electrical breakdown	1						×																									
	Defective design workmanship and quality contro	I			T	Γ	Γ	×							Т				Τ					1							Π		
<u> </u>	Defective workmanship and material	1			Т	1	1	×			1			T	1	1			T	1		:	×	1		T					H		
] -	Transitio site	1			╈	×		\square					-	+-		+	<u> </u>	-	+	+	Η			+		+-							
- 1	Choice of site			×	+	+		\square				+		┿	+			-	╈				·· · · ·	+-				· -			\vdash		
-		1-			╈	+	ł	-				-	+	+	+	+		-	+-		÷			┼─		╉	-			\vdash	++	\exists	
-	סכנון אויסה) סכנון אויסה) ער מרויר לכד בסוימנו אות חופי אוסואוסה אות					+	\uparrow					-+	+		+-						Ê					+				-			
-	Unforessen ale conditions (Ground movement, Unforessen ale conditions (Ground movement,				-	-	-		_					_				_		-						-	_						
) 	Unforessen Site Conditions(Topography and authce water moeth.Adverse geological and underground			×										1															i				
-	Acequacy of surveys and inspections	·		×	1	1	1				1				1	1		1	\top	1	Π			\uparrow	×	1	-			F	$[\uparrow]$		
-	Adequacy of site investigation	1		×		1		Π			1	-+	\uparrow	1		1	-		1	\uparrow	$\uparrow \uparrow$			\top	×	1			1	†	\square	\neg	
-	Lack of knowledge and checking		×		+	+-	1	H			1	+	+	\uparrow	+			-†-	+-	+	┞┨			1			\rightarrow				┟┤		
-	דאפצונצפרוכפ אומינג סו כאנפ	+	×		+	╋	1-		-		<u> </u>	+	+	+-	+	+		+-	+	+	╞╴┨			+	×	+-	×			-	++	\neg	
-	ערשו:	+			+	+	+					-+	╋	+	+	+		+	+	+	┼╌┤			+	<u> </u>	+-	 +		<u> </u>	┝	┝┤	\vdash	
-		+-	<u> </u>		+	+			-+		\vdash	+	+	+	+	+		+-	+	+	$\left - \right $			+-	Ļ^	+				┞	\vdash		
_	Internet of the second state (second second s	Ļ		<u> </u>	+	+-	$\left \right $	\vdash				_	+-	+	+	+ -		-	╀	+				+		+-			 		\square		
-	Extemal stability of Government	×			+-	\downarrow	 							+	+	\downarrow	—	-	+-	+	\square			1					 	L	\square		
	wing and morent, intractice, intrado and whithwind, Extremes of temperature. Cyclone. Earthquake)				1																								1			1	
	Acts of 0 od (Excessive Rainfall, Flood and immastion,	Î				1									1									1					1				
-	wat, nuclear reactions etc	×		†	+-	+	\square		+		<u> </u>	-+	+,	;				+	+	+	┢╌┥		·····	+		+	+			+	┟╌┥		
-	acts	×			+-	+			+			1	+	-	1-				+					+		-							
	Riot and civil commotion, Arson, Stak, Malicious		s		+	-	-		_		_	+	+		+	$\left - \right $		+		4 5				-				s	- v		$\mid \mid$	Н	
		jeur	ality/impc		variations	s		work	ces	actor, silure		uo	uation new	nding	9	ion	q	1S of facility	TITLE IN OF	response		2	t/material	ion	contract	ion of	nts	d method ction	d license.	tv		lity	
		Force Maj	Impractic: sibility	Latent site	Outantity v	Site acces	Weather	Defective	Labor for	Subcontre supplier fe	Contract	Terminati	Cost Esca	Project fire	Taxes	Accelerati	Delays an	disruption	Shenenein	Untimely r	Changes	Contractor	fumished	Coordinati	Defective	document Interested	requirement	Means and of constru-	Permuts an	Productivi	Site safety	Work Que	



Figure 22 Sample Chronology of Events for Dispute Case

	CASEANALYSIS	Project: A						
Case No.: 4	Subject: Façade glass	Section: Façade						
Synthesis:		General Observations:						
Façade details were raised in a	workshop meeting held after	1 ate intervention of						
<u>14 months</u> in a 30 months cont	CASE ANALYSISNo.: 4Subject: Façade glassNo.: 4Subject: Façade glassSchesis:In a 30months contract, this issue given its ality and that it is not clearly specified in terms of frit n should have been initiated earlier on. Also, both parties of account for the time period required to receive each le. The Contractor had clearly underestimated the time d to provide samples in the programme. This item was ibed as design build in the BOQ. There was <u>inderstanding regarding the description as the actor considered the internal skin as a design build m. to meet specified load whereas the Contractor dered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build m. It was the Engineer's understanding that both the hal skin and external skin are design-build. The neer did not request full range of samples at the first ission to save time. Moreover the notification of possible was rejected although this issue had become critical to oject and was delaying the works and an extension of was later granted for this delay.So frisk (as categorized by Zack, 1996) identified in The process of providing the 4 s The Engineer insisted on the fact that the delay wa responsibility.So the Engineer insisted on the fact that the delay wa responsibility.So the parties were late in attending to this subject.</u>							
críticality and that it is not cle	CASE ANALYSIS2 No.: 4Subject: Façade glassthesis:de details were raised in a workshop meeting held after onths in a 30months contract, this issue given its cality and that it is not clearly specified in terms of frit gn should have been initiated earlier on. Also, both parties not account for the time period required to receive each oble. The Contractor had clearly underestimated the time ed to provide samples in the programme. This item was ribed as design build in the BOQ. There was understanding regarding the description as the ractor considered the internal skin as a design build em to meet specified load whereas the Contractor idered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build em. It was the Engineer's understanding that both the nal skin and external skin are design-build. The uneer did not request full range of samples at the first aission to save time. Moreover the notification of possible y was rejected although this issue had become critical to roject and was delaying the works and an extension of was later granted for this delay.The process of providing the 4 s The Engineer was late in replyi The responsibility of designing allocated clearly							
design should have been initiat	gn should have been initiated earlier on. Also, both parties not account for the time period required to receive each ple. The Contractor had clearly underestimated the time ed to provide samples in the programme. This item was ribed as design build in the BOQ. There was <u>understanding</u> regarding the description as the cractor considered the internal skin as a design build em to meet specified load whereas the Contractor idered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build							
did not account for the time per	fod required to receive each							
sample. The Contractor had clea	arly underestimated the time							
needed to provide samples in the	the provide samples in the programme. This item was bribed as design build in the BOQ. There was <u>understanding</u> regarding the description as the tractor considered the internal skin as a design build tem to meet specified load whereas the Contractor							
described as design build in the	ded to provide samples in the programme. This item was cribed as design build in the BOQ. There was <u>sunderstanding</u> regarding the description as the stractor considered the internal skin as a design build tem to meet specified load whereas the Contractor sidered the fritting on the external skin as an aesthetical in to be chosen by the Engineer and not a design build tem. It was the Engineer's understanding that both the proved skin and external skin are design build.							
misunderstanding regarding t	CASE ANALYSIS3e No.: 4Subject: Façade glassanthesis:ade details were raised in a workshop meeting held after wonths in a 30months contract, this issue given its icality and that it is not clearly specified in terms of frit gn should have been initiated earlier on. Also, both parties not account for the time period required to receive each uple. The Contractor had clearly underestimated the time ded to provide samples in the programme. This item was pribed as design build in the BOQ. There was understanding regarding the description as the tractor considered the internal skin as a design build tem to meet specified load whereas the Contractor sidered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build tem. It was the Engineer's understanding that both the rnal skin and external skin are design-build. The gineer did not request full range of samples at the first mission to save time. Moreover the notification of possible ty was rejected although this issue had become critical to project and was delaying the works and an extension of e was later granted for this delay.as of risk (as categorized by Zack, 1996) identified if The Engineer was late in reply The responsibility of designin allocated clearlyavioral Observations:							
Contractor constaered the interi	CASE ANALYSISe No.: 4Subject: Façade glassuthesis:ade details were raised in a workshop meeting held after souths in a 30months contract, this issue given its cality and that it is not clearly specified in terms of frit gn should have been initiated earlier on. Also, both parties not account for the time period required to receive each ple. The Contractor had clearly underestimated the time ed to provide samples in the programme. This item was ribed as design build in the BOQ. There was understanding regarding the description as the rractor considered the internal skin as a design build em to meet specified load whereas the Contractor idered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build em. It was the Engineer's understanding that both the nal skin and external skin are design-build. The fineer did not request full range of samples at the first nission to save time. Moreover the notification of possible g was rejected although this issue had become critical to roject and was delaying the works and an extension of was later granted for this delay.as of risk (as categorized by Zack, 1996) identified uents allocated clearlyavioral Observations:The Engineer insisted on the fact that the delay v responsibility.both parties were late in attending to this subject							
system to meet specified tour w	CASE ANALYSIS e No.: 4 Subject: Façade glass athesis: Subject: Façade glass athesis: ade details were raised in a workshop meeting held after works in a 30months contract, this issue given its cality and that it is not clearly specified in terms of frit gn should have been initiated earlier on. Also, both parties not account for the time period required to receive each ple. The Contractor had clearly underestimated the time ed to provide samples in the programme. This item was ribed as design build in the BOQ. There was anderstanding regarding the description as the tractor considered the internal skin as a design build em to meet specified load whereas the Contractor idered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build em. It was the Engineer's understanding that both the nal skin and external skin are design-build. The ineer did not request full range of samples at the first nission to save time. Moreover the notification of possible y was rejected although this issue had become critical to roject and was delaying the works and an extension of was later granted for this delay. as of risk (as categorized by Zack, 1996) identified y and Disruption The Engineer was late in reply The responsibility of designin allocated clearly avioral Observations: The Engineer insisted on the fact that the delay weresponsibility. Both parties were late in attending to this subject.							
item to be chosen by the Engine	CASE ANALYSIS ie No.: 4 Subject: Façade glass nthesis: ade details were raised in a workshop meeting held after months in a 30months contract, this issue given its icality and that it is not clearly specified in terms of frit gn should have been initiated earlier on. Also, both parties not account for the time period required to receive each uple. The Contractor had clearly underestimated the time was worked to provide samples in the programme. This item was tribed as design build in the BOQ. There was understanding regarding the description as the tractor considered the internal skin as a design build tem to meet specified load whereas the Contractor sidered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build tem. It was the Engineer's understanding that both the rnal skin and external skin are design-build. The spineer did not request full range of samples at the first mission to save time. Moreover the notification of possible y was rejected although this issue had become critical to project and was delaying the works and an extension of e was later granted for this delay. as of risk (as categorized by Zack, 1996) identified the responsibility of designin allocated clearly avioral Observations: - The Engineer insisted on the fact that the delay v responsibility. - Both parties were late in attending to this subject.							
system. It was the Engineer's u	CASE ANALYSIS Ste No.: 4 Subject: Façade glass Inthesis: add details were raised in a workshop meeting held after months in a 30 months contract, this issue given its icality and that it is not clearly specified in terms of fritign should have been initiated earlier on. Also, both parties not account for the time period required to receive each typle. The Contractor had clearly underestimated the time ded to provide samples in the programme. This item was writed as design build in the BOQ. There was understanding regarding the description as the tractor considered the internal skin as a design build tem to meet specified load whereas the Contractor sidered the fritting on the external skin as an aesthetical to be chosen by the Engineer and not a design build tem. It was the Engineer's understanding that both the irrnal skin and external skin are design-build. The gineer did not request full range of samples at the first mission to save time. Moreover the notification of possible in y was rejected although this issue had become critical to project and was delaying the works and an extension of e was later granted for this delay. as of risk (as categorized by Zack, 1996) identified atimely responses the Engineer insisted on the fact that the delay v responsibility. The Engineer insisted on the fact that the delay v responsibility. Both parties were late in attending to this subject.							
internal skin and external skin	CASE ANALYSIS Ste No.: 4 Subject: Façade glass nthesis:							
Engineer did not request full r	CASE ANALYSIS Ste No.: 4 Subject: Façade glass whetesis: Sade details were raised in a workshop meeting held after months in a 30months contract, this issue given its citality and that it is not clearly specified in terms of frittign should have been initiated earlier on. Also, both partiel not account for the time period required to receive each uple. The Contractor had clearly underestimated the time ded to provide samples in the programme. This item was cribed as design build in the BOQ. There was sunderstanding regarding the description as the thractor considered the internal skin as a design build stem to meet specified load whereas the Contractor sidered the fritting on the external skin as an aesthetical in to be chosen by the Engineer and not a design build stem. It was the Engineer's understanding that both the ernal skin and external skin are design-build. The gineer did not request full range of samples at the first project and was delaying the works and an extension of the was later granted for this delay. eas of risk (as categorized by Zack, 1996) identified allocated clearly interpretation of the timely responses The process of providing the 4 the responsibility of designi allocated clearly avioral Observations: The Engineer insisted on the fact that the delay is responsibility. Both parties were late in attending to this subject							
submission to save time. Moreo	CASE ANALYSIS set No.: 4 Subject: Façade glass Inthesis: scade details were raised in a workshop meeting held after months in a 30months contract, this issue given its cicality and that it is not clearly specified in terms of frit ign should have been initiated earlier on. Also, both parties lost account for the time period required to receive each uple. The Contractor had clearly underestimated the time ded to provide samples in the programme. This item was cribed as design build in the BOQ. There was sunderstanding regarding the description as the tractor considered the internal skin as a design build istem to meet specified load whereas the Contractor sidered the fritting on the external skin as a design build istem. It was the Engineer and not a design build item. It was the Engineer and not a design build item. It was the Engineer sunderstanding that both the ernal skin and external skin are design-build. The gineer did not request full range of samples at the first insission to save time. Moreover the notification of possible ay was rejected although this issue had become critical to project and was delaying the works and an extension of ewas later granted for this delay. eas of risk (as categorized by Zack. 1996) identified ay and Disruption The Engineer insisted on the fact that the delay is responsibility. a The Engineer insisted on the fact that the delay is responsibility. both parties were late in attending to this subject.							
delay was <u>rejected although this</u>	ISE NO.: 4 Subject: Façade glass gathesis: Subject: Façade glass grade details were raised in a workshop meeting held after months in a 30months contract, this issue given its ticality and that it is not clearly specified in terms of frit sign should have been initiated earlier on. Also, both parties and account for the time period required to receive each mple. The Contractor had clearly underestimated the time edd to provide samples in the programme. This item was scribed as design build in the BOQ. There was sunderstanding regarding the description as the ntractor considered the internal skin as a design build stem to meet specified load whereas the Contractor nsidered the fritting on the external skin as an aesthetical m to be chosen by the Engineer and not a design build stem. It was the Engineer's understanding that both the ternal skin and external skin are design-build. The gineer did not request full range of samples at the first bmission to save time. Moreover the notification of possible ay was rejected although this issue had become critical to a: project and was delaying the works and an extension of ne was later granted for this delay. eas of risk (as categorized by Zack, 1996) identified i The Engineer was late in reply thingly responses The process of providing the 4 timely responses							
the project and was delaying th	Assessmente of newy effect							
time was later granted for this								
Areas of risk (as categorized	by Zack, 1996) identified in	r thís case analysís:						
Delay and Disruption Untimely responses Interpretation of requirements	The process of providing the 4 s The Engineer was late in replyi The responsibility of designing allocated clearly	amples took time ng to the submittals the frit glass pattern was not						
Behavioral Observations:								
 The Engineer insisted responsibility. Both parties were late 	l on the fact that the delay wa in attending to this subject.	s the Contractor's						

Figure 23 Sample Analysis Sheet for Dispute Case

For each of the 50 dispute cases, a chronology of events was prepared and the analysis sheet was filled. The chronology of events and the analysis sheet for a sample dispute case is shown in Figures 22 and 23. The chronology of events for the 50 dispute cases is included in Appendix G. The analysis sheets are included in Appendix H.

As such and to graphically depict the observation made in this analysis, the interrelationships between the risks, disputes and the behavioral attitudes are drawn in Figure 24. The risks eventuating on a project pass through a filter. Risks that are not captured by the filter will turn to a dispute factor. The soundness of the filter which describes the project environment is defined by two criteria:

- the clearness/soundness of the Contract Documents including Conditions of Contract
- the behavioral attitude of the project participants that involve not only the site personnel but also the higher management that have an influence on the works on site.

It was noticed that a 'behavioral risk' by itself did not promote a dispute but contributed to the dispute formation. However, although a 'contractual risk' is needed as a primary dispute factor, the 'behavioral risk' had a wider effect as it contributes to a bigger number of dispute emergence. In the case of Dispute 14 (Façade aluminum colour), the primary contractual risk that eventuated was the "missing specifications". This was the contractual risk. However, the "trial and error" approach used by the Engineer as opposed to clearly specifying the requirement emerged as a secondary behavioral risk. Had the project environment directed the participants behavior towards adopting a proactive approach in specifying requirements, the matter would have been resolved. In this case both risks seeped through the filter to become a primary contractual dispute factor and a secondary behavioral dispute factor. It should be noted that the trial and error approach contributed as a secondary behavioral factor in 6 other disputes on the same project.

ראכא סר כסתתו	l l				
ate staupstan	·····		٦		
seligence an					
ate steupsbeni	1				
porce of cont					
HE 130338 SHE					
mon nous ne V					
ineth ciency an					
annatence					
איסנ שעק כואון					
requirements					
Failure to com		- 2			
Programming 1	Work Quality	/			səpen
Theft and Burg	Site safety	1 /			Contractor poor coordination between
Human error		1 /			Late intervention of subcontractor
məT əvitəfəd					Slow attendance to responsibilities
g 'aouapisqug	Permits and licenses				stettingen zeizeimistellicheten.
tiz neszerolaU	noitsuration	$\vdash P$	ł	\rightarrow	
Mechanical ar	fo shorts and a sussimily				
Lefective desi	Interpretation of requirements				Poor management (precontract)
Defective worl	Defective contract documents				ρλ Ευβιαεει
Defective desi	C oordination				Lack of Proper Management Variation
ussimmoo bra	edupment/matenals				Assessment of war claim
Dangerous sub	C ontractor-furnished			-	Validity/Assesment of Vanation
Removal of Si			i		
		K			
amn nantiai xo	Untimely responses				
(automdooor	Suspension of work				סרטפנ שיורא
vtilidetgenA	Early use of facility	\rightarrow			Evading responsibility by blaming the
sersvad, itorur do hruorzishi U	Delays and disruptions	\rightarrow		\rightarrow	the other party's intentions
Unforeseen Si	ACCELETELION				
arti bras sa xa T					
Transit to site				\rightarrow	improper communication channel
Financial stab	Project funding				I usi and error attempt# sproach
Internal stabil	Failure to pay	\rightarrow			א עו א א א א א א א א א א א א א א א א א א
idsiz lemsi xA	Cost Escalation	\rightarrow		\rightarrow	Lack of experience
Cyclore, Earlin	Contract Termination				Contractor avoiding monetary losses
storm, Hurrican	procountactor, supprise ratius			\rightarrow	Failure to notice the technical problem
C) bo D to sto A					ettimdu2
Lack of safety				-	Permits regulations
aq ətsupəbanl	Defective work	\rightarrow			Price escalation
Use of unteste	Weather		ł	\rightarrow	Contractor failure to satisfy specification requirements
Failure to tak	steoos etis	\rightarrow			responsibility/obligation
Lack of know	Quantity variations	l			Clear allocation and highlighting of
State of the ar	ratent atte conditions	7	Γ	\rightarrow	Not achervable requirements
Negligence a	Limorenduit Gumonduit				כובוההנפונים או שוווספ ניסויניים
stenqorqqenI			1		
សុធា, ពាលខែស	Force Majeur	ſ	/		הפאצט פונטו

Whet Choice of Professional Team
sek of communication
nadequate ate supervision
egugence and lack of care
nadequate ate management
uo moissei emokim unun neus lagare auto
באינים אינים איני
קוסני שות בז און בסווויניסול אוזאר לווא אוויאו או אווייי אוא אוויייייייייי
a suite to comply with insurer's conditions and
I heft and Burglary, Illegal activities, Faud and infidelity
Human ettor
Defective Temporary works and their deagn Conogon
סמסמתבווכבי בעלהוסמטו אות עובי א וסואורסט אות סכנווואורסט)
Untoreseen ale conditions (Ground movement,
ואובנטפתוכפו פוטק פופכתוכפו סנפפר קסאט
ערוברת אב מבאלא אמוצוו מואדול מוח להמווא במווחט
Defective workmenning and material
Defective design
Removal of Support
Technical Complexity and new methods
Txtended duration of construction
Acceptability of projects by locals
Underground obstructions)
runoff, Adverse geological and underground characteristics,
Unforescen Site Conditions (Topostaphy and auface water
Taxes and the Stability of the Legal System. Red tape
Transitio ate
Financial stability
Internal stability of Government
External stability of Government
Cyclore, Earlinguske)
Acts of U od (Excessive Kautal, Flood and mundation, Wind and
ד קרא טו באדע לא בארמיו טווא
frampina to and motionaber[
Use of untested and proven techniques
Failure to take account of foreseen problems
Lack of knowledge and checking
State of the art codes and knowledge
Negugence and lack of care
urapproprate cnoice of design
Adequacy of finance and related aspects
Adequacy of surveys and inspections
Adequacy of site investigation Adequacy of surveys and inspections
Choice of ate Adequacy of ate investigation Adequacy of surveys and inspections

Work Quality											>								-		~
viste safety	\$				>														Ì		Ē
Productivity									>				>								~
seare oil bus stimmed	1								-				Ī						5		2
noitourianoo fo sbodiam bus ansaM							-						<u>+</u>	1	>						÷
atnamatupat lo nottatanta		>		>			>	>		>	1	>		>	İ	>		>	5		10
atnamuoob taattoo avitaafaO	>	>	5			>	>	>								>	>		>		G,
Coordination	>						>									>	>	>	>		છ
e qui pri ent/materi al s																					0
Changes	>	>	>			>	>	>		>	>				>	>		>		>	12
a9200q291 YlaminU	>	-		>			>	>		>	>	>		>	>	>	>	>		>	13
Arow fo noisinsquuz					>																Ŧ
E arly use of facility																					0
ano itqure ib bas evelo	5			>	>		>	>	>		>	>		>	>	>		>	>		÷
Acceletation	1																			_	Ō
sə xe T																				_	0
Project funding									_							-					0
Failue to pay																					0
Cost Escalation		>	>										>						_		~
C ontract T ermination																					0
Subcontractor, supplier failure																_					2
	>						>		_												
zeroi forces					>				>				>								<u> </u>
Defective work																					_
Veather																					<u> </u>
Site access																					°
Quantity variations						>															<u>—</u>
Latent site conditions																					0
Impracticality/impossibility	>																				
Force Majeur					>				>				>								ŝ
	Lift Overhead	Façade False Ceiling	Procurement of new material	Façade glass	War Effect	Add. Shop dwgs. for agreed late variations	Façade Lighting	Frit Glass	Employee Turnover	Safety Film	False Ceiling	Lighting Fixtures	Increase Labor Rate	Façade Alum. Color	Steel Structure Design	Trench Heaters	Shop drwg. Submittal schedule	Marble Works	ledu	PS for Health cub	CASEA
	1 4	2	4 0	ব ক	<u>حر</u> ای	4	4	4	<u>₹</u>	4	ব ল	2	ୟ ന	4	2 2	₹ 9	A N	₹ 0	<u>₹</u>	A D	
					1					-	-1	-	-	-	-	-	-	-		3	

Work Quality											ō			:								8
Tite safety						>					-	Γ	Ī			>						=
Productivity						>			-		-					>						—
Remits and licenses											0											0
											0					-	-	-				0
atnemetation of requirements	>	>	>	>	>		>	>	>		8	>	>	>	>						>	ŝ
	>		>			:	>	>	>		5	>					>	>	>		>	5
Coordination											0										>	1
equipment/materials											0											0
Changes	>		>	>	>		>	>	>	<	8	\mathbf{i}	>	>	\mathbf{i}						>	5
Untimely responses	>			>	>			>	>	~	6						>	>	>	>		+
Suspension of work						>					1					>						— —
Early use of facility											0											0
Delays and dizuptions						>		>	>		3	>				く	>	>	>	\mathbf{i}	>	7
Acceletation											0											0
sə xs T											0											0
Project funding											0											0
Failure to pay											0											0
Cost Escalation			>								Ţ		>									÷
Contract Termination											0											0
Subcontractor, supplier failure											0											0
Labor forces						>			_		-					>						Ŧ
Defective work		>									÷				_							0
Weather											0											5
szecces											0											0
Quantity variations											0											5
Latent site conditions											0			>							>	2
Impracticality/imposalbility											0											0
Force Majeur						>										>						Ŧ
	3 Rejected NPVs	Tree Damage	Optional Works	Cleanouts	Fire Alarm System	War Delay	Soft Landscaping	Smoke detectors	Brire Fighting System	Towel Dryer	CASEB	Louvers at Roof	Chillers	Restoration	Glass Balustrade	War Effect Delay	External Works	Stone Flooring	Suhmittal Schedule	Alum. Composite Parel	Points of Drainage	CASEC
		2	m	4	- S	9	7	8	6	<u>–</u>	-	-	7	с	4	2	6	신		허	늵	
I	, I					1	l				i I	, I	ļ					1		, I	-1	

Work Quality	Ţ										C	~
visite safety	; [10	m
				1			-	1			0	7
səznəəil bus stinnə ^q	i T										0	2
Means and methods of construction	[İ		1				Ì	0	-
stnamanupar lo notistarquata		>	>	>					İ		~	26
efective contract documents	5	>						>	5	5	5	24
Coordination								5		-	F	8
s ls in ste m't na mq iups	Γ							Ī			0	0
Changes	>	>	>	>	5	>		>	>	-	8	33
Untimely responses		>	>		>			-		>	-	27
Zuspension of work									!		0	3
E ady use of facility					-						0	0
ano itquis ib bus systed	>	>	>	>	>	>	>	>		>	6	32
Acceletation											0	•
zə xs T											0	0
Project funding											0	0
Failure to pay											0	0
Cost Escalation											0	ŝ
Contract T ermination											0	0
Subcontractor, supplier failure											0	2
Labor forces							>				—	ې
Defective work											-	_
Weather											°	_
Site access											0	<u> </u>
											0	_
Latent zite conditions							_				°	\sim
Impracticality/imposability											0	—
Force Majeur							>				-	ي
	1 D Access to Roof Area	2 D Raised Flooring	3 D Additional Fireplaces	4 D New Kitchen	5 D Aluminum Works	5 D lighting works	7 D Lahor Decrease	BD EDL room	9 D Fire Rated Doors] [] Late Issue of Drawings	CASED	
		` '	1		~'	-			- 1	Ξ		

As shown above the 50 disputes from 4 different project were categorized by the areas of risk identified by Zack (1996) in Table 7. The areas of risk from Table 10 are listed in what follows in the order of frequency of occurrence:

Risk Type	Occurrence
Changes	33
Delays and disruptions	32
Untimely responses	27
Interpretation of requirements	26
Defective contract documents	24
Coordination	8
Force Majeur	6
Labor forces	6
Productivity	5
Cost Escalation	4
Work Quality	3
Suspension of work	3
Site safety	2
Subcontractor, supplier failure	2
Permits and licenses	2
Latent site conditions	2
Defective work	1
Quantity variations	1
Means and methods of	
construction	1
Impracticality/impossibility	1

Table 10 Risk Type Occurence

Other risks mentioned by Zack such as site access, weather, contract termination, failure to pay, project funding, taxes, acceleration, early use of facility, and contractor-furnished equipment material were not witnessed in these case studies. The top 5 risks with the highest frequency of occurrence are described below:

Changes: The risk with highest occurrence is "changes" where the Contractor's entitlement to additional cost is disputed. This is the case in 13 out of the 20 disputes in project A, 8 out of 10 disputes in project B, 5 out of 10 in project C and 8 out of 10 in project D. These changes result in some cases to delays, the assessment of which is

disputed. Also, as seen in the cases, the changes might result in abortive works the value of which is again disputed.

Delay and Disruption: Delay and disruption is a risk raised in 13 out of the 20 disputes of Project A. 3 of 10 in Projects B. 7 of 10 in Project C. and 9 out of 10 in Project D. These can be divided into 2 categories: allocation of responsibility of delay and assessment of the delay effect. Allocation of responsibility of delay as in the case of dispute 1 from Project A where the lift overhead problem required more than 6 months to be resolved technically and where both parties share responsibility of this delay as the contractor failed to submit coordinated shop drawings and the Consultant failed to notice the error in calculation and its nonconformance to the permit requirements. Assessment of the delay effect is yet another area disputed as in the cases of war effect, the Contractor would attribute the delay due to the loss of labor to the incidence of war whereas the Employer would argue that there were other factors that further augmented this delay. It is worth noting that the political situation in the country and the hostile events that resulted caused a disputed delay claim in all four projects. Assessment of delay is also a dispute factor in the case of delay by the Engineer to provide further technical details where the Engineer might consider it part of the Contractor's obligation to investigate the market and find the item that both meets the specification requirements and is fit for purpose as in the case of the frit glass in dispute 8 of project A, safety film in dispute 10 of project A, and lighting works in project D.

Untimely response: The third recurring risk among the cases is the untimely response where the Engineer is late in providing the missing/additional information required by the Contractor. This is the case in 13 out of the 20 disputes in project A, 7 out of 10 disputes in project B, 4 out of 10 in project C and 4 out of 10 in project D. A total of 28 out of 50 cases shows that again more than 50% of the time there was a problem with untimely response. This delay was mainly critical in the cases of replies to submittals and replies to notifications of possible variations.

Interpretation of requirements: The fourth in order of recurrence is the "IInterpretation of requirements where both parties disagree on the reading of the contract requirements. This is the case in 10 out of the 20 disputes in project A, 8 out of 10 disputes in project B, 6 out of 10 in project C and 4 out of 10 in project D. A total of 28 out of 50 cases which is more than 50% of the time. These include cases of disputed reading of liability in verifying the functionality/performance of the trench heater design or in designing an aesthetic item such as façade frit glass.

Defective contract documents: This risk is of equal recurrence as the risk of "Interpretation of requirements. This is the case in 9 out of the 20 disputes in project A, 5 out of 10 disputes in project B, 6 out of 10 in project C and 8 out of 10 in project D.

In Table 11, the disputes factors on each of the 50 dispute cases are identified. These are listed in the order of frequency of occurrence in Table 12.

contractor poor coordination between trades																		X				×		~
ate intervention of subcontractor	I ×				×				×															~
seitilidiznoqzat ot senabratts wold	\$				×	×			×			×		×		×	×			×	×	×	×	1.1
Contractor late/missing submittals						Ì			×	×			×	×						×	×			9
Late approval of submittals by Engineer	(1			×				×				×	×		×	×	×		×	×			6
Velab to the masses A	'				×	×			×	×			×	×		×	×	×			×	×		Ļ.
Lack of Proper Management/Monitoring by Engineer					×			_	×	×							×	×		×		×		∞
Assessment of war claim	·	-				×					×				×					_			_	3
V alidity/Asseament of V an ation	×	-		×		<u>.</u>		×		×							×		 		×		×	r
Late issue of missing design/ran ation	:	<u> </u>	<u> </u>				<u> </u>					×	×								×			~~
Influence by the employer		×				×		×	×												_		×	رم ا
Engineer firm although contract is grey		×		×	×			×		×	×			X									-+	2
party		<u>.</u>						_														_	-	
Evading responsibility by blaming the other	ľ							_		×		×						×		×	×			6
L anguage expressing ill perception of the other party's intentions																								0
9tuqa ib 9vlo291 ot 229qq illiwaU								×																1
Lack of cooperation	×													_								Ť	1	-
Iaproper communication channel																						\neg		•
Trial and error attempts/aproach					×					×		×	×			×				Ť	×	_	1	6
Human error/negligence	×									×										1				2
Lack of experience	×				×			_	×	×		×	×			-						×	T	7
C ontractor avoiding monetary losses				×				×		×	×				×									5
Failure to notice the technical problem	×																	×					1	2
subatist is stranduz				×																×		Ť	Ţ	2
Permits regulations	×														_							×	1	2
Price escalation			-	×											×		_	_				-+		~
Contractor failure to satisfy specification requirements										-				×				_						
responsibility/obligation					×									_	×			×					-+	3
	×								<u></u>	_	-				_					_	_	+	-+	3
		×		×	×			×		×	_			_				~		$\frac{1}{2}$	-		+	
	×												-					×						3 3
								_						_			E				-	_		
	Lift Overhead	Façade False Ceiling	Procurement of new	material	Façade glass	War Effect	Add. Shop dwgs. for	agreed late variations	Facade Lighting	Frit Glass	Employee Turnover	Safety Film	False Ceiling	Lighting Fixtures	Increase Lahor Rate	Façade Alum. Color	Steel Structure Desig	Trench Heaters	Shop drwg. Submittal	schedule	Marble Works	EDL	PS for Health cub	CASEA
	A	A		₹	₹	4	•	₹	≤	<u><</u>	₹	Ā	4	₹	₹	∢	4	4		<	<u>र</u>	হা	₹	
	-			יי	4	-	(الع	\sim	ω	m	믭	=	믭	С	4	9	16	. !	\geq	뛰	<u>ម</u>	別	

											-		_	_								
Contractor poor coordination between trades											0										×	-
Late intervention of subcontractor											•					ļ				-		0
Slow attendance to responsibilities	5										0		×	×			×	×		×		5
contractor late/missing submittals					-						0	×										-
Late approval of submittals by Engineer									×	×	2						×		×	×		~
Asish to transser						×			×		~	3					×	×	×	×	×	0
Engineer											0	×										_
Lack of Proper Management/Monitoring by			<u> </u>		_	-	<u> </u>															
Assessment of war claim					<u> </u>	×										×						-
V ali di ty/Assement of V an ation	×	×	×	×	×	<u> </u>	×	×	×	×	6,	×	×	×	×			×			×	ي
Late issue of missing design/variation											0						×	×		×		~
Influence by the employer					<u> </u>					×												2
Engineer firm although contract is grey			×				×	×	×		-+		×									
Pading responsibility by blaming the other Pady	×	×	×	×	×			×	×	i	2	×		×					×		×	-+
Language expressing ill perception of the other party's intentions						×	×				2								X			
Unwillingness to resolve di spute		×					×				2			_								0
Lack of coopstation											0										-+	0
Inproper communication channel											0										\uparrow	0
Trial and error attempts/stgmests rome bus lainT											0										1	0
Human error/negizence		X																			+	0
Lack of experience											0	×	İ									-
Contractor avoiding monetary losses	×	×	×	X	×		×	×	×	×	6,	×	×	×	×							-+
Failure to notice the technical problem										-	•										\uparrow	0
subadaz tettimduZ											0	_							×			, -
Permits regulations											•											0
Price escalation			×										×									
Contractor failure to satisfy specification requirements				×	×			×	×		-+	×									×	~
Ciest allocation and mgmignung of responsibility/obligation	×		×	×	×			×	×		9			×							×	2
Not achervable requirements											0							_			+	0
Contract Document unclear	×	×	×				×					×		×						_	×	~
ຼ ຖຣສຣີນ ຣແວເ	×										÷	_									×	2
· · · · · · · · · · · · · · · · · · ·																				el		
	Rejected NPVs	Tree Damage	Optional Works	Cleanouts	Fire Alarm System	War Delay	Soft Landscaping	Smoke detectors	Fire Fighting System	Towel Dryer	CASEB	Louvers at Roof	Chillers	Restoration	Glass Balustrade	War Effect Delay	External Works	Stone Flooring	Suhmittal Schedule	Alum. Composite Pan	Points of Drainage	CASEC
				m	m	œ	m	m		Ξ		$\frac{1}{2}$	<u></u>	U U	С	<u>с</u> і	<u>ں</u>	<u>с</u>	с U	ပ	ပ	
		2	က	4	ហ	۵	~	œ	တ	9		-		സ	4	Ъ	۵	\sim	ω	တ	9	

	1 D Access to Roof Area	2 D Raised Flooring	3 D Additional Fireplaces	4 D New Kitchen	5 D Aluminum Works	6 D lighting works	7 D Lahor Decrease	8 D EDL room	9 D Fire Rated Doors	10 D Late Issue of Drawing	CASED	
Dealgn error			<u> </u>							E.A.	-	
Contract Document unclear					<u> </u>		×	×		×	3	7 11
Mot acheivable requirements											0	8
Clear allocation and highlighting of responsibility/obligation											0	11
Contractor failure to satisfy specification requirements											0	7
Price escalation										_	0	-+
גרנשונג ופנשונג ופנשמט אין אין אין אין אין אין אין אין אין אין	-					_				—	0	5
maldore leaindaat adt aaitor of antijee											0	~
Contractor avoiding monetary losses												
Lack of experience												8
Human error/negligence		×									-	-+
Trial and error attempts/aproach											•	9
Improper communication channel										×		
Lack of cooperation					×	×					2	~
Unwillingness to resolve dispute											•	~
L anguage expressing ill perception of the other party's intentions					×						÷	-+
Evading responability by blaming the other party											0	17
Engineer firm although contract is grey				×								13
Influence by the employer				×								r
Late issue of missing design/variation	X	×	×	×	×	×	_		×		~	13 2
V alicutyr Assesment of v an auon	x	×	×	×		×	_	×	×		-	5
Lack of Proper Management/Monitoring by		×	×		×	×					-+-	
אראפגנגש שטו טנ קשו אייי דעצועפנ		Â	Â			_			_	Â	_	2
The second state and a lettimeter of the property of the second state of the second st		×		J	×				~	-	~	€
rounding to statimate to be to the to the total of total of the total of the total of tot											-	2
Slow attendance to responsibilities		×	×		×	×			×	×	6	22
Late intervention of subcontractor											0	~
Contractor poor coordination between trades											0	3

The top 5 dispute factors with highest frequency of occurrence from tables 12 are: validity/assessment of variation, assessment of delay, slow attendance to responsibilities, contract documents unclear, contractor avoiding monetary losses, late approval of submittals by Engineer. They match with the top 5 risks with highest frequency of occurrence. This further confirms the observation made earlier and depicted in Figure 24 that disputes are the offspring of the risks trespassing the filter. The 5 dispute factors with highest occurrence are further described below.

Dispute Factor	Occurrence
Validity/Assessment of Variation	29
Assessment of delay	26
Slow attendance to responsibilities	22
Contract Document unclear	18
Contractor avoiding monetary losses	18
Late approval of submittals by Engineer	17
Evading responsibility by blaming the other party	16
Lack of Proper Management/Monitoring by Engineer	13
Engineer firm although contract is grey	13
Late issue of missing design/variation	13
Clear allocation and highlighting of	
responsibility/obligation	11
Lack of experience	8
Design error	7
Contractor failure to satisfy specification requirements	7
Influence by the employer	7
Contractor late/missing submittals	7
Trial and error attempts/approach	6
Assessment of war claim	6
Price escalation	4
Human error/negligence	4
Language expressing ill perception of the other party's intentions	4
Not achievable requirements	3
Submittal schedule	3
Lack of cooperation	3
Unwillingness to resolve dispute	3
Late intervention of subcontractor	3
Contractor poor coordination between trades	3
Permits regulations	2
Failure to notice the technical problem	2
Improper communication channel	1

Table 12	Dispute	Factor	Occurence
----------	---------	--------	-----------

Validity/Assessment of Variation is the dispute factor with the highest occurrence. 29 cases out of the 50 are related to disagreement regarding the validity or assessment of the variation order. In some cases disagreement regarding the validity is related to contract documents being unclear. In other cases it is the valuation of the variation that is being disputed.

Assessment of delay is the dispute factor with the second highest occurrence. As mentioned above it is more prevalent in projects A and D. This dispute factor has been correlated with he dispute factor "Late attendance to obligations by both parties" that will in most cases lead to disputed assignment of responsibility and assessment of those delays.

Slow attendance to responsibilities: these are cases of delay that do not have contractual timelines but even where they are not critical they would be consuming the float allocated to the respective activities and any delay thereafter becomes a critical delay. This is confirmed by the high correlation with the dispute factor 'Assessment of Delay'. It is evident in Projects A, C and D but did not occur in Project B.

Contract documents not clear: Two cases of unclear contract document include ambiguity in the specification of technical requirements such for the external façade detail in cases 4 and 8 of project A and louvers at roof in case 1 of project B. Other cases of ambiguity relate to unclear description of the allowed number of submittals per week, the method of calculating the additional cost of new material procured as per late instruction and time validity of optional works included in the contract.

Late approval of submittals by Engineer was noted in 16 cases. Although the Engineer in many of these cases referred to the Contractor's failure to submit/abide by the submittal schedule, this does not relieve the Engineer from his/her responsibility to fulfill the contractual obligation in the best interest of the project.

Contract Administration	Valuation of Variations	Late issue of missing design/variation	Valuation of Variation	Assessment of war claim	Assessment of EOT/ Monitoring of Delav	Poor Management/Monitoring	by Engmeer Assessment of delay		Timely response to contractual matters	Late approval of submittal by Engineer	Contractor late/missing submittals	Slow attendance to responsibilities	Contractor's Expertise in Management	late assignment of subcontractor	Contractor poor coordination between trades
Behaviour	Key Persons Assigned and Adequacy of staff allocated	lack of experience	human error/negligence	Trial and error	Proactive attitude in resolving problematic issues	Improper communication channel	lack of cooperation	disputes	Language expressing bad relation	Evading responsibility by blaming the other party	Engineer firm although	contract is grey	Employer's representative (scope of work,)	Influence by the employer	witnessed in the Dispute Cases
Risk	Risk Allocation (Best Practice/ Common	Practice)	Clear risk allocation and highlighting	Contractor failure to	Sausty requirements Permit regulations	Price Escalation	Submittal schedule	Contraction of allowers of for	contractor s anowance jor risk premium/ Contractor's Risk Allocation	Failure to notice the problem	Contractor avoiding monetary losses				Figure 25 Dispute influencing areas
Tender	Contract Documents		design error	Contract Document unclear	Not achievable	requirements	contract negotiation								

6.3.Dispute Influencing Areas

The dispute factors are categorized in Figure 25 under the dispute influencing areas examined previously. This further confirms the impact of these areas on disputes as witnessed in the dispute cases.

6.4.Summary

This chapter was devoted to examine 50 dispute cases on four ongoing projects. For each of the dispute cases the chronology of events that gave rise to the dispute was listed. An assessment sheet was filled where the risks that eventuated, the behavioural attitudes and the dispute factors were tracked. An interrelationship is drawn between those 3 aspects of risks, disputes and behavioural attitude. Also, the risks and the dispute factors with the highest occurrence were examined. The dispute factors identified were categorised under the dispute influencing areas already identified through the interviews in Chapter 5.

It is worth noting that not all dispute influencing areas that surfaced in the interviews were witnessed in the case studies. The reasoning behind this is that the analysis of the 50 case studies was an in-depth analysis of disputes occurring during the execution of the works. As such, the influence of factors such as the tender period and the tender evaluation technique on disputes could not be tracked. Similarly, no inference could be made of the Employer's influence on risk allocation and the impact of the Forms/Types of Contract adopted including the Particular conditions. The dispute resolution mechanism was again beyond the scope of the case study analysis as it necessitates examining the project over a longer period.

CHAPTER 7: THE FIDIC FORM OF CONTRACT

7.1 Introduction

The FIDIC 4th Edition of the Red Book is the form of contract used commonly in Lebanon as witnessed from the fieldwork conducted in this research. The abbreviation stands for Federation Internationale Des Ingenieurs-Conseils and is a national association of Consulting Engineers. It has been in existence since 1913 and have their headquarters and secretariat in Lausanne in Switzerland. FIDIC has produced standard forms of contract for civil engineering projects since 1957. The second edition was published in 1969 and the 3rd in 1977.

7.2 Conditions of Contract

The FIDIC conditions of contract comprise two parts: Part 1 General Conditions and Part 2 Particular Conditions.

Previous empirical studies on the sources of disputes also reveal that the conditions of contract always appear in the analysis and that certain conditions of contract contribute to a higher frequency of disputes (Fenn *et al.*, 1997).

Particular Conditions refer to the changes made to the General Conditions of Contract. Corbett (1991) advises that 'great care' must be taken when amending the general conditions. He explains the danger of amending those conditions as follows: "These FIDIC conditions are generally well-balanced and, as with any contract, there are a great number of links and relationships between different clauses, not all of which are express or otherwise obvious. With any amendment, therefore, there is the danger of upsetting the balance or of creating unintended consequential changes to related provisions. It is in the interest of all parties that changes should be kept to minimum."

The Contract assigns duties and obligations to the parties involved. The Engineer is expected to have a double duty under the contract. The Engineer is expected to be the agent for the Employer in ensuring that the Contractor satisfies contract requirements. The Engineer is also expected to be impartial in contract administration and dispute resolution. The Engineer's duty to have due consultation meetings with both parties on 25 occasions regarding time and cost prior to requesting Engineer's Decision by any of both parties. Consultation in this case is intended to provide for a clear sign of Engineer's impartiality. According to Corbett (1991) "impartiality depends upon the ability of the Engineer to exclude from that part of his mind which is making a determination under the contract all considerations other than those required to achieve a fair decision in accordance with the spirit of the Contract."

The Employer may add restrictions to the Engineer's authority where the Engineer would be required to seek his approval before exercising such authority. However, these restrictions should be carefully set as the Engineer has a role as an adjudicator and should exercise his discretion impartially. Also, 24 sub-clauses require the Engineer to conduct due consultation meetings with Employer and Contractor before cost assessments and Extensions of time are determined. As such the Employer under the general conditions should be duly consulting with the Engineer and not dictating him/her (Bunni, 2005).

The obligations of the Employer's representative as set in the FIDIC General Conditions of Contract 4th Edition include the following (Corbett, 1991):

- 1. Appointment of the Engineer
- 2. Give possession of site
- 3. Refrain from taking any action that would impede or interfere with the progress of the works
- 4. Supply materials and carry out works if these form part of the work as defined in the contract
- 5. Nominate specialist sub-contractors and suppliers when they are required
- 6. Make payments on time
7. Among other bligations give consent where the contractor wishes to assign the contract or any part thereof.

Knowles (2005) in his book "150 Contractual Problems and their Solutions" raises 150 contractual questions that are normally disputed. By examining these questions and in some cases the controversy among awards made addressing those disputes, it becomes evident that by setting the general conditions the forms of Contract do not attend to numerous particular questions that might surface during the execution of the contract and that might eventually lead to disputes. Such potential dispute areas surfaced in the interviews and case studies.

7.3 Dispute Influencing Clauses

Since the FIDIC Red Book 4th Edition was used in most of the projects in the fieldwork conducted, a brief overview of the corresponding dispute influencing clauses as witnessed from the interviews and the dispute cases is addressed in what follows:

 \Box Price escalation

Clause 70.1 Increase or Decrease of Cost: There shall be added to or deducted from the Contract Price such sums in respect of rise or fall in the cost of labour and/or materials or any other matters affecting the cost of the execution of the Works as may be determined in accordance with Part II of the Conditions.

Part II of the Conditions describes 3 alternative methods for dealing with the fluctuation issue: no adjustment, adjustment based on difference of prices between base prices and the current prices, adjustment based on indices to a formula. FIDIC advises that projects spanning for more than a year should contain fluctuation clauses based on the view that the Employer should take the risk of fluctuated prices as opposed to asking tenders to quote fixed rates that account for possible increased cost. In civil code countries the *Theorie de l'imprevision*, that allows for reducing the Contractor's losses to reasonable limits through compensation by the Employer, would be applicable in cases where the fluctuation clause has been deleted. However, the Contractor's claim in this case is prone to be disputed (Corbett, 1991).

Clause 70.2 Subsequence Legislation:

If, after the date 28 days prior to the latest date for submission of tenders for the Contract there occur in the country in which the Works are being or are to be executed changes to any National or State Statute, Ordinance, Decree or other Law or any regulation or by-law of any local or other duly constituted authority or the introduction of any such State Statute, Ordinance, Decree or other Law or any regulation or by-law which causes additional or reduced cost shall, after due consultation with the Employer and the Contractor, be determined by the Engineer and shall be added to or deducted from the Contract Price and the Engineer shall notify the Contractor accordingly, with a copy to the Employer.

This clause of the general condition is clear in allocating the risk of changes in legislation to the Employer is the one taking the risk of changes in legislation. According to Corbett (1991), the Contractor would be taking 'grave' risk if he accepts amendments to this clause. This clause is considered to be an equivalent to the doctrine of *Fait du Prince* where the Contractor is fully compensated is he has suffered increased costs or losses due to a change in the law.

□ Procedure for notification of possible variation and time barring.

Clause 52.2 Power of Engineer to Fix Rates:

Provided that no varied work instructed to be done by the Engineer pursuant to Clause 51 shall be valued under sub-Clause 52.1 unless, within 14 days of the date of such instruction and, other than in the case of omitted work, before the commencement of the varied work, notice shall have been given either:

- a. by the Contractor to the Engineer of his intention to claim extra payment, or a varied rate or price, or
- b. by the Engineer to the Contractor of his intention to vary a rate or price.

This clause requires a notice of varied work within 14 days of receiving an instruction. The Clause has been varied from the 3rd Edition which required notice "as soon after the date of order as is practicable."A failure to give notice would lead to loss of entitlement claim. Similarly in clause 44 (Extenstion of Time) the Engineer should give notice for possible delay within 28 days. Under clause 53 the Contractor should give notice of claim for any additional payment within 28 days and substantiating the claim in a timely manner thereafter (Corbett, 1991). Brewer (1993), Kangari (1995) and the Wood Report (1975) stress the importance of having proper project activity documentation and factual evidence. The importance of preparing claims with required detail level and supporting documents should be recognized by contractors. In reality proper human expertise that would ensure continuous attention to claims-related matters is not allocated.

Design Liability:

Although FIDIC gives guidance for amendments of the Red book to 'Lump Sum' form. According to the Guide The 'Lump Sum' form is to be used on projects where the design has been developed by the Employer to a sufficiently complete stage where from the information supplied the Contractor can prepare all drawings and details necessary for construction without having to refer back to the Engineer for clarification or further information. However, the intention of this lump sum form as presented by FIDIC is for works which are simple and straightforward of relatively low value. For larger works it is recommended that the FIDIC Conditions of Contract for Design-Build (which is also a lump sum form) be used. This recommendation is being disregarded in common practice when a multimillion dollar project is awarded on Lump Sum basis.

- Employer's prior approval to any Engineer's time and cost determination
 Clause 2.1 Engineer's Duties and Authorities: of the General Conditions:
 (a) The Engineer shall carry out the duties specified in the Contract.
 - (b) The Engineer may exercise the authority specified in or necessarily to be implied from the Contract. provided, however, that if the Engineer is required, under the terms of his appointment by the Employer, to obtain the specific approval of the Employer before exercising any such authority, particulars of such requirements shall be set out in Part II of these Conditions. Provided further that any requisite approval shall be deemed to have been given by the Employer for any such authority exercised by the Engineer.
 - (c) Except as expressly stated in the Contract, the Engineer shall have no authority to relieve the Contractor of any of his obligations under the Contract.

As witnessed in the Particular Conditions of the four projects examined in Chapter 6, the following restriction was added to this clause:

Provided that the Engineer shall obtain the specific approval of the Employer before exercising any of the following duties or authorities:

- a) Approving an extension of the Time for Completion of the Works
- b) Approving any additional payment to the Contract Price.
- c) Giving consent to the Contractor to Subcontract any part of the Works pursuant to Sub-Clause 4.1.
- d) Issuing an instruction to suspend works pursuant to Sub-Clause 40.4
- e) Issuing an instruction in respect of any Provisional Sum pursuant to Sub-Clause 58.2.

The requested Employer's approval for Engineer's assessment in granting an Extension of Time and additional payment contradicts with the Engineer's role under the contract as an impartial contract administrator.

□ Order of precedence of contract documents.

Clause 5.2 Priority of Contract Documents:

The several documents forming the Contract are to be taken as mutually explanatory of one another, but in case of ambiguities or discrepancies the same shall be explained and adjusted by the Engineer who shall thereupon issue to the Contractor instruction thereon and in such event, unless otherwise provided in the Contract, the priority of the documents forming the Contract shall be as follows:

- 1- The Contract Agreement (if completed)
- 2- The Letter of Acceptance
- 3- The Tender
- 4- Part II of these Conditions
- 5- Part I of these Conditions; and
- 6- Any other document forming part of the Contract

This sub-clause has been amended from the 3rd Edition to provide a full listing for the contract documents. However, as noticed in Appendix D on some projects the order of precedence is removed leaving the interpretation of contracting documents to the Engineer's sole discretion which in turn could become a dispute cause.

Engineer's Decision

Engineer's Decision: Clause 67.1 FIDIC states that if 'any dispute whatsoever' arises between the parties it can be referred to an Engineer's Decision. This means that the category of disputes to be referred to the Engineer includes breaches of contract. As such the Engineer under this clause is under obligation to become an impartial judge to take decisions towards his own actions or breaches or even the Employer's breaches. No arbitration may be instigated without an Engineer's decision. The Employer has no power to remove and replace an Engineer unilaterally. The reason being stated in the FIDIC's guide that the contractor in calculation of his tender took the identity of the Engineer as one of the criteria (Corbett, 1991).

The Dispute Adjudication board was introduced in the 1996 Supplement to the 1992 Fourth Edition of the Red Book where the decision-making role of the Engineer under clause 67.1 was reallocated to an independent impartial and neutral Dispute Board. The Dispute Board concept originated in the late 1960's with the Dispute review board which proved to be a success (Bunni, 2005). Gaitskell (2005) states that broadly around 97 percent of disputes raised to DRB are resolved before going to arbitration. Interesting enough, the author observes that an "unexpected dynamic develops" among parties working with each other on site as they view the DRB as intruders. As such when the DRB arrives on site the parties will put on a common front and resolve small disputes so that they do not have the DRB interfering with what is considered to them "the site's private business" (Bunni, 2005).

In 1995 it received major encouragement when the World Bank introduced a mandatory requirement of assigning a DRB in its standard bidding document for the Resolution of disputes. Unlike the DRB, the DAB is neither consensual nor amicable in nature. It is binding until the decision is revised in an amicable settlement or an arbitral award. Although it has not been implemented in construction contracts in Lebanon and although interviews had mixed views towards its implementation , the success reported would encourage introducing the DAB as a replacement to the Engineer's Decision under clause 67.1 (Bunni, 2005).

□ Assessment of extension of time

Clause 44.1 Extension of Time for Completion In the event of: (a) The amount or nature of extra or additional work,

(b) Any cause of delay referred to in these Conditions,

(c) Exceptionally adverse climatic conditions,

(d) Any delay, impediment or prevention by the Employer, or

(e) Other special circumstances which may occur, other than through a default of or breach of contract by the Contractor or for which he is responsible,

being such as fairly to entitle the Contractor to an extension of the Time for Completion of the Works, or any Section or part thereof, the Engineer shall after due consultation with the Employer and the Contractor, determine the amount of such extension and shall notify the Contractor accordingly, with a copy to the Employer.

Where a delay occurs on a project, the forms of contract normally state that the Employer shall determine the Contractor's entitlement for extension of time. However, there has been in the literature a list of different approaches to delay analysis. If the Engineer makes an analysis using one approach and the Contractor makes the analysis based on a different approach this will create a dispute. Also, in using the same delay analysis approach where there is a case of concurrency i.e. a delay event by the Employer that is running in parallel with a delay event by the Contractor, there is no hard and fast rule concerning which delay would be a dominant one. This could lead to specific situations where for example a delay by the Contractor would be delaying the works for 4 weeks and the Employer makes a change at the fourth week which only delays the project one day. There has been a good number of legal proceedings that on different projects have addressed this question from a different perspective and have thus passed different judgments. The fact that each case would be judged at its own merits and that each party will try to defend its best interest by supporting its stand by referring to these contradictory cases would by itself create an area of dispute that has two valid points. Yogeswaran el al. in reviewing principles applied in

extension of time claim explain that most general conditions of contract do not spell out clearly details of the principles that will be used for the assessment of claims for extension of time and this is left to the parties involved in each project.

The Society of Construction Law issued the Delay and Disruption Protocol in October 2002 was intended to be a useful guide in delay and disruption issues. It is meant to be a balanced and viable approach to resolve and avoid unnecessary disputes. The protocol provides that although standard forms of contract address the issue of delay and compensation for prolongation they do not do so completely by presenting a balanced view to a number of issues that "do not have absolute answers" and in most cases turn into a dispute that is often referred to a third party. The Protocol further provides a unified mechanism to assess delay and disruption matters. The AACE International Recommended practice published in 2009 again proposes a unified technical reference for forensic application of critical path method.

□ Valuation of fair rates

In study performed by Kumarsawamy (1997) in projects in Hong Kong variations and delayed site possession proved to be a frequent cause of conflict and disputes.

In the FIDIC Form Fourth edition, Valuation of Variation is dealt with in clause 52.1 of the Conditions of Contract:

All variations referred to in Clause 51 and any additions to the Contract Price which are required to be determined in accordance with Clause 52, shall be valued at the rates and prices set out in the Contract if, in the opinion, the same shall be applicable. If the Contract does not contain any rates or prices applicable to the varied work, the rates and prices in the Contract shall be used as the basis of valuation so far as may be reasonable, failing which, after due consultation by the Engineer with the Employer and the Contractor, suitable rates or prices shall be agreed upon between the Engineer and the Contractor. In the event of disagreement the Engineer shall fix such rates or prices as are in his opinion, appropriate and shall notify the Contractor accordingly, with a copy to the Employer. Until such times as rates or prices agreed or fixed, the Engineer shall determine provisional rates or prices to enable on-account payments to be included in certificates issues in accordance with Clause 60.

One common cause of disagreement is the difference of opinion towards the unit prices of resources necessitated by a variation order. Vidogah and Ndekugri (1998) emphasizes the importance of agreeing to some of the matters whose assessment are normally disputed such as percentage for main office overhead and unit cost of resources.

The principles of and approaches to valuation generally stem from established or common practice. Courts take a subjective view and hold that the rate must be "fair" in the ICE conditions or "appropriate" in the case of FIDIC conditions. Disputes normally arise often due to disagreement over the manner in which fair rates are to be calculated; whether prorated rates from existing unit rates of similar BOQ items can be implemented and on whether the Contractor is entitled to profit in these variations. In the case of Weldon Plant vs. Commission for New Towns (2000) for example, Weldon was contracted the construction of Dunston Mill Reservoir. The Contractor was instructed to excavate all the gravel below the bed and backfill it with clay to the design level. The Arbitrator decided that Weldon was entitled to the cost of these works but not to the overheads and profit based on the premise that "the amount paid should keep the contractor in the same financial situation he would have been in had the instruction not been issued" and that awarding Weldon would put him in a better position. Weldon appealed the issue to the courts. The courts held that a fair valuation has "ordinarily to include elements for cost of labour, the cost of plant, cost of materials, the cost of overheads and profit, otherwise it would not be a fair valuation within the meaning of the contract." This is an example where 'fair valuation' was defined differently by an arbitrator and a judge on the same project. The valuation of variation clause should be more detailed in the particular conditions.

7.4 Summary

This Chapter addresses the Dispute Influencing Clauses that surfaced from the fieldwork done through the preliminary examination of projects, interviews and case studies. This is done in the context of the FIDIC Red Book 4th Edition which is the Form of Contract commonly used in Lebanon. This examination reveals that with certain clauses care should be taken in applying amendments so as not to disturb the balance of the contract. Much risk is allocated to the Contractor where for example the Contractor's right to compensation in the cases of entitlement to price escalation and subsequent legislation. Again, similar high risk is allocated to the Contractor when the liability to proceed with execution of a lump sum contract is assigned to the Contractor where the contract design is not complete. Also in the cases where the particular conditions required the Engineer to have the Employer's approval to an Engineer's Determination contradicts with the FIDIC's definition of the Engineer's role of being the fair judge. Other clauses need further detailing in the Particular conditions namely the valuation of variation and the extension of time.

CHAPTER 8: CONCLUSION ON DISPUTE INLFUENCING AREAS

8.1. Dispute Influencing Areas

The Dispute influencing area identified in Chapters 5 and 6 are shown in Figure 26. Section 8.2 below analyses each of these areas in light of the interview and case study findings. Conclusion for each is drawn through comparison of the finding from the interviews, the case studies and the literature review.

8.2. Analysis of Findings

For each dispute influencing area, a comparison figure is drawn where the upper rectangle shows finding from the literature review, the rectangle on the right describes the observations from the case studies whereas the figure at the left includes the finding from the interviews. Comparison of the three boxes yields the lower box which is the conclusion.





RISK

• Risk Allocation (Best Practice/ Common Practice)

Risk must be allocated to the party that is best able to hande it. Modern projects have more complicated risks and due to the tough economic situation that has continually decreased the Contractor's profits. Risk allocation should be clearly recognized and communicated. Risk sharing was proposed as one possible option to achieve cost reduction. Dissimilar perception of fairness is a source of conflict that might result in dispute.

The interviews state that there is a tendency to allocate much of the risk to Contractors.

The quality of the modern projects is more complex with more intricate design that in many cases leads to discrepancies, missing details, and in some cases underestimation of the cost of specified systems at the bidding phase. Some interviewees considered the contract to be unfair where others stated that since the contractor has examined the tender documents and accepted them then the contract can not be considered unfair. All interviewees agreed that risk should be allocated to the party that is best able to handle it. Deviation from this rule were explained by Employer's desire to minimize cost overrun and safeguarding against Contractor's abuse of some clauses.

Looking at the case studies, it is noticed from the extract of the Particular Conditions shown in Exhibit D that risk is being shifted to the Contractor in the clauses examined

The risk should be clearly identified and better allocated and a relationship of trust must be developed between the parties. • Employer's Influence in risk allocation

The literature states that the Employer must have a clear understanding that allocating all the risk to the contractor will not reduce the actual cost of the project.

The interviewees stated that the tendency to allocate the risk to the Contractor is partly driven by the Engineer preparing tender documents and partly driven by the Employer who would try to reduce this risk exposure.

The influence of the Employer's interference in risk allocation is beyond the period examined in the case studies. As such no inference can be made regarding this dispute factor from the case studies.

Both the interviews and the literature confirm the influence of the Employer in risk allocation. The literature adds that there should be awareness by the Employer at Contract Award stage that the project is subject to time and cost overrun.

• Contractor's allowance for risk premium

There is a tendency to avoid contingency allocation in budget submissions not to raise the project budget too high. But this increases the risk of facing a crisis situation during the execution of the work if any of these risks arise. The Engineer should ascertain whether "the additional burden imposed on the Contractor can be carried out without the risk of financial failure".

The interviewees stated that moving the risk to the Contractor will mean including a high premiums to account for this risk which will lead to employer paying for the risk ahead of time. However, interviewees confirmed that the contractor doesn't allow for risk premiums to maintain the competitive edge which might lead to incurring substantial losses not accounted for when these risks arise. Looking at Projects B & C, it can be inferred that since the Contractors were reported to be awarded the contract at a rate much lower than the average, they had not allowed for the inherent risks. This becomes evident from the high occurrence of the dispute factors of 'Contractor avoiding monetary losses' and 'Validity/assessment of variation'. The Contractor's in Projects A and D were again claiming for risks/ambiguities that led to cost overruns. But no inference can be made regarding their allocation of risk premiums.

It can be concluded that if a bid is excessively low and has not allowed for risk premiums then the Contractor is at financial risk. If the risk eventuates the project will suffer.

• Form/Type of Contracts adopted

All interviewees have worked and are familiar with the FIDIC Fourth Edition. Only Two have worked on projects using the FIDIC 1999. The type of contract commonly used as explained by interviewees are the remeasured and the lump sum. The cost plus would reduce disputes but puts the contractor at more risk of cost overrun. The design build contract is not commonly used in Lebanon but could be a good alternative where the design is not completed at tender

The literature describes different types of contracts and different procurement selection criteria devised and utilized in other countries to best fit the project needs.

> The four projects are based on the FIDIC Form of Contract Fourth Edition. They are divide equally between 2 lump sum contracts and 2 remeasured contracts. No inference can be made from the dispute cases to the effect of adopting lump sum vs. remeasured contract on disputes.

The FIDIC Form of Contract Fourth Edition is commonly used. The interviewees are not familiar with the new FIDIC 1999 Form. The type of contract should be carefully selected based on the project specifics. For example, where there is a need to expedite and have fast track works design build can be used instead of going for the traditional lump sum contract with missing design.

• Particular Conditions

"These FIDIC conditions are generally well-balanced and, as with any contract, there are a great number of links and relationships between different clauses, not all of which are express or otherwise obvious. With any amendment, therefore, there is the danger of upsetting the balance or of creating unintended consequential changes to related provisions. It is in the interest of all parties that changes should be kept to minimum." (Corbett, 1991)

The Interviewees pointed out contract clauses that create loopholes in the contract and render the contract unbalanced mainly relating to:

- Price escalation
- Notification of claims
- Design liability.
- Employer's prior approval to any Engineer's time and cost determination
- Order of precedence of Contract documents
- Deletion of works

The clauses mentioned by the interviewees were examined where the risk was shifted to the Contractor through the particular conditions as shown in Exhibit D. Inference of their direct impact on influencing disputes could not be made through the case study examination.

Care should be taken in changing particulars not to cause loopholes or make the Contract unbalanced.

TENDER

• Contract Documents

"Insufficient or incorrect design information" was identified in the literature as the most significant of the eight risk areas examined by Shen (1997). Careful planning and specifications were identified as important factors that could reduce disputes. The hazards of short design periods were also discussed. "Insufficient or incorrect design information" was identified in the literature as the most significant of the eight risk areas (Shen, 1997). Also, Contract documents were identified to be critical. (Spittler et al., 1992).

The importance of having complete and clear contract document and its effect on time and cost overrun was stressed by all interviewees. Clear achievable design/specifications was raised as an important factor that could minimize ambiguities and disputes resulting from contradicting explanations of the contract requirements. The importance of having complete and clear contract document and its effect on time and cost overrun was stressed by all interviewees.

Design error and contract documents unclear occurred in 22 out of the 50 dispute cases. This led to disputes of additional time and cost or both. Also, two cases of unclear design requirement resulted in trial and error approach in specifying the unclear items, which in turn resulted in delay. "Defective contract documents" had the fifth highest recurrence among the risk areas in Table 10 of Chapter 5 (21 out of 50) and the Contract documents not clear had the highest frequency of ocurrence among Table 12 of dispute factors (18 out of 50).

Insufficient and incorrect design would cause disputes due to the fact that ambiguities will be argued as the contractor would want to reduce his losses and go for the cheaper solution. The Engineer would want to maximize on the quality and go for the more expensive reading. The specifications should be written/tailored for the particular aspects of the project and not copied and pasted from other projects trying at all time to have clearly specified requirements otherwise disputes will arise.

• Tender Period

The literature mentions that in some cases short bid times are given to price a mass of documents. The Contractor might make hurried assumptions while pricing, and if they are contrary to the designer's intent, the contractor will raise a dispute as the designer's interpretation might cause incurring more time and cost.

The Bid Time was raised as a problem issue in the interviews however, some of the interviewees argued that the bid time allowed might be enough. Mostly it was agreed that the bid time must be set based on the complexity of the project.

Although the Contractor is deemed to have satisfied himself with the Contract Documents, the dispute areas and the Contractor's claims in many cases indicate that the Contractor might not have been aware/did not price certain items in the tender. However, even if the Contractor has not accounted for them, this does not lead to the conclusion that a longer bid period would have reduced those cases.

The Contractor should be allowed for sufficient bid time so that any ambiguities or discrepancies are cleared especially where the clause 'Sufficiency of Tender' will clearly allocate risk of missing information to the Contractor. However, there is no conclusive evidence of the influence of tender period on disputes in both the interviews and the case studies.

• Tender Evaluation

The literature states that if the lowest bid is awarded and in case there is an error or the Contractor has planned for additional profit through claims then there is a high risk of future disputes

There was a clear flag against the danger of awarding the lowest bidder unless thorough examination is made and the figures are justified.

In the projects where the lowest tenderer was awarded mainly in Projects B and C, higher occurrence of the dispute factors "Contractor avoiding monetary losses" and "Evading responsibility by blaming the other party" were reported.

Where a contractor with an unreasonably low bid is accepted, the Contractor will be bearing high financial risks that could jeopardize successful completion of the project.

BEHAVIOUR

• Key persons assigned and adequacy of staff allocated

Characteristics of the key participants recommended is fairness, efficiency, wisdom and stability. The literature also stresses on the importance of maintaining a team building and partnering spirit to reduce conflicts.

Contractor stated that where the Engineer does not have the necessary site expertise that would allow him to have reasonable constructive instructions instead of insisting on every detail in specification that would lead to disruption of the works (preferred case referred to by interviewees would be an Engineer that has experience in contracting). Also, interviewees commented on the adequacy of staff size that would allow the Engineer to make timely assessments

Dispute factors reflecting adequacy of staff found in the dispute cases were: lack of experience witnessed in 7 cases, human error/ negligence witnessed in 4 cases and trial and error attempts approach that again resulted from the lack of experience was witnessed in 6 cases. It should be noted that both the **Engineer and the Contractors** had problems of lack of experience, human error and negligence especially in Project A where the designer was innovative

- Adequacy of staff should be checked in terms of: - Size of staff allocated
- Experience of the staff (better have good experience with site work and understanding constructability of the specified requirements)
- Knowledge in the technical areas to accommodate innovative or state-of-the-art project requirements.

• Proactive attitude in resolving problematic issues

Cooperation spirit must be maintained where participants focus on finding the best solution. The relationship is affected by: previous experience, perceived fairness, satisfaction with previous dispute resolution process, and behavior during the job.

Proactive behavior can be better achieved by keeping contractual methods simple. The contractor's financial positions will affect his behavior towards disputes. As such Distributive attitude of contractors resulting from low or even no profit margins should be avoided.

The notion of 'chemistry' among participants as raised by the interviewees in referring to the instinctual reaction of the participants. In some cases, one person that might be a troublemaker on one project, might fit perfectly on another project. Behavioral attitudes might have a bigger effect in Lebanon because the people in the eastern society are more emotionally driven. 11 interviewees said the Engineer has the most influence and should be holding the threads to a project. 4 stated the Contractor has control over actual progress of the works. 4 stated that the Employer's interference is important on critical matters.

Behavioral attitude in attending to disputed issues was witnessed through the dispute factors: Improper communication channel, Lack of cooperation, Unwillingness to resolve disputes, Language expressing bad relation, Evading responsibility by blaming the other party, Engineer firm although contract is grey, Contractor not given incentive

Care should be taken to the ongoing spirit to detect any negative attitude among the participants that could risk the progress of works. Behavioral attitude of participants should be examined to make sure that the proactive behavioral attitude is maintained in a spirit of cooperation and problem solving. This includes at the initial stages of the project ensuring proper tender evaluation that might create a distributive attitude if the contractor's position is at risk.

• Employer's representative (scope of work)

The Employer may add restriction to the Engineer's authorities set in the general conditions however these restrictions should be carefully set to maintain the Engineer's role of being impartial. Some owners demand full audit of the claim assessment before authorizing payment.

Most interviewees agreed that the Employer's Representative should only interfere on strategic issues related to time and cost. He may pinpoint mistakes or problematic issues that the PM might be overloaded to notice. However, his principle role is in examining recommendations by the Engineer to take decisions that are to the best interest of the project. 6 cases report the Employer's interference as a dispute factor. It is worth noting that the cases of Employer interference witnessed in these dispute cases are the ones that were obvious from the documents examined and do not necessarily reflect all attempts of Employer influence.

Employer's representative scope of work should be agreed to and set clearly in the conditions of contract. His role should focus on ensuring proper management of the works by the Engineer without influencing the Engineer's impartiality.

• Dispute resolution mechanism

The Engineer's decision is a vital part of the dispute resolution mechanism in the FIDIC Fourth Edition where the Engineer becomes an impartial judge. The FIDIC 1999 specifies the dispute adjudication board instead. The DRB which is a more consensual and amicable version of the dispute boards has proved to be cost effective for medium sized projects and upwards.

Interviewees from both sides said it would be a better practice to have the Engineer's Decision carried out by a different entity from the Engineer's head office. The participants had conflicting views regarding the benefits of having a DAB. However, it is worth noting that none of the interviewees had the experience of working on a project to which a DAB was assigned. Interviewees stated that there is a common trend to resolve disputes in amicable settlement towards the completion of the works. This trend normally encourages the Contractor to submit more claims to have a bigger negotiation basket.

The four projects have the FIDIC Fourth edition dispute resolution mechanism specified which implies that any of the parties in disagreement with the Engineer's assessment would proceed with requesting an Engineer's Decision. Where any of the parties is not satisfied with the Decision and where amicable settlement is not reached within the specified period, a notice to proceed with arbitration would be given. It was noted in the cases examined that the Engineer's Decision was rejected by the Contractor and was not successful in resolving disputes

The Engineer's Decision did not prove to be efficient in resolving disputes in the case studies examined on the four projects. The interviews reflected conflicting views of whether replacing the Engineer's Decision with a DAB would provide better dispute resolution mechanism noting that the interviewees have no experience with DAB. However, the literature confirms based on previous experience the efficiency of using a DAB. • Assessment of extension of time/ monitoring of delay

Most forms of contract do not spell out clearly details of the principles that will be used for the assessment of claims for extension of time and this is left to the Engineer's discretion.

Not raised in the interviews. The interviews referred to the influence of the Employer's representative in approving the determination of the contractor's entitlement to extension of time.

The dispute factor 'Assessment of delay" was identified to have the second highest occurrence which is 26 out of 50 dispute cases.

The case studies confirm the effect of this dispute factors in more than half the dispute cases. The literature confirms the need to describe clear methods of assessment in more detail to avoid disputes. Where this is not specified, assessment of a fair entitlement is left to the Engineer's discretion.

• Valuation of variations

The importance of agreeing to some of the matters whose assessment are normally disputes e.g. percentage for main office overhead and unit cost of resources was raised. A case where 'fair valuation' was defined differently by an arbitrator and a judge was reported as an example The requirement in the Contract for the The dispute factor Employer's 'Validity/assessment of representative approval variations" was identified to assessment of to have the highest variations was reported occurrence in 28 out of to be a dispute factors. the 50 dispute cases. Similar to the case of assessment of EOT, there is a need for detailed mechanism in particular conditions for assessment of variations. This would help maintain objectivity in assessment and minimize Employer's representative influence reported in the interviews.

• Timely Response to Contractual Matters

The contractor should at all times remind the head office and the field personnel of giving timely and proper notice. On the other hand 'the ability of parties to identify on a regular basis the claimant's entitlement with adequate documentation to ensure an interim payment mechanism' is as well necessary.

It is very important for Contractor to give a notice of a time or cost overrun to allow the Engineer take alternative decisions as may be necessary. Also, the Engineer should make timely determinations. However, specifying contractual periods for Engineer reply to request of time and cost assessments might have a negative impact.



Timely response to contractual matters were examined and proved to be dispute influencing through the dispute factors: Late approval of submittal by Engineer Contractor late/missing submittal Slow attendance to responsibilities

To maintain proper progress of the works, the Contractor should give timely notices and prepare the necessary substantiation to allow the Engineer to make the determinations. The Engineer on the other hand should not be late in issuing those determinations as this might affect the Contractor's cash flow and his progress of works.

• Contractor's Expertise in Management



8.3.Summary

This Chapter examines the sixteen dispute influencing areas identified through the interviews and the case studies. Each is examined through the literature, interviews and case studies where conclusion is drawn through triangulation.

CHAPTER 9: RECOMMENDATION

9.1. Provisional proposals as derived from the Conclusions

Based on the conclusion reached in Chapter 8, a proposed recommendation is set below to minimize disputes for each of the dispute influencing areas. These proposals are then combined in a recommendation that was sent to 5 experienced practitioners for their review and comment.

Dispute	Conclusion	Recommendation
Influencing Area		
Risk Allocation (Best Practice/ Common Practice)	The risk should be clearly identified and better allocated and a relationship of trust must be developed between the parties.	Pre-bid meeting and Pre-award conference: Risks that are included in the Contract and allocated to the Contractor should be discussed especially those deviating from the risk allocation set in the General Conditions. The particular conditions should be addressed. The Contractor would verify/confirm his allocation of risk in the Tender submitted.
Employer's Influence in risk allocation	Both the interviews and the literature confirm the influence of the Employer in risk allocation. The literature adds that there should be awareness by the Employer at Contract Award stage that the project is subject to time and cost overrun.	During preparation of contract documents, the Employer/Employer's representative should be educated on the alternative of contract types available and the Engineer's recommendation should be transmitted. This shall include a discussion/agreement regarding the risk allocation strategy.
Contractor's allowance for risk premium/ Contractor's ability to sustain risk	It can be concluded that if a bid is excessively low and has not allowed for risk premiums then the Contractor is at financial risk. If the risk eventuates the project will suffer.	Pre-award workshop: Risks that are included in the Contract are allocated to the Contractor that should be discussed especially those deviating from the risk allocation set in the General Conditions. The particular conditions should be addressed. The Contractor would verify/confirm his allocation of risk in the Tender submitted.

Dispute	Conclusion	Recommendation
Influencing Area		
Form/Type of Contracts commonly used	The FIDIC Form of Contract Fourth Edition is commonly used. The interviewees are not familiar with the new FIDIC 1999 Form. The type of contract should be carefully selected based on the project specifics. For example, where there is a need to expedite and have fast track works design build can be used instead of going for the traditional lump sum contract with missing design.	During preparation of contract documents, the Employer/Employer's representative should be educated on the alternative of contract types available and the Engineer's recommendation should be transmitted. This shall include a discussion/agreement regarding the risk allocation strategy.
Particular Conditions	Care should be taken in changing particulars not to cause loopholes or make the Contract unbalanced.	Particular Conditions: The provisional recommendation addresses the following critical areas that should be carefully addressed in the particular conditions: Employer's prior approval to any Engineer's time and cost determination Engineer's Decision Price escalation Amendments for Lump Sum Contracts
Contract Documents	Insufficient and incorrect design could cause disputes due to the fact that ambiguities will be argued as the contractor would want to reduce his losses and go for the cheaper solution. The Engineer would want to maximize on the quality and go for the more expensive reading. The specifications should be written/tailored for the particular aspects of the project and not copied and pasted from other projects trying at all time to have clearly specified requirements otherwise disputes will arise.	Pre-bid meeting and Pre-award conference: The Engineer is taking a responsible role to avoid unnecessary disputes. It is true that the Contractor upon reading and pricing a tender is deemed to have understood the requirements and raised queries for any ambiguity. And although contractually upon signing the contract the Contractor becomes liable for all contract requirements. By hosting the workshop, the Engineer is minimizing the risk of having a Contractor that has unreasonably priced (whether deliberately or mistakenly). Such a Contractor could be a 'recipe for disputes' even on issues that he is clearly not entitled to. This will be extremely important for lump sum projects that have design development responsibility.

Dispute	Conclusion	Recommendation
Influencing		
Area Tender Perdiod	The Contractor should be allowed for sufficient bid time so that any ambiguities or discrepancies are cleared especially where the clause 'Sufficiency of Tender' will clearly allocate risk of missing information to the Contractor. However, there is no conclusive evidence of the influence of tender period on disputes in both the interviews and the case studies.	Same recommendation as that of Contract Documents above applies.
Tender Evaluation	Where a contractor with an unreasonably low bid is accepted, the Contractor will be bearing high financial risks that could jeopardize successful completion of the project.	Pre-award meeting: The Engineer should carry out an independent pricing of the project. Also, at tender analysis thorough comparison should be made against the Engineer's pricing and across different tenderers. This should confirm the reasonable contract value and allow for identifying deviations in pricing by tenderers. These deviations should be raised and cleared. Where the Engineer ascertains that the Tenderer's pricing is too low, the Engineer should not accept such unjustifiably low pricing.
Key persons assigned and adequacy of staff allocated	Adequacy of staff should be checked in terms of: - Size of staff allocated - Experience of the staff (better have good experience with site work and understanding constructability of the specified requirements) - Knowledge in the technical areas to accommodate innovative or state-of-the-art project requirements.	The workshop mentioned above is to be sponsored by higher management from both sides. Unlike the pre-bid meeting the pre- award conference shall be attended by most of the project participants to get familiarized with the project requirements, introduce participants to each other and serve as a facilitation workshop to promote a team building spirit. The higher management's intention to maintain a cooperative spirit shall be expressed and an MOU shall be signed. This should reflect the higher management's support to generate and sustain a collaborative approach. The follow up meetings of higher management should reinforce the collaborative approach through predefined reward system. However, it should be noted that previous literature has warned
Proactive attitude in resolving problematic issues	Care should be given to the ongoing spirit to detect any negative attitude among the participants that could risk the progress of works.	

Dispute	Conclusion	Recommendation
Influencing Area		
Proactive attitude in resolving problematic issues (Cont'd)	Behavioral attitude of participants should be examined to make sure that the proactive behavioral attitude is maintained in a spirit of cooperation and problem solving. This includes at the initial stages of the project ensuring proper tender evaluation that might create a distributive attitude if the contractor's position is at risk.	that such approach will not survive if surrounding economic conditions become unfavorable. As such maintaining a 'gainshare/painshare' arrangement is imperative.
Employer's representative (scope of work)	Employer's representative scope of work should be agreed to and set clearly in the conditions of contract. His role should focus on ensuring proper management of the works by the Engineer without influencing the Engineer's impartiality.	Pre-bid meeting and Pre-award conference: The workshop should be preceded by a meeting between the project manager and the Employer to inform the Employer of the tender finding and the agenda of the workshop
Dispute Resolution Mechanism	The Engineer's Decision did not prove to be efficient in resolving disputes in the case studies examined on the four projects. The interviews reflected conflicting views of whether replacing the Engineer's Decision with a DAB would provide better dispute resolution mechanism noting that the interviewees have no experience with DAB. However, the literature confirms based on previous experience the efficiency of using a DAB.	meetings clarifying the intent of the meeting and the steps that will follow. Also, during preparation of contract documents, the Employer/Employer's representative should be educated on the alternative of contract types available and the Engineer's recommendation should be transmitted. This shall include a discussion/agreement regarding the risk allocation strategy and the dispute resolution mechanisms to be set in the contract. Also, the Employer Representative's duties and obligations could be agreed to during this meeting.

Dispute	Conclusion	B acommon detier
Influencing Area	Conclusion	Recommendation
Assessment of Extension of Time/Monitoring of Delay	The case studies confirm the effect of this dispute factors in more than half the dispute cases. The literature confirms the need to describe clear methods of assessment in more detail to avoid disputes. Where this is not specified, assessment of a fair entitlement is left to the Engineer's discretion	The clause for assessment of extension of time should be modified in the particular conditions to account for the following details that would reduce disagreement on time assessments
Valuation of Variations	Similar to the case of assessment of EOT, there is a need for detailed mechanism in particular conditions for assessment of variations. This would help maintain objectivity in assessment and minimize Employer's representative influence reported in the interviews.	Clarifications are proposed to better define what is deemed by the contract as being "appropriate" and "suitable" in the FIDIC Red Book Fourth Edition or "appropriate rate" and "reasonable profit" in the FIDIC 1999 suite to reduce disagreement on cost assessments.
Timely response to contractual matters	To maintain proper progress of the works, the Contractor should give timely notices and prepare the necessary substantiation to allow the Engineer to make the determinations. The Engineer on the other hand should not be late in issuing those determinations as this might affect the Contractor's cash flow and his progress of works.	Pre-bid meeting and Pre-award conference: An agreement should be reached on administrative matters such as time of reply of submittals, no. of submittals per month allowed, notice of variations, notice of delay, encouragement of open communication.
Contractor's Expertise in Management	The Contractor should perform due diligence in proper management in execution of the works.	No specific proposal is made beyond the Contractor exercising his due diligence in proper management of his obligations.

9.2. Provisional Proposals

Almost each of these areas has been addressed by the literature producing recommendations to promote better practice. Inspite of those attempts, the disputes witnessed have led us to consider further recommendations. Those recommendations presented in 3 sections were based on conclusions triangulated from the literature, interviews, and dispute cases in Chapter 7. They intend to add or even re-emphasize what has already been recommended, published and known as best practice based on collected empirical evidence. The nature of the recommendation developed under three themes presents not a list of instructions but rather a reminder to the spirit that needs to be instilled pre- and post- contract signature. It is worth noting that the recommendation was based on problem areas witnessed and substantiated through our case studies, other areas of equal significance in dispute minimization could be similarly examined focusing on the contractual mechanism for dispute resolution.

Section 1: Conditions of Contract

Recommendations regarding drafting the particular conditions are of two types: promote introducing more details in some conditions mainly related to time and cost that have proved in our study to be highly prone to disputes and warn against upsetting the balance of the general conditions through commonly witnessed amendments to certain clauses.

I.I. Introducing Mechanism for Time and Cost Assessment

The 50 dispute cases examined showed the highest occurrence of disputes resulting from disagreement on time and cost assessments. Also, the Contractor's entitlement to time and cost resulting from occurrence of Employer's Risk events and was disputed in the four projects examined. As such the following is recommended to minimize such disputes.

I.I.I. Extension of Time

(applicable to Clause 14 in FIDIC Red Book Fourth Edition)

The clause for assessment of extension of time should be modified in the particular conditions to account for the following details that would reduce disagreement on time assessments:

- Form and size of baseline programme to be submitted i.e. number of activities, level of resource detailing should be set in the Contract.
- The period for submitting updates (whether weekly or bi-weekly) and the format for listing the delayed activities, the float available for each activities, their impact on the project completion should also be specified.
- The delayed activities should be identified as Employer's delay or Contractor's delay. The Employer's right to mitigate his own delay should be agreed and clarified. Where the delay is by the Contractor, he/she should submit details of the plan to mitigate the same. Where there is an ongoing delay by the Employer, the level of detail required to substantiate a claim for extension of time shall be set. Where the Contractor fails to do so, the Engineer shall make the assessment to the best of his knowledge based on the data available.
 - Where EOT is granted or actual sequence of work has changed this shall be reflected in a revised programme. Accordingly, a list of all changes made to the baseline programme should be submitted. Failing to do so, the Engineer's planner shall revise the programme to accommodate the changes reflected in the new sequence of ongoing works on site until such time that the Contractor provides the update. The updates prepared by the Engineer shall be binding and shall be used for assessment of EOT for any ongoing excusable (Employer's) delay.
 - Contentious issues such as concurrency and disruption should be addressed. The definition of concurrency should be clearly set out i.e. whether concurrent delay will be defined as two or more delay events occurring at the same time or the Employer Risk event and the Contractor risk event having concurrent effects should be cleared. Also, basis and technique for disruption substantiation and analysis should be specified.

• The SCL Delay and Disruption Protocol should be used as a guide for the points raised above including the delay analysis method to be adopted, the float 'ownership', concurrency, acceleration...etc.

The particular conditions could specify the following for example:

- Maximum number of activities in the baseline programme not to exceed 5,000 activities
- Updates submitted on a bi-weekly basis detailing the progress of work through the ongoing activities, the float of these activities, any delay incurred in these activities.
- For delay occurring from Contractor's Risk events, the Contractor shall submit proposal to accelerate or resequence the work to reduce this delay. This shall be submitted in the bi-weekly programme to be checked and approved by the Engineer.
- For delay occurring form Employer's Risk event, the Engineer shall make necessary efforts in agreement with the Contractor to reduce the delay impact such as expediting the material and approval submittal cycle.
- The Window Analysis techniques should be used for assessing the Contractor's entitlement to delay resulting from Employer's events. This shall be based on bi-weekly windows as per the submitted updates.
- Concurrency is considered to take effect when an Employer's Risk Event and Contractor's Risk Event occur in the same window and have concurrent effects.
- Claims for Disruption submitted shall be substantiated by comparing disrupted operations with performance of similar works in another window.

I.I.II. Valuation of Variations

(applicable to Clause 52 in FIDIC Red Book Fourth Edition)

The clause for valuation of variation should be further detailed in the particular conditions. FIDIC's guide to the Fourth edition suggests a variation procedure to reach an agreement prior to issuing a variation. This however, introduces the risk of
delay in executing the works where a timely submission of rates is not made or where an agreement is not reached. The following clarifications are proposed to better define what is deemed by the contract as being "appropriate" and "suitable" in the FIDIC Red Book Fourth Edition or "appropriate rate" and "reasonable profit" in the FIDIC 1999 suite to reduce disagreement on cost assessments:

- In the case of additions/omissions of quantities, the applicability of contract rates throughout the contract period should be clearly specified. Also, a mechanism should be set for the fixing of rates beyond that period.
- In the case of new items, it should be stated whether prorated contract rates apply or whether market rates should be adopted. This should be clarified in line with the concept of 'equitable price adjustment' discussed below.
- The Contract should state the Contractor's entitlement to overhead (if applicable) and profit (along with subcontractor's overhead and profit) to be set as a percentage to both cases of additions and omissions of works.

It should be noted that in the cases where price adjustment for escalation of material and labor rate is not allowed, the matter of 'equitable price adjustment' should be cleared. This concept preserves the Contractor's entitlement to a price adjustment that would leave the Contractor in the same economic position he/she would have been in had the variation not been issued. It is intended to make the "Contractor whole." This matter has been raised in the Weldon Plant v Commission for New Towns [2000] (TCC BLR 496) case where arbitrator basing his award on the concept of equitable price adjustment allowed for the incurred additional cost but not for overhead and profit. Where reference was made to the court, the judge remitted the award and held that the Contractor is entitled to overheads and profits.

This can be based on the 1999 Red Book clause 12.3 as amended below:

For each item of work, the appropriate rate or price for the item shall be the rate or price specified for such item in the Contract or, if there is no such item, specified for similar work. However, a new rate or price shall be appropriate for an item of work if:

(a) (i) the measured quantity of the item is changed by more than 10% from the quantity of this item in the Bill of Quantities or other Schedule,

(ii) this change in quantity multiplied by such specified rate for This item

> exceeds 0.01 % of the Accepted Contract Amount, (iii) this change in quantity directly changes the Cost per unit quantity of this item by more than 1%, and (iv) this item is not specified in the Contract as a fixed rate item":

> In this case the additional quantities shall be assessed based on market rates. The Contractor will be entitled to the overhead and profit percentage for variations set in the Appendix to Tender.

or

(b) (i) the work is instructed through variation

(ii) no rate or price is specified in the Contract for this item, and (iii) no specified rate or price is appropriate because this item of work is not of similar character, or is not executed under similar conditions, as any item in the Contract

In this case

- if the instructed work necessitated new items replacing other items specified in the contract (the latter will be removed in the assessment as an omission) then the Contractor will be entitled to the difference in market rate between the new item and the replaced item at the time the variation order was issued along with the overhead and profit percentages specified in the Appendix to Tender. - If the instruction work necessitated new items constitue additional works and do not replace any other items in the Contract, then market rates at the time the variation order was issued will be used as base for assessing the cost of executing the works. The Contractor will be entitled to the overhead and profit percentages specified in the Appendix to Tender.

For variation orders issued beyond the contract period, market rates at the time issued will be used as base for assessing the cost of executing the works. The Contractor will be entitled to the overhead and profit percentages specified in the Appendix to Tender.

I.I.III. Details of Compensation Entitlement in the case of Employer's Risk (applicable to Clause 65 in FIDIC Red Book Fourth Edition)

- Given the fact that Lebanon is going through a political turmoil that might result at times in political instability, a clear listing of the Contractor's entitlement to extension of time due to suspension or loss of productivity resulting from an Employer's risk, could minimize disputes regarding the same.

This would include clear listing of the costs the Contractor will be entitled to in the case of Employer's Risk and the details of the substantiation that should be provided: site staff salaries, head office overhead calculation formula, site operational cost based on invoices, subcontractor's compensation, cost of extension of advance payment guarantee, performance bond, and site insurance as substantiated with bank receipts, plant and equipment depreciation (if allowed under this clause), any other cost the Contractor adds to this Clause during Tender and the Engineer approves.

I.II. Warning Against Amendments to Particular Clauses

I.II.I. Employer's prior approval to Engineer's time and cost determination (applicable to Clause 2 in FIDIC Red Book Fourth Edition)

Employer's prior approval that is set in the particular conditions should not introduce obstacles to the Engineer in performing his duties for the cases where he/she should act as an independent certifier. As such the requirement of the prior approval of the Employer should be limited to approval in principal to variations unless deemed necessary at the technical level. However, a budget limit for variations could be set:

For example, US\$300,000 or 0.001% of contract value can be set by the Employer as a VO budget. The Engineer proceeds with the assessments of the VOs. Where this budget is reached, a cost report is submitted by the Engineer to the Employer for his/her review/approval detailing the list of variations certified and their values and an assessment of the VOs issued that are not yet certified. This would allow the Engineer to proceed with the valuation while giving the Employer an update/indication of the value of the ongoing variations and control on budget overruns.

It is worth noting that the general conditions of the FIDIC Red Book Fourth Edition and the 1999 Red Book do safeguard the Employer's right to seek a decision and arbitration where the Employer is in disagreement with the Engineer's determination as an independent certifier.

I.II.II. Engineer's Decision

(applicable to Clause 67 in FIDIC Red Book Fourth Edition)

Although none of the projects examined adopted the DAB, however, based on the following:

 In the cases examined, the Engineer's Decision was a reconfirmation of the Engineer's determination previously made. And these cases were referred to arbitration. As such the Engineer's Decision had no contribution in resolving the disputes. The DAB has been reported to be cost effective for medium sized projects and upwards. That is the reason why the World Bank made the procedure mandatory for all International Bank for Reconstruction and Development –finance project exceeding US\$50M.

As such it is recommended that DAB be adopted through the 1996 Supplement to the FIDIC Fourth Edition or through the 1999 Red Book in projects exceeding US\$50M.

I.II.III. Amendments for Lump Sum Contracts

(applicable to several Clauses in FIDIC Red Book Fourth Edition)

Although FIDIC gives guidance for amendments of the Red book to 'Lump Sum' form, the 'Lump Sum' form is used on projects where the design has been developed by the Employer to a sufficiently advanced stage where from the information supplied the Contractor can prepare all drawings and details necessary for construction without having to refer back to the Engineer for clarification or further information. However, the intention of this lump sum form as presented by FIDIC is for works which are simple and straightforward of relatively low value. For larger works it is recommended that the FIDIC Conditions of Contract for Design-Build (which is also a lump sum form) be used.

As such care must be taken in adopting the 'Lump Sum' form of Contract and it is advisable that this form be avoided for complicated and large projects.

I.II.IV. Price Adjustment

(applicable to Clause 70 in FIDIC Red Book Fourth Edition)

Although the particular conditions allow the drafter to omit the Contractor's entitlement to price adjustment, this has only been advised for short term contracts ie one year maximum. For long term contract periods, this would be in contradiction with the civil code where the Théorie de l'imprévision provides for reducing the Contractor's losses to reasonable limits by way of compensation where

the Contractor experiences excessive losses due to the same. However, if the drafter intends to reduce price escalation allowance then a percentage limit could be set beyond which cost escalation of labor and material clause would be applicable. This would reduce the Contractor's allocation for such risk in his tender price. On the other hand, where the market rates decrease, the Employer would be allowed to cost savings.

The adjustment to the monthly statements in respect of changes in cost could be set as per the following formula:-Pn = A + b(Bn/Bo) + c(Cn/Co) + d(Dn/Do) + e(En/Eo) + f(Fn/Fo) +g(Gn/Go) + h(Hn/Ho) + i(In/Io)where:

Pn is a price adjustment factor to be applied to the amount for the payment of work carried out in the subject month, excluding variations (set as per conditions (a) and (b) of item 1.1.2 above) and daywork; The Price adjustment would be applicable where Pn > 1.1.

The following are specified in the Appendix to Tender: A is a constant representing the nonadjustable portion in contractual payments;

b, c, d etc., are coefficients representing the portion of each cost element in the following order: cement, reinforcing steel, aggregates, copper, black steel, galvanized steel, labor wages, Euro rate (for items imported from Europe) in the works carried out in the subject month; the sum of A, b, c, d, etc., shall be one;

Bo, Co, Do, etc., are the base cost indices or reference prices corresponding to the above cost elements on the day 28 days prior to the latest date of submission of Tender. Sources of base cost indices except for Euro Rates shall be those issued by Governments as set in item 3.6.5 below.

Bn, Cn, Dn, etc. are the current cost indices or reference prices of the cost

elements for month "n", determined on the average of the past 4 months prior to the last day of the period to which monthly statement n relates. Sources of current cost indices except for Euro Rates shall be those issued by Governments as set in item 3.6.5 below.

If the Contractor fails to complete the Works within the Time for Completion, adjustment of prices thereafter until the date of completion of the Works shall be made using either the indices or prices relating to the prescribed time for completion, or the current indices or prices, whichever is more favorable to the Employer.

Section 2: Procurement Practices

The Procurement practices presented below is meant to minimize on disputes that would result from misinterpretation given the prevalent contract documents condition at the time of tender in terms of design completeness, contract clarity, risk allocation. This shall be achieved at three stages:

II.I. Pre-bid Meeting

This meeting is intended to clarify: scope of project and design intent, the conditions of contract, the project programme, method of measurement, technical requirements, design development expected, design verification and calculation required for certain systems, definition of equivalents, aesthetical requirements, acceptable range for certain architectural items like marble should be presented. This meeting shall be minuted to be followed by queries raised in the question and answer period.

- The pre-bid meeting is commonly conducted. Reference is made to it as a reminder to its significance and impact in ensuring that the bidders understand the project well enough to provide responsible and responsive bids. As such the Engineer should utilize this meeting to clarify ambiguities which have proved to be a major source of disputes in our empirical studies.
- For items in the specification where the architect insists on a specific supplier or a product and to avoid monopoly of prices, offers for these prices should be

received negotiated and included in the contract as nominated suppliers for which the Contractor will be entitled to an overhead and profit.

II.II. Pre-award Conference

This conference shall be held after tender analysis to achieve the following:

- The conference will consist of two parts. The first part will consist of scrutinizing and negotiating the bid, checking qualifications, making sure that there are no errors in pricing and that all items required including risks allocated to the Contractor in the Conditions have been reasonably priced. The second part of the meeting should be attended by most project participants attending/involved to reconfirm the Contractor's understanding of all points/issues raised in the pre-bid meeting and the Q&A in more detail. This is the last attempt to reduce grounds of misinterpretation.
- Any detail that is not mentioned or raised during this workshop does not relieve the Contractor from his/her contractual obligations. However, the Engineer will be held responsible for all statements made by him/her during the conference. All discussions shall be minuted.

II.III. Higher Management MOU

The workshop mentioned above is to be sponsored by higher management from both sides. Unlike the pre-bid meeting the pre-award conference shall be attended by most of the project participants to get familiarized with the project requirements, introduce participants to each other and serve as a facilitation workshop to promote a team building spirit.

The higher management's intention to maintain a cooperative spirit shall be expressed and an MOU shall be signed. This should reflect the higher management's support to generate and sustain a collaborative approach.

The MOU shall be signed in the presence of all project participants. Provisions should be made for a periodical higher management follow up meetings to monitor:

- Progress of the work and any ongoing critical delay that would require a higher management involvement.
- Disputes that were not resolved at the project management level.
- Direct control of on-site participant behavior. Major attitudinal clashes that might create tension should be identified where action may be taken as necessary to relocate such personnel to different responsibilities.

The follow up meetings of higher management should reinforce the collaborative approach through predefined reward system. However, it should be noted that previous literature has warned that such approach will not survive if surrounding economic conditions become unfavorable. As such maintaining a 'gainshare/painshare' arrangement is imperative.

Section 3: Regulating the Industry

- It is contingent upon the Engineer's willingness and commitment to take more responsibility towards achieving a successful project as opposed to preparing a set of contract documents that would provide a good 'shield' during contract administration keeping both the Engineer and Employer at the safe side of the contract. This spirit will have to be encouraged and sponsored by the Employer. Where correct practices are not adopted by the Engineer and the Employer who 'set the rules of the game'; they will be taking the risk of increased disputes at their own peril. This does not by any way undermine the significance of having the Contractor with the right ingredients "qualifications, finance, responsiveness".
- The above requires a certain degree of awareness and self-monitoring by project participants. However, where wrong practices are applied whether intentionally or due to lack of knowledge, the consequence will be an increased amount of disputes and an unhealthy construction environment. The Construction industry has a significant contribution to GDP (8.2% reported in 2005). This justifies the need for a regulatory body (set by the government) to set rules and regulations for this industry similar to those set in other sectors (banking, aviation,...).
- In their direct role as clients in public projects, governments in other countries maintain relationships with contractors which transcend that between parties to a

normal construction contract. Such efforts are made to nurture local construction companies in some Far Eastern countries. Japan protects its local industries often under the 'infant industry' principle. Singapore makes special efforts to make their contract terms favorable to and supportive of contractors. Government also influences the industry indirectly through the availability of labour and the levels of wages which are dependent upon regulations on the recruitment of foreign workers, statutory charges payable, insurances required, and so on (Ofori, 1990).

- The ministries responsible for construction in the developed countries maintain subsidiary bodies for policy formulation, implementation and monitoring. In Singapore for example, the regulatory body has administrative responsibility for the construction industry training center. It administers all the schemes formulated to facilitate the technological upgrading of the construction industry. These take the form of tax incentives, financial assistance for the acquisition of plant and equipment. The Board has initiated or acted as the point of focus for research into problems facing the local industry. It maintains a central register of contractors, and has published a number of guides on good construction practice (Ofori, 1990).

3.6. Similar regulatory body in Lebanon could prevent misuse/abuse of contracts by both parties and thus promote a healthy environment. This could be achieved by enforcing regulations for the following five areas:

- Drafting Conditions. Unbalanced unfair contracts should be prohibited. Care should be taken to clauses that are likely to cause disputes. Particular conditions should be standardized where the Engineer is allowed to modify the conditions only to account for project specifics and not to manipulate/exploit them such as removing the Contractor's entitlement to 'Employer's Risk Event' consequences or extension of time due to 'Other Special Circumstances'.
- **Procurement Practices:** Procurement practices including the pre-bid and pre-meeting mentioned above along with the proper tender analysis methods should be reinforced.

- Professional skill requirements:. In public projects merit schemes are devised on certain countries where good performance records of contractors are awarded merit stars that give them a bidding preference. On individual level, qualifications could be set recognising necessity of main project participants from both sides to be members in institutes like the PMI, CIOB, RICS or CIArb. A Lebanese PMP (Project Management Practitioner) degree could be designed and awarded, and set as a requirement for practitioners in the industry. As such practitioners would undergo an exam that would test their knowledge of contracts, drafting conditions (item 3.6.1 above), and procurement practices (item 3.6.2 above). Seminars could be hosted to provide necessary training and yearly seminars to be attended by those holding the degree. This can be done in collaboration with the institutes mentioned above.
- Violations of Safety Code: The site might be periodically inspected by officials of public agencies for building control purposes and for ensuring that working conditions are safe and healthy. Flagrant violation of safety codes should be dealt with as felony offenses punishable by more than just monetary penalties.
- **Price Indices:** The government should issue on a monthly basis price indices reflecting increase/decrease of rates for material used in the construction industry such as: cement, reinforcing steel, aggregates, copper, black steel, galvanized steel along with labor wages. These shall be used as basis in calculating entitlement to price adjustment.

9.3.Recommendations

The replies received from the experts is summarized below. The full commentary received is presented in Appendix

Section 1: Conditions of Contract

Introducing Mechanism for Time and Cost Assessment

All experts except Expert C confirmed on the importance of both parties to adopting the proposed philosophy and act professionally throughout. The Employer must recognize the Contractor's aim to make money and the Contractor must recognize the Employer's aim to achieve value for money. This also places emphasis on the Employer who should recommend the correct procurement, and empower the Engineer to act impartially.

Extension of Time

Three of the five experts agreed that a mre detailed clause would help minimize disputes. Expert A considered it a necessary condition for the success of the introduced mechanism is set as 'the Engineer genuinely acting not only impartially but independently without the influence of the Employer.' The clause will help obviate the tactic of continual referral for 'further and better particulars' employed by the Engineer. Expert B states that it will make the Contractor prepare the resource required for proper implementation of this clause. Expert C and D disagreed stating that detailed requirements would increase the risk of certain disputes. But at the end of the section Expert C states that a Lebanese version of the 'SCL' Protocol could work for projects over a certain value.

Valuation of Variations

All 5 experts agreed that introducing more clarity would help reduce disputes. The importance of introducing greater certainty and clarity is stressed by expert A as long as both parties fully understand the risk. Substantiation of the words "appropriate" or "reasonable" for something with pre-defined parameter from FIDIC 1999 would assist reduce disputes as stated by Expert B. Expert C confirmed that detailed wording of this clause would provide transparency in establishing and agreeing rates but expressed concern to the sample amendments presented as it might lead to detailed measurement and calculation of numerous items.

Details of Compensation Entitlement in the case of Employer's Risk

All experts agreed that detailing compensation entitlement in thecase of Employer's Risk would help reduce disputes. Expert B adds that the conditions should firstly better define and classify political uncertainty or instability as Force Majeur. Expert B also, proposes expansion of sphere of any works to areas such as Afghanistan, Iraq, Kurdistan etc to merit some specialist advice from experience in these areas.

Warning Against Amendments to Particular Clauses

All experts agreed that care must be taken in amending particular conditions. Expert B commented that the amended forms of contract used in the Middle East are so heavily in favour of the Employer that the Contractor is carrying an unreasonable and disproportionate amount of risk (in his views) and the contractor that wins the tender will invariably be the one that either is incompetent in assessing the risks or, is hoping all goes well regardless and that he can just muddle through.

Employer's prior approval to Engineer's time and cost determination

Four experts agreed that the Engineer should be allowed to act impartially. Expert A states that the proposed mechanism does not provide the Contractor with any greater cost certainty until such approval is received. However, he recognizes the importance of "agreement in principle". Expert C disapproved of this "agreement in principle" since delay will cause disputes.

Expert B neither agreed nor disagreed. He proposed an alternative employed in the UK under certain tripartite financing arrangements, which can be incorporated in certain forms of Contract, is provision for a "Bank Monitoring Surveyor" typically in Development Projects, usually an RICS Chartered Quantity Surveyor directly employed by, and acting specifically to protect the interests of the Funding Bank.

Engineer's Decision

The five Experts encourage the use of DAB as a better alternative in a balanced contract. Both Experts A and B refer to the success of the Adjudication process that became mandatory in the UK and helps avoid protracted litigation.

Amendments for Lump Sum Contracts

The five Experts agree to the importance of choosing the correct procurement route and contractual mechanism as a major factor in the success of a project. Expert D defended the use of lump sum contracts as it has proved to be successful on many projects provided competent cosultants are approved.

Price Adjustment

The five experts agree to introducing a price escalation formula. However expert B proposes adopting the British Cost Information Service and Royal Institution of Chartered Surveyors approach of fair and transparent adjustment of fluctuations by using simple weightings published on a monthly basis in times of base price turbulence. Expert D proposes implementation of Clause 70.1 fully details as an alternative.

Section 2: Procurement Practices

Four experts agreed to the procurement practices proposed. Expert A had attended such workshops and considers these processes can greatly reduce the incidences of claims through misinterpretation of the tender and Employers requirements. Expert A stated that the combined result of both meetings should be that the Employer is made fully aware of what he is buying for the proposed Contract price and the Contractor is fully aware of what the Employer is expecting to get for this price.

Expert C disagreed to the significance of such procurement practices as it would be too late to make meaningful changes. However, Expert C reported no previous experience in attending such workshops.

Expert D had experience on projects carried out on 2 stage competitive basis where meetings were conducted at "time of issue" and "mid tender".

Section 3: Regulating the Industry

All three Experts agreed to the significance of regulating the industry and its impact on reducing disputes. Expert A reported that regulating the industry in the UK had a mostly positive effect particularly in terms of dispute resolution.

9.4. Summary

A provisional recommendation was formulated in this Chapter based on the conclusion reached of the dispute influencing factors. The provisional recommendation consisted of 3 sections. The first section addresses conditions of contract where introducing a mechanism for time and cost assessment is proposed. For other dispute influencing clauses a warning is set against amendments in particular conditions that would render the conditions unbalanced. The second section addresses procurement practices where prebid meeting/pre-award conference is encouraged. Also, this section proposes having an MOU of higher management. The last section recommends regulating the industry mainly the procurement practices, professional skill requirements, violations of safety code and price indices.

These recommendations were sent to five experts for their review. There was common consensus by at least 3 of the 5 experts on each of the points raised in the recommendation.

CHAPTER 10: SUMMARY, LIMITATIONS AND FUTURE RESEARCH

10.1.Summary

The research conducted to minimize disputes started with a thorough literature review. The examination of disputes has led to the study of risks, conflicts, claims, procurement methods, and behavioral attitude of participants which were found to be interrelated. The literature revealed abundant research studying different aspects of the problem and proposing preventive and remedial measures at different stages of the construction project. Different studies of the behavioral aspect of parties to a construction contract and their impact on disputes resolution was also examined.

The literature review allowed for having a broad overview of the interrelated factors that influence disputes. At this point different forms of research were examined to devise a methodology that would best serve the purpose of our study and provide valid and reliable findings. The realism paradigm was adopted and accordingly guidelines for conducting preliminary examination of projects, interviews and case studies along with selection criteria and analysis procedure. A pilot study was conducted on 20 projects in Lebanon where figures project cost and duration reflected the time and cost overruns on these projects and the occurrence of disputes. As such a need for investigating the Lebanese construction industry further was established to identify the disputed influencing areas.

Based on both the literature review and the pilot study, questions were set for a semi-structured interview that was conducted with twenty four practitioners in the Lebanese construction industry including both project managers and contract administrators divided equally between Engineers (the Consultant as defined under a FIDIC form of Contract) and Contractors. The responses received led to identifying fifteen dispute influencing areas which were categorised under four themes: risk, tender, behaviour, and contract administration. Also, the interviews revealed that one third of the practitioners at the project management level had limited knowledge in contractual matters.



Figure 27 Summary of Research Methodology

Fifty dispute cases on four ongoing projects were closely examined where for each the chronology of events was drawn and the cases were analyzed where risks that eventuated and dispute factors were identified based on categorisation of previous literature. Again, the dispute factors were categorised under ten dispute influencing areas of those identified from the interviews.

The dispute influencing areas from both interviews and cases studies were combined in a set of sixteen dispute influencing areas categorised under the four themes mentioned earlier. For each of these dispute influencing areas, data was collected from the literature, the interviews and the case studies and compared. Conclusions were derived through comparing this data. A set of provisional proposals were suggested against these conclusions. These were reviewed by 5 expert opinions. An overview of the work carried out in this thesis is illustrated in Figure 27.

In brief, the research conducted identified sixteen dispute influencing area as witnessed from the interviews and the case studies. Although many of the dispute influencing factors had been addressed in the literature carried out in other countries, these dispute influencing factors have proved to be existent in the Lebanese construction industry. Moreover, further dispute influencing areas related to the conditions of contract and the behavioral aspect of participants were more evident in the research examined in this thesis than had been reported in the previous literature. A set of recommendation is proposed which includes the 'good practice' principles but also stresses on approach that could help resolve specific dispute influencing factors that had not been directly addressed in previous studies.

10.2. Research Criteria

The four tests to judge the quality of research design were satisfied as follows:

- 1. Construct validity: comparison of the results from different sources was adopted where a chain of evidence is maintained between them to maintain construct validity.
- 2. Internal validity: was establishing through explanation building where the iterative approach of examining the cases helped confirm the causal like through several iterations.
- 3. External validity: replication logic among the four projects and among the 24 interviewees that work for prominent consultant and contracting companies allows for generalization inside the Lebanese borders.

4. Reliability: Procedure of different stages along with the case study database was properly documented to allow other researchers reach the same results by following the same steps.

10.3. Meeting the Aim and Objectives

The initial research aim was met where the causes of common disputes in Lebanese construction industry were identified through 16 dispute influencing areas listed in Figure 26. Also, the relationship between disputes, risk allocation and behavioral attitudes in contributing to dispute emergence was defined as shown in Figure 24. Recommendation was proposed to minimize the occurrence of disputes.

Each of the following objectives set in Chapter 1 were met. They are re-listed below:

- Identify and map the interrelated factors causing disputes based on literature and previous research.
- Examine common practices in contract administration and claims management in Lebanon mainly focusing on the procurement trends, forms of Contract used, and risk allocation strategies.
- Gather and analyse data on the nature, incidence and frequency of disputes in the Lebanese construction industry
- Use the processed data to address the importance of sound contract conditions administered by experienced and knowledgeable practitioners and the likely impact on the minimization of disputes.
- Make educative recommendations for academics and practitioners

10.4.Research Limitations

The following limitations were faced / witnessed while carrying out this research work:

• The research for dispute factors led to a study of different interrelated areas as shown in Figure 7. This made the research broad in nature examining those interrelated factors extensively. As such there was a limitation to the level of detail and depth that could be achieved in studying each of those areas. Examination of the interrelated areas could be the focus of future studies.

- There was a limitation to the access of empirical evidence including the pilot studies, the interviews and the dispute cases. There is also a limitation to the time for collecting these information within the research time frame.
- The subject by itself examines details of disputes some of which have reached arbitration and litigation and are ongoing. As such given the sensitivity of the dispute details draws lines of confidentiality. Also, there might be a limit to the openness in the interviews in discussing the amounts of disputes encountered especially on ongoing projects as this may reflect weak management.
- Another limitation is having the research carried out in the Lebanese context only due the time and budget constraints. There was an initial interest to conduct this research in the Middle East context but that could not be achieved given those limitations. However, similar future research can be conducted in other countries and data can be further compiled and compared among countries accordingly.

10.5.Future Research

Future research can be carried out at two level:

- Similar research using the same methodology can be used to examine the dispute occurrence, causes and factors in other countries. Similar work has already been carried out in some countries as witnessed thorough literature review but that leaves yet other countries where similar research could be conducted. But the significance of the subject and its impact on the construction industry should promote more research. Findings in other countries could identify approach in contract procurement and administration which prove to be successful.
- More focussed studies could be carried out in Lebanon to tackle some of the influencing areas identified in this work:
 - Behavioural examination could be performed through a study of one ongoing project where daily in-depth examination of behavioural attitudes of participants towards disputes would be studied. Also, the change in their reactions could be tracked to better understand this aspect of participant behaviour. In previous studies this was done through attending the project weekly meetings and requesting the

project participants to reflect their views on the ongoing disputed matters through daily agendas.

- Attempts made in adopting the recommended procurement practices could be tested on new projects to further refine the process including the terms of the MOU, and its application.
- Also, further studies could be made to examine the dispute resolution techniques and their success. This would allow focusing on dispute resolution technique with an emphasis on how best the accumulated disputes can be resolved towards the end of the project to avoid arbitration and litigation. As such the success of different amicable dispute resolution methods could be tracked.
- Follow up work could be done with the governmental bodies provided they become aware of such a need and accept to devise a system for regulating the industry which would include specifying a set of regulations. This would include researching regulating attempts made in other countries to benefit from their experience. This exercise could also account for setting standards of knowledge for practitioners.

References

- 1. "Centre for Strategic Studies in Construction" (1990) Building towards 2001 National Contractors Group.
- "National Power (1995) Constructing the Team" Results and Analysis of the Action Review 95 Survey National Power.
- Abdul-Malak,, M.A., El-Saadi M., Abu Zeid, M, G. (2002) "Process Model for Administering Construction Claims." *Journal of Management in Engineering*, 18(2): p. 84-94.
- Ahmed, S.M., Ahmad R., and De Saram, D.D. (1999) "Risk management trends in the Hong Kong construction industry: a comparison of contractors and owners perceptions." *Engineering Construction and Architectural Management*, 6(3): p. 225-234.
- 5. Akintoye, A.S. and MacLeod, M.J. (1997) "Risk analysis and management in construction". *International Journal of Project Management*, 15(1): p. 31-38.
- Alhazmi, T. and McCaffer R. (2000) "Project Procurement System Selection Model". Journal of Construction Engineering and Management, 126(3): p. 176-184.
- Alkass, S., Mazerolle, M. and Harris F. (1996) "Construction delay analysis techniques". Construction Management and Economics, 14: p. 375-394.
- 8. Al-Momani, A.H. (1999) "Construction delay: a quantitative analysis" Inter national Journal of Project Management, 18: p.51-59.
- 9. Arditi, D. and Patel, B.K. (1989) "Expert system for claim management in construction projects." *International Journal of Project Management*, 7(3): p. 141-146.

- Arditi, D. and Yasamis F. (1998) "Incentive/Disincentive Contracts: Perceptions of Owners and Contractors". *Journal of Construction Engineering and Management*, 124(5): p. 361-373.
- 11. Assaf, S., Al Khalil, M. and Al Hamzi, M. (1995) "Causes of delay in large building construction projects." *Journal of Management in Engineering*, 11(2): p. 45-50.
- Atkin B. and Flanagan R. (1995) "Improving Value for Money in Construction", *RICS* p. 18.
- Banwell Committee (1964) "The Placing and Management of Contracts for Building and Civil Engineering Works" *HMSO*.
- Banwell H. (1964) "Report of the Committee on the Placing and Management of Contracts for Building and Civil Engineering Works", *London HMSO*.
- Berends, T. C. (2000) "Cost plus incentive fee contracting experiences and structuring." International Journal of Project Management, 18: p. 165-171.
- Betant H., Monrad F. and Stuhr E. (1995) "How to Handle Disputes in Major Construction Contracts – A Danish Approach". *The International Construction Law Review*, p. 614-625.
- Bing, L., A. Akintoye, Edwards P. J. and Hardcastle C. (2005) "The Allocation of Risk in PPP/PFI Construction Projects in the UK." *International Journal of Project_Management*, 23(1): p. 25-35.
- 18. Blake, R.R. and Mouton, J.S. (1964). The managerial grid. Houston, Texas: Gulf Publishing.
- Bonoma, T. (1985) "Case research in marketing: opportunities, problems and a process." Journal of Marketing Research, 12: p. 199.
- Brandt, S. and Murphy J.(2000) "Productive Conflict Management in the Construction Industry" In ASC Proceedings of the 36th Annual Conference: p. 115-130.

- 21. Brewer G. (1993) "What is a claim". Contract Journal. 21 October, 14.
- 22. Bristow, D. and Vasilpoulos, R. (1995) *The new CCDC 2:* "Facilitating dispute resolution of construction project", *Construction Law Journal*, 11(4): p. 95-117.
- 23. Bufaied, A.S. (1987) "Risks in the Construction Industry: their Causes and their Effects at the Project Level. Ph.D." *Thesis, University of Manchester UMIST*
- 24. Bunni G. N. (2005), The Fidic Forms of Contract. Oxford UK, Blackwell Publishing Ltd
- 25. Bunni, N.G. (2003), Risk and Insurance in Construction, Second Edition, Spon Press, USA.
- 26. Burell, G. and Morgan, G. (1979), Sociological Paradigms and organizational analysis. London, Heinemann Educational Books.
- 27. Carson, D., Gilmore, A., Perry, C. and Gronhaug, K. (2001), *Qualitative Marketing Research, Sage, London*.
- Chan E.H.W., Suen H.C.H. and Chan C.K.L. (2006) "MAUT-Based Dispute Resolution Selection Model Prototype for International Construction Projects." *Journal of Construction Engineering and Management*, 132(5), p. 444-451.
- Chan, D.W.M. and Kumaraswamy, M.M. (1997) "A comparative study of causes of time overruns in Hong Kong construction projects." *International Journal of Project Management*, 15(1): p. 55-63.
- 30. Chan, E. H. W. and Yu, A. T. W. (2005) "Contract strategy for design management in the design and build system." *International Journal of Project Managemen*, t 23: p. 630-639.
- Chang, C. Y. and Ive G. (2002) "Rethinking the Multi-attribute Utility Approach Based Procurement Route Selection Technique". *Construction Management_Economy*, 20: p. 275-284.

- 32. Chang, C.Y. and Ive, G. (2003) "Discussion of 'Model for understanding, preventing, and resolving project disputes' by Panagiotis Mitropoulos and Gregory Howell." *Journal of Construction Engineering and Management*, 129: p. 223-231.
- 33. Cheung S. O., Yiu K. T. and Chim P. S. (2006) "How relational are construction contracts"? *Journal of Professional Issues in Engineering Education and Practice*, 132 (1): p. 48-56.
- 34. Cheung, S. O., Lam K. C., Lam T. I. and Wan Y. W. (2001) "Improving Objectivity in Procurement Selection." *Journal of Management in Engineering*, 17(3): p. 132-139.
- 35. Cheung, S., Tam, C., Ndekurgri, I. and Harris, F. (2000). "Factors Affecting Clients' Project Dispute Resolution Satisfaction in Hong Kong" *Construction Management and Economics*, vol. 18: p. 281-294.
- 36. Corbett, E.C. (1998) "FIDIC 4th A Practical Legal Guide". Sweet and Maxwell, Ltd. London, UK.
- 37. Crowley, L.G. and Hancher, D.E. (1995) "Risk Assessment of Competitive Procurement." *Journal of Construction Engineering and Management*, 121(2): p. 230-237.
- Denzin, N.K. & Lincoln, Y.S. (Eds.) (1994) "Handbook of Qualitative Research". Thousand Oaks, CA: Sage.
- 39. Diekmann J. and Girard M. (1995) "Are contract disputes predictable?" Journal of Construction Engineering and Management, 121(4): p. 355-362.
- 40. Diekmann J., Girard M. and Abdul-Hadi, N. (1994) "Disputes potential index: a study into the predictability of contract disputes", *Construction Industry Institute*. Source Document 101.
- Diekmann, J.E. and Nelson, M.C. (1985) "Construction Claims: Frequency and Severity." Journal of Construction Engineering and Management, 111(1): p. 74-81.

- Dozzi, P., Hartman, F., Tidsbury, N., and Ashrafi, R.(1996) "More-Stable Owner-Contractor Relationships." *Journal of Construction Management and Engineering*, 122(1): p. 30-35.
- 43. Drexler Jr, J. A. and Larson, E. W. (2000) "Partnering: Why Project Owner-Contractor Relationships Change". *Journal of Construction Engineering and Management*, 126(4): p. 293-297.
- 44. Easterby-Smith, M. Thorpe, R., and Lowe, A. (1991) Management Research an Introduction, Sage, London.
- 45. Egan J. (1998), Department of the Environment, Transport and the Regions Rethinking Construction HMSO.
- Eggleston, B. (1993), "The ICE Condition of Contract: Six Edition" A User's Guide. Blackwell Science, Oxford
- 47. Eisenhardt, K., (1989) "Building Theories from Case Study Research." Academy of Management Review, 14: 532.
- 48. Epling, J. A. (1987). "Resolving international construction disputes." *International Journal of Project Management*, 5(4): p. 217-219.
- 49. Essex, R. J. (1996) "Means of avoiding and resolving disputes during construction." *Tunneling and Underground Space Technology*, 11(1): p. 27-31.
- 50. Fenn P., Lowe D. and Speck C. (1997) "Conflicts and disputes in construction." *Construction Management and Economics*, 15: p. 513-518.
- 51. Fenn, P. (1999) "An aetiology of construction disputes". PhD thesis (unpublished), University of Manchester Institute of Science and Technology, Manchester
- 52. Fenn, P. (eds. Lowe, D. and Leiringer, R.) (2006). Commercial Management of Projects. Oxford, Blackwell Publishing.

- 53. Fenn, P. (1997) "Rigour in research and peer review". Construction Management and Economics, 15: p. 383-385.
- 54. Follett, M.P. (1925), Williams and Wilkins, (1926) "Constructive conflict. In H.C. Metcalf (Ed.), Scientific Foundations of Business Administration" *Baltimore*, *MD*.:
- Gaitskell R. (2005) "Current Trends in Dispute Resolution". Arbitration, 71(4), p. 288-299.
- 56. Gardiner, P.D. and Simmons, J.E.L. (1998) "Conflict in small and medium sized projects: case of partnering to the rescue". *Journal of Management in Engineering*, 14(1): p. 35–40.
- 57. Gerson, K. and Horowitz, R. (2003), "Observation and Interviewing: Options and choices in qualitative research" in May, T., *Qualitative Research in Action*, Sage Publications, London.
- 58. Gould, N., Capper, P., Dixon, G. and Cohen, M. (1999), Dispute Resolution in the Construction Industry Thomas Telford, London.
- 59. Gransberg, D. D. and Ellicott M. A. (1997) "Best-Value Contracting Criteria". *Cost Engineering*, 39(6): p. 31-34.
- Guba, E. and Lincoln, Y. (1994), "Competing paradigms in qualitative research", in Denzin, N.K., Lincoln, Y.(Eds.), *Handbook of Qualitative Research*, Sage Publications, Newbury Park, CA.
- Gummesson, E. (2005) "Qualitative Research in Marketing". European Journal of Marketing, 39(3/4): p. 309.
- 62. Gummesson, E. (2nd Ed.) (2000) Qualitative Methods in Management Research, Sage Publications, Beverly Hills: CA.
- 63. Hanson, D. and Grimmer, M. (2007) "The mix of qualitative and quantitative research in major marketing journals 1993-2002". *European Journal of Marketing*, 41(1/2): p. 58.

- 64. Harmon K.M. (2003) "Conflicts between Owner and Contractor: Proposed Intervention Process." *Journal of Management in Engineering*, 19(3): p. 121-125.
- 65. Harris, R.A. and Scott S. (2001) "UK practice in dealing with claims for delay". Engineering, Construction and architectural management. 8 (5/6): p. 317-324
- 66. Hartman, F.T. (2003) "Reducing or eliminating construction claims: a new contracting process (NCP)". *Project Management Journal*, 25(3): p. 25-31
- 67. Healy, M. and Perry, C. (2000), "Comprehensive criteria to judge validity of qualitative research within the realism paradigm". *Qualitative Market Research*, 3(3): p. 118.
- Henderson, R. (1991) "Analysis of Construction disputes and strategies for cost minimization". *Cost Engineering*, 33(11): p. 17-21.
- Jannadia, M.O., Assaf S., Bubshait, A.A., and Naji A. (2000) "Contractual methods for dispute avoidance and resolution (DAR)". *International Journal of Project Management*. 18(1): p. 41-49.
- Jean Lee, S.K. (1992) "Quantitative versus qualitative research methods: two approaches to organization studies". *Asia Pacific Journal of Management*, 9(1): p. 87.
- Jeargeas, G. and Hartman F. (1996) "A Contract Clause for allocating risks". *Transaction of AACE Journal*, D&RM: p. 1.1-1.3.
- 72. Jergeas, G and Hartman, F. (1994) "Contractors' construction-claims avoidance". Journal of Construction Engineering and Management, 120(3): p. 553-560
- 73. Kangari, R. (1995) "Risk Management Perceptions and Trends of U.S. Construction". Journal of Construction Engineering and Management, 121(4): p. 422-429.
- Kartam, N.A. and Kartam S.A. (2001) "Risk and its management in the Kuwaiti construction industry: a contractor's perspective". *International Journal of Project Management*, 19(6): p. 325-335.

- 75. Kehinde, J and Aiyetan, O. (2002) "A study of the nature of contractual claims in building contracts in Nigeria". *Journal of Financial Management of Property and Construction*, 7(3): p. 197-202.
- Kidd, S.A. (2002) "The role of qualitative research in psychological journals". *Psychological Methods*, 7(1): p. 126.
- 77. Knowles, R. (2005) "2nd Ed. 150 Contractual Problems and their Solutions". *Blackwell Publishing Ltd. Oxford, UK.*
- Kululanga, G., Kuotcha, W., McCaffer, R. and Edum-Fotwe, F. (2001) "Construction contractor's claims process framework". *Journal of Construction Engineering and Management*, 127(4): p. 309-314.
- 79. Kumaraswamy M.M. (1997) "Conflicts, claims and disputes in construction Engineering." Construction and Architectural Management, 4(2): p. 66-74.
- Kumaraswamy M.M. (1998) "Consequences in the Construction Conflict: A Hong Kong Perspective." Journal of Management in Engineering, 14(3): p. 71-83.
- Kumaraswamy, M.M. (1997b) "Common categories and causes of construction claims". Construction Law Journal, 13(1), p. 21-34
- Kumaraswamy, M. M. (1996). "Contractor evaluation and selection: a Hong Kong Perspective". *Building and Environment*, 31(3): p. 273-282.
- 83. Kumaraswamy, M. M. and Dissanayaka S. M. (2001) "Developing a decision support system for building project procurement". *Building and Environment*, 36(3): p. 337-349.
- 84. Latham M. Sir (1994) "Constructing the Team", HMSO p. vii.
- 85. Lee, J C. (1994) "Construction disputes and arbitration practice". *Commercial Arbitration*, 39: p. 24-29 (*in Chinese*)

- 86. Lewis, J, Cheetham, D W & Carter, DJ (1992) "Avoiding conflict by risk management the role of the client's project management". *In: Construction Conflict : Management and Resolution (eds P. Fenn & R Gameson), Chapman Hall, London; pp 72-95*
- 87. Lincoln, Y. and Guba, E. (1985), *Naturalistic Inquiry*, Sage Publications, New burry Park, CA.
- 88. Lo, C H (2002) "Comparing Western and Eastern conflict management and dispute resolution MSc thesis", *University of Manchester Institute of Science and Technology, Manchester*
- Long, R.G., White, M.C., Friedman, W.H., Brazeal, D.V. (2000) "The 'qualitative' versus 'quantitative' research debate: a question of metaphorical assumptions". *International Journal of Value-Based Management*, 13(2): p. 189.
- Love, P. E. D., A. Gunasekaran, and Li H. (1998) "Concurrent engineering: a strategy for procuring construction projects". *International Journal of Project_Management*, 16(6): p. 375-383.
- 91. Lowe D. and Leiringer R. Commercial Management of Projects, 2006, Blackwell Publishing, Oxford.
- 92. Luu, D. T., Thomas Ng S., and Chen S. E. (2003) "A case-based procurement advisory system for construction." *Advances in Engineering Software*, 34(7): p. 429-438.
- 93. Mahmoud-Jouini, S. B., C. Midler, and Garel G. (2004). "Time-to-market vs. time-todelivery: Managing speed in Engineering, Procurement and Construction projects". *International Journal of Project Management*, 22(5): p. 359-367.
- 94. Mak, S. and D. Picken, (2000), "Using Risk Analysis to Determine Construction Project Contingencies". *Journal of Construction Engineering and Management*, 126(2): p. 130-142.
- 95. Manning, K. (1997), "Authenticity in constructivist Inquiry: methodological considerations without prescription". *Qualitative Inquiry*, 3(1): p. 93.

- 96. Mathews, J.J. & Ashley, D.B (1985) "Contract Clause Study Data". University of Texas at Austin, Department of Civil Engineering, Construction Engineering and Project Management, Report No UICEPM-85-1. Austin, TX
- 97. Mitropolous, P. and Howell, G. (2001) "Model for understanding, preventing and resolving project disputes." *Journal of Construction Engineering and Management*, 127(3): p. 223-231.
- Molenaar, K, Washington, S and Diekmann, J (2000) "Structural equation model of construction contract dispute potential". *Journal of Construction Engineering and Management*, 126(4): p. 268-277.
- Molenaar, K. R. and Songer A. D. (1998). "Model for Public Sector Design-Build Project Selection". Journal of Construction Engineering and Management, 124(6): p. 467-479.
- 100. Molenaar, K. Washington, S. and Diekmann, J. (2000) "Structural equation model of construction contract dispute potential." Journal of Construction Engineering and Management, 126(4): p. 268-277.
- 101. Morgan, G. and Smircich L. (1980), "The Case of Qualitative Research". Academy of Management Review, 5(4): p. 491.
- 102. Murdoch, J & Hughes, W (1996) "Construction Contracts" Law and Management (2nd edn). E & F N Spoon, London.
- Ng, S. T., Luu, D. T., Chen, S. E. and Lam, K. C. (2002) "Fuzzy Membership function of procurement selection criteria." *Construction Management and Economics*, 20: p. 285-296.
- 104. Ofori, G. (1990) "The Construction Industry": Aspects of Its Economics and Management, NUS Press, Singapore
- 105. Palaneeswaran, E. and Kumaraswamy M. M. (2001) "Recent advances and proposed improvements in contractor prequalification methodologies". *Building and Environment*, 36: p. 73-87.

- 106. Palaneeswaran, E., Kumaraswamy M., Rahman, M. and Thomas Ng, S.(2003) "Curing congenital construction industry disorders through relationally integrated supply chains". *Building and Environment*, 38: p. 571-582.
- 107. Parkhe, A. (1993) "Messy' research, methodological predispositions and theory development in international joint ventures". *Academy of Management Review*, 18(2): p. 227.
- Patton, M.Q. (eds.) (2002), Qualitative Research and Evaluation Methods, Sage Publications, Beverly Hills, CA.
- 109. Pawson, R. and Tiley, N. (1997), Realistic Evaluation, Sage Publications, London.
- Perry, C. (1998) "Process of case study methodology for postgraduate research in marketing". *European Journal of Marketing*, 32(9/10): p. 785.
- 111. Perry, C. Alizadeh, Y. and Riege, A. (1997), "Qualitative methods in entrepreneurship research" *Proceeding, Small Enterprise Association of Australia and New Zealand, Southern Cross University, Coffs Harbour.*
- 112. Pickavance, K. (2000), "Delay and Disruption in Construction Contracts", Second Edition, London: LLP
- 113. Pollnet, M. (1987), Mundane Reason: Reality in everyday and sociological discourse, Cambridge: Cambridge University Press.
- 114. Public Construction Commission (2003) "Disputes factors and types analyses for mediation procedure of government procurement Research report submitted to the Public" Construction Commission of the Executive Yuan of Taiwan, Taipei, Taiwan (in Chinese).
- 115. Rahim, M. A. (1983) "A measure of styles of handling interpersonal conflict." Academy of Management Journal, 26(2): p. 368-376.

- 116. Rahim, M.A., Manger, N.R. and Shapiro, D.L.(2000) "Do justice perceptions influence styles of handling conflict with supervisors?: What justice perceptions, precisely"? *The International Journal of Conflict Management*, 11(1): p. 9–31.
- 117. Ren, Z., Anumba J. and Ugwu O. (2001) "Construction claim management: towards and agent-based approach." *Engineering, Construction and Architectural Management*, 8(3): p. 185-197.
- 118. Renn, O. (1998) "The role of risk perception for risk management". *Reliability Engineering & System Safety*, 59(1): p. 49-62.
- 119. Revay, S. (1992) "Can construction claims be avoided"? Building Research and Information, 12: p. 56-58
- 120. Rhys Jones, S. (1994) "How constructive is construction law?" Construction Law Journal, 10(1): p. 28–38.
- 121. Robbins, S.P. (1994) "Essentials of Organizational Behavior". New Jersey- Prince Hall.
- 122. Romano, C. (1989) "Research strategies for small business: a case study". International Small Business Journal, 7(4): p. 35.
- Rwelamila, P. D. and Meyer, C. (1999) "Appropriate or Default Project Procurement System". Cost Engineering, 41(9): p. 40-44.
- 124. Semple, C. Hartman, F. I. and Jergeas, G. (1994) "Construction claims and disputes: causes and cost/time overruns". *Journal of Construction Engineering and Management*, 120(4): p. 785-795.
- 125. Shen, L.Y.,(1997) "Project risk management in Hong Kong". International Journal of Project Managemen,t 15(2): p. 101-105.
- 126. Shleifer, T.C. (1990). "Construction Contractor's survival guide", John Wiley & Sons, Inc. New York, N.Y.

- 127. Songer, A. D. and K. R. Molenaar (1997). "Project characteristics for successful public-sector design-build". *Journal of Construction Engineering and Management* 123(1): p. 34-40.
- 128. Spalj G. T. (2005) "Construction Disputes". Construction Bulletin, October, p. 3-6.
- 129. Spittler, J. R. and Jentzen G. H. (1992) "Dispute Resolution: Managing Construction Conflicts with Step Negotiations". AACE Transactions. D.9: 1-10.
- 130. Sykes, J (1996) "Claims and disputes in construction suggestion for their timely resolution" Construction Law Journal, 12(1): p. 3-13
- 131. Sykes, J. (1999) "Construction Claims". Sweet and Maxwell, London
- 132. Tenah, K. A. (2000). "The design-build approach: an overview." *Cost Engineering* 42(3): p. 31-37.
- 133. The National Contractors Group (1990) "Building Towards 2001 Building Magazine" p.10.
- 134. Travers, M. (2001), Qualitative Research through Case Studies, Sage Publications, London.
- 135. Tsoukas, H. (1989), "The validity of idiographic research explanations", Academy of Management Review, 14(4): p. 551.
- 136. Uher, T.E. and A.R. Toakley, (1999). "Risk management in the conceptual phase of a project". *International Journal of Project Management*, 17(3): p. 161-169.
- 137. Vaaland T. I. (2004) "Improving project collaboration: start with the conflicts." International Journal of Project Management, 22: p. 447-454.
- 138. Vidogah, W. and I. Ndekugri, (1998) "Improving management of claims: Contractor's perspective." Journal of Management in Engineering, 13(5): p. 37-44.

- Wang, P C (2001) "Guidance for employers in coping with construction claims Pormosan". Brothers Law Journal, 9: p. 2-14 (in Chinese).
- 140. Wang, W.C., H.H. Wang, Lai Y.T. Li J. C.C. (2005) "Unit-price-based model for evaluating competitive bids". *International Journal of Project Management*, 26(2): p. 102-113.
- 141. Ward S, Chapman C, Curtis B.,(1991) "On the allocation of risk in construction projects". *International Journal of Project Management*, 9(3): p. 140–7.
- 142. Watts, V & Scrivener, J (1994) "Review of Australian building disputes settled by litigation. In: Construction Conflict: Management and Resolution (eds P Fenn & R. Gameson)", Chapman Hall, London; p. 209-219.
- 143. Weinberg, D. 2002, Qualitative Research Methods. Blackwell Publishers Ltd. Mas. USA.
- 144. Wilson, J.B. and Natalie, S.M. (2001), "'Qualitative' and 'qualitative' research : an analysis". *International Journal of Value Based Management*, 14(1): p. 1.
- 145. Wood K. (1975) "The Public Client and the Construction Industries", A Report of the Joint Working Party by the Economic Development Councils (EDCs) for Building and Civil Engineering 1st Ed. HMSO, London.
- 146. Wood, G.D. (2001) "Conflict Avoidance and Management, Postgraduate Course in Construction Law and Arbitration", Leeds Metropolitan University.
- 147. Yan, A Y M (2002) "Dispute management in construction: work towards dispute prediction and avoidance In: Proceedings of the Chinese Decision Science Conference, Taipei, Taiwan".
- 148. Yao, N J (2001) "The often seen public construction disputes and their prevention".*Taipei Bar Journal*, 265 p. 2-7.
- 149. Yeo, K.T. (1991) "Risks, Classification of Estimates and Contingency Management". Journal of Management in Engineering, 6(4): p. 458-470.

- 150. Yin, R.K. (2003), Case study research: Design and methods, Beverly Hills, CA: Sage.
- 151. Zack, J.G. (1995). "Practical dispute management". Cost Engineering. 37(12): 55.
- 152. Zack, J.G., Jr., (1996)"Risk-sharing—good concept, bad name". Cost Engineering, 38(7): p. 26.
- 153. Zaghloul, R. and F. Hartman,(2003) "Construction contracts: the cost of mistrust". International Journal of Project Management, 21(6): p. 419-424.

•
Appendix A: Chronological Listing of Research on Sources of Dispute (Fenn, 2006)

Researchers	Findings (sources of disputes)
Diekmann & Nelson	The two most common causes of contract claims were (1) design errors (46%) and (2) discretionary or
(1985) .	mandatory changes (26%). Other claim headings include: differing site conditions; weather; strikes and value engineering.
Mathews & Ashley (1985)	96 items were concluded as construction disputes within contract clauses
Watts & Scrivener	59 categories of disputes and 117 sources of disputes within which the subgroups are: (1) determination of the agreement,
(1992)	(2) payment, (3) the site and execution of work, (4) unite, (3) mean gence and musance, (0) muaticate payment.
Lewis et at. (1992)	Five causes of conflict: (1) one of the potential risk events occurs, (2) one or more of the parties suffers some
	but insufficient steps were taken to mitigate its effects, (5) the allocation of risks between the various parties to the contract
	was not clearly established in the first place.
Revay (1992)	Seven most frequent causes for claims: (1) inadequate site and/or soil investigation prior to starting the design, (2) starting design efforts and/or soil investigation too late prior to starting the design (3) calling for bids with an incomplete set of
	drawings, (4) endeavouring to complete design through shop drawing review, (5) introducing untimely design revisions
	without allowing commensurate time extension for the completion of the project or without recognizing the contractor's
	right to impact costs, (6) interfering both with the sequence and the timing of construction (e.g. to compensate for the delay
	in the delivery of owner-supplied equipment/material), (7) continuing to introduce changes under the disguise of correcting
	deficiencies.
Construction Industry	Six categories of main reasons for disputes: (1) general, (2) consultants, (3) client, (4) contractor, (5) subcontractors, (6)
Council (1994)	manufacturers and suppliers.
Dickman et at. (1994)	Three areas: (1) people, (2) process, (3) project.
Jergeas & Hartman	Well known reasons by which claims arise: (1) increase in scope of work, (changes, extras and errors), (2) inadequate bid
(1994)	information, (3) faulty and/or late owner-supplied equipment and material, (4) inferior quality of drawings and/or
	specifications giving rise to ambiguities in contact requirements, (5) insufficient time for bid preparation, (6) stop-and-go
	operations because of lack of coordination, design information, equipment, or material, (7) work in congested areas and
	overcrowding, (8) acceleration to regain schedule, (9) inadequate investigation before bidding, (10) unbalanced bidding and
	$\frac{1}{2}$
Lee (1994)	Disputes from contract problems: (1) untair contract clauses, (2) vague definition of contract documents (in terms of performance period, payment, quality and variations), (3) not comprehensive stipulation
Rhys Jones (1994)	Ten factors in the development of disputes: (1) poor management, (2) adversarial culture, (3) poor communications, (4)
	inadequate design, (5) economic environment, (6) unrealistic tendering, (7) influence of lawyers, (8) unrealistic client expectations, (9) inadequate contract drafting, (10) poor workmanship

Table 2 Chronological Listing of Research on Sources of Dispute (Fenn, 2006) (cont'd)

Researchers	Findings (sources of disputes)
Semple et at. (1994)	The most common contributing factors in claims are: (1) increase in scope of the work, (2) weather, (3) restricted access,
	(4) acceleration
Watts & Scrivener (1994)	290 sources of disputes are identified from 60 cases in each country and the 21 categories are grouped into five sub-groups. The most frequent source of disputes in the UK is negligence, while in Australia failure and determination have the highest
	occurrence
Assaf et at. (1995)	56 factors, which were grouped into nine major areas; (1) materials, (2) manpower, (3) equipment, (4) financing, (5) environment (6) charoes (7) covernment relations (8) contractual relations (9) scheduling and controlling techniques
Bristow &	Five primary causes of claims: (1) unrealistic expectations by the parties. (2) ambiguous contract documents. (3) poor
Vasilopoulous (1995)	communications between project participants, (4) lack of team spirit among participants, (5) a failure of participants to deal
	promptly with changes and unexpected conditions
Murdoch & Hughes	Background to disputes: (1) motivation factors of individuals, (2) pre-conceptions about roles, (3) project success or failure,
(1//0) Sukes (1996)	Two maior sources: (1) misunderstanding due to lack of clarity (2) unnredictability of unforeseen circumstance
advers (1770)	I WO IIIAJOI BOURCES. (1) IIIIBUILUCIDIUM AND NA TURN OF CHARLES, (2) AIPPEARMANING OF AIROTEDAN CITATIBUILO
Kumaraswamy (1997a)	Heads of claim categories: (1) cost, (2) time extension
Kumaraswamy	Two categories: (1) root causes, (2) proximate causes
(1997b)	
Hu (1998)	Factors of construction disputes: (1) defective performance in quality, (2) unfair contract clauses, (3) conservative attitude of public employer and supervising engineers, (4) negligence of design and supervision
Vidogah & Ndekugri	Eight heads of claims likely to be disputed (in rank order): (1) cost of disruption, (2) head office overheads, (3) interest and
(1998)	finance charges, (4) cost of preparing claims, (5) loss of profit, (6) inflation of costs, (7) on-site overheads, (8) others
Al-Momani (1999)	Seven categories: (1) poor design, (2) change orders, (3) weather, (4) site condition, (5) late delivery, (6) economic condition. (7) increase in quantity
Fenn (1999)	Three independent variables predicted construction disputes well: (1) the level of variations, (2) the success of the
	employer's advisors on a past project, (3) the tender period
Gould et at. (1999)	Nine areas of issues in dispute: (1) project delays, (2) change in the scope of the work, (3) payment issues, (4) differing site conditions, (5) design issues, (6) defective work or product, (7) site administration problems, (8) property
	damage, (9) personal injury
Liang (1999)	Four types of most frequently seen claims: (1) scope of work claim, (2) delay claim, (3) change of site condition claim/adverse physical conditions or obstructions, (4) acceleration claim
Sykes (1999)	Origins of disputes: (1) omission and unforeseen events, (2) lack of capacity to settle claims, (3) different expectations.

Researchers	Findings (sources of disputes)
Molenar et at. (2000)	Three factors found to have influence on or be closely related to dispute potential: (1) people issue, (2) project complexity, (3) owner management ability
Mitropoulous & Howell (2001)	Basic factors driving the development of disputes are: (1) project uncertainty, (2) contractual problems, (3) opportunistic behaviour
Kululanga <i>et at.</i>	Four basic sources: (1) contract documents due to errors, defects and omissions, (2) failure to appreciate the real cost of a project at the beginning (3) changed conditions (4) stakeholders involved in a project
Ren et at. (2001)	Three factors: (1) social, (2) industrial, (3) project
Tsai (2001)	Five areas of disputes: (1) performance period, (2) payment, (3) quality, (4) quantity, (5) contract
Wang (2001)	Grounds for claims: (1) unfairness of contract/unfairness of risk allocation, (2) variations, (3) defective contract documents, (4) delay claim, (5) circumstance changes, (6) breach of obligations of employers, (7) termination of contract
Yao (2001)	Disputes can be categorized into two stages; (1) before contract award, (2) after contract award
Kehinde & Aiyetan (2002)	Highest source of contractual claims in most building contracts is constituted by (1) variations, (2) additional works
Lo (2002)	Causes of construction conflict: (1) differences in goals and objectives of parties in the project, (2) differences in contract
~	interpretation between the construction manager and contractor that have to be compromised by serious negotiation which
	may take a long period of time, (3) lack of understanding about the needs of others also involved in the planning, design and
	construction process, (4) uncertainty about role, responsibility, authority and procedure ambiguity, (5) unclear reward
	structure or opportunity for the project participants, (6) specific allocation of limited resources such as materials, capital,
	labour, etc., (7) excessive demands on resources normally depended on to assist in the resolution of conflict, (8) incorrect
	assumptions made from biased perceptions, (9) demands for higher quality than specified, (10) failure to provide products in
	conformity with user requirements, (11) insufficient time to make required decisions, (12) inability to do the job, (13)
	subcontractor performance problems, (14) frustration over a lack of control of events affecting performance, (15) desire to
	take advantage of those in a weaker position, (16) adverse relationship between client and contractors, (17) work slowdowns
	and strikes, (18) interpersonal conflicts, (19) regulatory problems, (20) lack of communication
Yan (2002)	Construction disputes in Taiwan fall into four categories: (1) problems occur in planning and designing stage, (2) problems
	occur in contract performance stage, (3) problems occur in completion and acceptance stage, (4) problems occur in
	maintenance stage, ; four sources of disputes: (1) contractual factors, (2) technological factors, (3) external factors, (4)
	artificial factors
Chang & Ive (2003)	Natures of two types of disputes are: (1) pure cognitive dissonance, (2) opportunism intention to take advantage of one party's vulnerability

Table 2 Chronological Listing of Research on Sources of Dispute (Fenn, 2006) (cont'd)

Appendix B: Chronological Listing of Other Research on Sources of Dispute

Researchers	Findings (sources of disnutes)
Assaf and Al-Khalil (1995)	They conducted a survey in Saudi Arabia to study the causes of disputs over delay in large buildings. It was based on a questionmaire that was filled by 24 Contractors, 15 Consultants and 9 Public Owners. The authors identified an extensive questionmaire that making the suborase, changes in types and specifications during construction, slow delivery of abourts, attainoulds in storage, delay in the special manufacture of the building material, shortage of labour, labour skills, nationality of labourers, equipment failure, equipment shortage, unskilled operators, slow delivery of equipment, sexills, nationality of labourers, equipment failure, equipment shortage, unskilled operators, slow delivery of equipment, equipment productivity, financing by contractor during construction, delays in contractor's progress payment by owner, cash problems during construction, design changes by owner or his agent during construction, design errors made by designers, foundation conditions encountered in the field, mistake in soil investigation, water table conditions on site, geological problems on site, obtaining permits from municipality, obtaining permits for labourers, excessive bureaucracy in project owner optation, building code used in the design of the project, preparation and approval of shop drawings, waiting for sample material approval, preparation of scheduling networks and revisions, lack of training personnel and management support, lack of database in estimating activity duration and resources, judgement of experience in estimating time and ensources, project downers, slowness of the owner decision and process, the project, propertience in estimating time and ensources, propertience but evers, slowness of the owner decision and the romitation between owner and designer at the design phase, unavailability of professional construction management, inspection and difficulty of coordination between the various parties, insufficient communication between owner and designer at the design phase, unavailability of profess
Alkass et al. (1996)	They attribute disputes over delays in general terms to strikes, rework, poor organization, material shortage, equipment failure, change orders, acts of God.
Maden (2005)	He studied the causes of disputes and categorised them as follows: Legal: Existence of contract, notice issues, performance criteria, payment, interference, over-inspection, breach of contract, contract interpretation, condition precedent, site access, shop drawing processing. Technical: Compliance with plans and specifications, quality, changed conditions, performance, quantity, constructive changes, design error, warranty, incomplete design, supervision, incomplete design, effect of conditions on operations, use
Molenaar <i>et al.</i> (2000)	of equipment. Quantum: Total costs claim/modified, cost estimating, damage analysis, cost accounting, overhead, delay analysis. They considered the significance of the behavioural practice of the project participants and classified factors that affect the disputes in three categories: people issue, process issue and project issues. Each category comprises several project

	characteristics. Characteristics that were considered to be significant indicators of disputes were classified into seven hybrid variables mainly: owner management and organisation, contractor management and organisation, project complexity, project size, financial planning, project scope definition, and risk allocation. The writers conclude that this model can be similarly used in other types of analysis in the construction industry where the latent variables can be better measured through surrogate variables.
Mitropoulos and Howell (2001)	They move beyond individual factors and study the effect of interaction of technical, contractual and behavioural factors on developments of disputes. The research was based on 24 claims which occurred on 14 projects. For each claim the writers examined the development of disputes which was analyzed through studying (1) the issue and basis of initial disagreement, (2) attempts to solve the problem, (3) the type of magnitude of dispute that developed due to the failure to solve the problem and (4) the level of dispute escalation. Also, the amount claimed and settled and the levels of resolution were specified. The writers then identify three basic factors that directly affect disputes: project uncertainty, contractual problems and opportunistic behaviour. A model was developed inductively to understand and explain how disputes developed. The writers explain that the proposition of the model should be considered as a hypothesis for further research.
Essex (1996)	He focuses on disputes arising from unforeseen subsurface conditions. The author examines four improved contracting practices to minimize disputes resulting from unforeseen subsurface conditions mainly: geotechnical baseline reports, alternative dispute resolution, escrow bid documents and partnering.

Appendix C: Chronological Listing of Risk Categorization

Researchers	Categorization of Risks
Zack (1996)	Zack (1996) presents an exhaustive list of risk allocated in standard construction contracts that includes: physical risks, acts of God, impractical/impossibility, latent site conditions, quantity variations, site access, weather, capability-related risks, defective works, labour forces, subcontractor, supplier failure, economic risks, bonding, contract termination, cost escalation, economic disasters, failure to pay, insurance, project funding, taxes, time-related risks, acceleration, delays and disruptions, early use of facility, suspension of works, untimely responses, union strike, engineering and construction risk, changes, contractor furnished equipment/material, continuation of work, coordination, defective contract documents, interpretation of requirements, means and methods of construction, owner-furnished equipment materials, permits and licenses, productivity, site safety, and work quality.
Shen (1997)	Shen (1997) categorised the general risks as construction risks, legal risks, incorrect design information, shortage of materials/plant resources, poor accuracy of project programme, subcontractor's manpower shortage, variation in ground and weather conditions, abortive works due to poor workmanship, shortage of skills/techniques, and poor coordination with subcontractors. A survey was conducted in Honk Kong to study the Contractor's perception of the relative importance of risks.
Chan & Kumaraswamy (1997)	Chan & Kumaraswamy (1997) conducted a survey in Honk Kong based on 83 hypothesized delay factors based on the results of previous studies and interviews with local clients, consultants and contractors. The 83 factors were grouped into eight major factor categories: project-related, client-related, design team- related, contractor-related, materials, labour, plant/equipment and external factors. Respondents were required to indicate the relative importance of these factors. Results received from respondents in Hong Kong were compared to those in Saudi Arabia and Nigeria.
Akintoye & MacLeod (1997) Kartam & Kartam (2001)	Akintoye & MacLeod (1997) and Kartam & Kartam (2001) identified major construction risk categories risk categories as physical, environmental, design, logistics, financial, legal, political, construction and operation risks. Akintoye & MacLeod (1997) requested the survey respondents to indicate the importance set by their organisations to these risks. Contractors and Project managers agreed that the two most important risks are the contractual and financial risks. Contractual risk is associated with flaws in contract documents, inappropriate documents or improper contractual relationships. These risks normally result in claims and disputes, disruption of works, stoppage of work, lack of coordination, delays, and inflated costs. Financial risks on the other hand are associated with the owner's capability of providing the necessary cash flow. Financial capabilities are essential for the proper progress of the works.

Researchers	Categorization of Risks
Ahmed <i>et al.</i> (1999)	Ahmed <i>et al.</i> (1999) in comparing the Contractor and Owner perception of risk allocation in Hong Kong identify 26 different types of risk that include: acts of God, change in works, change order negotiations, changes in government regulations, contractor competence, cost of legal processes, defective design, defective materials, deficiencies in specifications and drawings, delayed payment on contracts, delays in resolving contractual issues, delays in resolving litigation, arbitration disputes, environmental hazards of the project, financial failure, inflation, labour and equipment productivity, labour disputes, labour, equipment and material availability, permits and ordinances, political uncertainty, quality of work, safety, site access right of way, suppliers subcontractors poor performance, third party delays, and unforeseen site conditions.
Pickavance (2000)	Pickavance (2000) classified risk in 10 main categories that include legal risk, dispute risk, design risk, buildability risk, procurement risk, biddability risk, construction risk, financial risk, political risk and insured risk.
Kartam & Kartam (2001)	Kartam & Kartam (2001) further defined 26 risk types that included permits and regulations, scope of work definition, site access, labour, material and equipment availability, productivity of labour and equipment, defective design, changes in work, differing site conditions, adverse weather conditions, acts of God, defective materials, government acts, accuracy of project program, labour disputes, accidents/safety, inflation, contractor competence, change order negotiations, third party delays, coordination with subcontractors, delayed dispute resolutions, delayed payment on contracts, quality of work, financial failure, actual quantities of work and war threats.
Bunni (2003)	Bunni (2003) presented a full spectrum of more than 50 types of risks that were categorised as feasibility, design, site and its location, technical aspects, and acts of man.

Appendix D: Extracts of the Particular Conditions on the 4 Projects

	Ω	The Engineer shall	obtain the specific	written approval of	the Employer before	taking any of the	following actions:	i. approving the	subletting of any	part of the works	in accordance with	Clause 4.	ii. Certifying	additional cost	determined in	accordance with	sub-clause 12.2	iii. Determining an	extension of time	in accordance with	subclause 44.1	iv. Issuing a variation	in accordance with	subclause 51.1.	v. Fixing of rates or	prices in	accordance with	sub clause 52.2	vi. Approving	equivalent	materials and/or	equipment.		
rticular	U	The Engineer shall	obtain the specific	written approval of	the Employer before	taking any of the	following actions:	i. approving the	subletting of any	part of the works	in accordance	with Clause 4.	ii. Certifying	additional cost	determined in	accordance with	sub-clause 12.2	iii. Determining an	extension of time	in accordance	with subclause	44.1	iv. Issuing a	variation in	accordance with	subclause 51.1.	v. Fixing of rates or	prices in	accordance with	sub clause 52.2	vi. Approving	equivalent	materials and/or	equipment.
Раг	В	The Engineer	shall obtain the	specific approval	of the Employer	before exercising	his rights under	the following	clauses:	7.1.1a & b, 7.1.2,	7.2, 8.2, 17.1,	36.7, 36.8, 36.11,	37.3, 37.4, 48.1,	48.2, 48.3, 49.2,	49.3, 54.8, 62.1	Hence the	Engineer shall in	no way and under	no circumstances	be responsible for	any approval,	consent and the	like neither on	selection of	materials,	equipment and	samples nor on	the approval of	construction	drawings,	coordination	drawings and as-	built drawings.	
	4	Provided that the	Engineer shall obtain	the specific approval	of the Employer	before exercising any	of the following	duties or authorities:	a)Approving an	extension of the	Time for Completion	of the Works	b)Approving any	additional payment	to the Contract Price.	c)Giving consent to	the Contractor to	Subcontract any part	of the Works	pursuant to Sub-	Clause 4.1.	d)Issuing an	instruction to	suspend works	pursuant to Sub-	Clause 40.4	e)Issuing an	instruction in respect	of any Provisional	Sum pursuant to Sub-	Clause 58.2.			
	General		(a) The Engineer shall carry out	the duties specified in the	Contract.		(b) The Engineer may exercise	the authority specified in or	necessarily to be implied	from the Contract. provided,	however, that if the Engineer	is required, under the terms	of his appointment by the	Employer, to obtain the	specific approval of the	Employer before exercising	any such authority,	particulars of such	requirements shall be set out	in Part II of these	Conditions. Provided further	that any requisite approval	shall be deemed to have been	given by the Employer for	any such authority exercised	by the Engineer.		(c) Except as expressly stated in	the Contract, the Engineer	shall have no authority to	relieve the Contractor of any	of his obligations under the	Contract.	
	Title	Engineer's	Duties and	Authority																														
	Clause	Clause 2.1																																

	Ω	If the Employer	intends to replace the	Engineer, the	Employer shall not	less than 42 days	before the intended	date of replacement	give notice to the	Contractor of the	name, address and	relevant experience	of the intended	replacement	Engineer.	The Contractor shall	have no right to	object to the	replacement of the	Engineer and will	have no ground for a	claim in connection	therewith.		
ticular	U	If the Employer	intends to replace	the Engineer, the	Employer shall not	less than 42 days	before the intended	date of replacement	give notice to the	Contractor of the	name, address and	relevant experience	of the intended	replacement	Engineer.	The Contractor shall	have no right to	object to the	replacement of the	Engineer and will	have no ground for a	claim in connection	therewith.		
Par	B				-																				
	A																							 	
	General																								
	Title	Replacement	of the	Engineer																					
	Clause	Clause 2.7																							

				Par	ticular	
Clause	Title	General	٨	B	c	۵
Clause 5.2	Priority of	The several documents forming	The several	1) The Contract	(1) The Contract	(1) The Contract
	Contract	the Contract are to be taken as	documents forming	Agreement and	Agreement (if	Agreement (if
	Dogimente	mutually explanatory of one	the Contract are to	the Annexes	completed);	completed);
	nocalitalitis	another, but in case of	be taken as mutually	2) The Tender and	(2) The Letter of	(2) The Tender;
		ambiguities or discrepancies the	explanatory of one	its Appendix	Acceptance-,	(3) Part II of these
		same shall be explained and	another and what is	3) Part II of these	(3) The Tender;	Conditions;
		adjusted by the Engineer who	required by any one	Conditions	(4) Part II of these	(4) Part I of these
		shall thereupon issue to the	document shall be	4) Part I of these	Conditions;	Conditions; and
		Contractor instructions thereon	required by all, but	Conditions	(5) Part I of these	(5) Specifications
	-	and in such event, unless	in case of	5) Addenda	Conditions; and	(6) Drawings
		otherwise provided in the	ambiguities or	excluding Bills	(6) Addenda	(7) The Bill of
		Contract, the priority of the	discrepancies the	of Omissions	excluding	Quantities
		documents forming the Contract	same shall be	and Additions	Breakdown of	(8) Breakdown of the
		shall be as follows:	explained and	6) The	the lumpsum	lump sum
			adjusted by the	Specifications	(7) Specifications	(9)Any other
		(1) The Contract Agreement (if	Engineer who shall	7) The Drawings	(8) Drawings	document
		completed);	thereupon issue to	8) Bills of	(9) Breakdown of	forming part of
		(2) The Letter of Acceptance-,	the Contractor	Omissions and	the lump sum	the Contract.
		(3) The Tender;	instructions thereon	additions issues	(10)Break down of	
		(4) Part II of these Conditions;	and in such event,	as part of the	unit rates	
		(5) Part I of these Conditions;	unless otherwise	Addenda	(11)Any other	
		and	provided in the	9) The Bills of	document	
		(6) Any other document forming	Contract. The	Quantities	forming part of	
		part of the Contract.	priority of the	10)Breakdown of	the Contract.	
			documents forming	Unit rates		
			the Contract shall be	11)Any other		
		_	as determined by the	document		
			Engineer.	forming part of		
				contract		

	Ω	No payment will be	made to the	Contractor on	account of increases	in rates of wages,	custom and import	duties of the market	prices of materials	and goods specified	for incorporation in	the works which	occur after	submission of the	tender, unless such	increase are a result	of Subsequent	Legislation as	defined in sub clause	70.2		The following	material shall be	exempted from the	hereabove:	1. The	reinforcement	steel	2. The sand	3. The gravel	4. The silica fume						
ticular	U	The Contract Price	shall not be subject	to any adjustment in	price in respect of	rise or fall in the	cost of labour,	materials, fuel or	any other matters	affecting the cost of	the execution of the	Works.																									
Par	Ω	Subject to	Subclause 70.2	the Contract	Price shall not be	subject to any	adjustment in	respect of rise or	fall in the cost of	labour, materials	or any other	matters affecting	the cost of	execution of the	contract.																						
	A	No adjustment shall	be made to the unit	rates, prices or sums	in the Breakdown of	the Lump Sum, nor	to the Contract	Price, nor to unit	rates in Daywork	Schedule, nor the	Schedule of Unit	Rates for Works of	Phase 3, in respect	of increases or	decreases in the	costs of labour,	materials, Plant,	Temporary Works,	Contractor's	Equipment or other	matters, including	but not limited to	variations in rates of	currency exchange,	affecting the cost of	execution of the	Works from the date	of signing the	Contract by the	Contractor.		The provisions of	this sub-clause shall	not apply to any	subsequent	legislation pursuant	to sub clause 70.2.
	General	There shall be added to or	deducted from the Contract	Price such sums in respect of	rise or fall in the cost of labour	and/or materials or any other	matters affecting the cost of the	execution of the Works as may	be determined in accordance	with Part II of these	Conditions.																										
	Title	Increase of	Decrease of	Cost																																	
	Clause	Clause 70.1																																			

lar	Ω	le Contract Price ull not be subject any adjustment in pect of sequent islation which uses additional or iced cost to the intractor in the cution of the intract.
Partici	В	Delete the words "other shi words "other shi 70.1" 70.1" sul- leg der can can der can can der can der
	A	
	General	If, after the date 28 days prior to the latest date for submission of tenders for the Contract there occur in the country in which the Works are being or are to be executed changes to any National or State Statute, Ordinance, Decree or other Law or any regulation or bye-law of any local or other duly constituted authority, or the introduction of any such State Statute, Ordinance, Decree, Law, regulation or bye-law which causes additional or reduced cost to the Contractor other than under Sub-Clause 70.1, in the execution of the Contract, such additional or reduced cost shall, after due consultation with the Employer and the Contractor, be determined by the Engineer and shall be added to or deducted from the Contract Price and the Engineer shall notify the Contractor accordingly, with a copy to the Employer.
	Title	Subsequent Legislation
	Clause	Clause 70.2

Appendix E: A Brief Overview of the Dispute Cases

PROJECT A

Dispute 1 - Lift Overhead: During execution of the works, an approved shop drawing is noticed to have a mistake in calculating the overall structure height on top of the lift area. With the correct calculations, the already ordered lift was found to be not adequate for the available space. Responsibility was disputed.

Dispute 2 – Façade False Ceiling: The Specification allows for 3 alternatives for facade false ceiling in the addendum to the contract. However, when the engineer chooses the more expensive option, the Contractor disputes the same stating that the price reduction was applied at time of signature based on a VE priced based on the least expensive alternative. **Dispute 3** – Procurement Date: A variation order is issued where the contractor is requested to procure new material different than that originally specified for parts of the works. The Engineer sets the rates based on the unit rate one month after the release of order. The Contractor argues that more than one month is required in certain cases to procure.

Dispute 4 – Façade Glass: The frit glass of the façade has a very general specification that only give details of percentage transparency but no pattern. Reaching the approved pattern requires the submittal of 31 patterns in 4 submittals. The Engineer argues that it is part of the contractor's obligation to provide an approved sample. The Contractor's request for additional cost and an extension of time is disputed.

Dispute 5 – War effect: War breaks in Lebanon in July 06 where the works are put on halt for security reasons. Even after the war ends there is a continuing effect due to loss of labor and productivity. The cost and time impact of the same are disputed.

Dispute 6 – Additional Shop Drawings for variations: At contract signature allowance is made for set of design modifications where the time frame for the issuance of these modification is set. However, the additional cost of preparing shop drawings due to the same is not mentioned in the contract documents. This creates an ambiguity in assessment that causes a dispute.

Dispute 7 – Façade Lighting: The façade lighting consist of an LED system along with a crown lighting system at the top of the building. The required lighting effect does not define the technical specifications of the lighting color. This requires multiple trial and error attempts to reach the required lighting effect which leads to dispute on the time and cost incurred.

Dispute 8 - Frit Glass: The Frit glass for the façade is specified in the contract to provide a 10 year guarantee. However, the Specification does not mention whether it shall be applied at Face 1 or Face 2 of the double glass. The Contractor considers the Engineer's request to have it on Face 1 as a variation. The Engineer insists on having the frit glass on face 1 to achieve the required façade design intent.

Dispute 9 – Employee Turnover: The Contractor claimed that there was employee turnover due to the war in the region. The Engineer rejects the claim on the basis that Employees are being lost due to the boom in the Gulf and this does not constitute an Employer's risk event. **Dispute 10** – Safety Film: A variation order was issued to remove the roller shutter and replace them with an anti-vandalism glass. Different alternatives for anti-vandalism glass were submitted. Examining the different options consumes time as the new specifications are not set clearly in the variation order. This leads to an extension of time dispute.

Dispute 11 – False Ceiling: The Engineer requests the Contractor to install C-channels to the false ceiling system. The Engineer considers the same as a requirement to achieve the stability. The Contractor claims that this is a designed efficiency and it is not the contractor's duty to design the system.

Dispute 12 – Lighting Fixtures: The Contractor submitted an alternative lighting fixture without samples. The Engineer requests samples and detailed comparison sheets of both the specified fixture and the alternative. Upon receiving the same the Engineer rejects the alternative as being not equivalent. The Contractor disputes the rejection as being not justified.

Dispute 13 – Increase Labor Rate: Following the war the Contractor claimed that due to the unstable political situation, there has been an ongoing increase in the labor rates. The Engineer disagreed as the increase of rate due to the war is only justifiable for a limited period after the 2006 war ended.

Dispute 14 – Façade Aluminum Color: The Contractor submits the RAL color for the façade Aluminum and sends a field inspection request to the Engineer. The Engineer requests the mockup to be relocated and then requests additional mockup submittals for two other colors. The Engineer approves the original mockup after 105 days. The contractor requests an extension of time due to the same which the Engineer rejects.

Dispute 15 – Steel Structure Design: The Contractor submits the roof steel structure shop drawings reply of which are delayed for more than 3 months. The Contractor sends an alternative for painting instead of hot dip galvanizing. The Engineer accepts the same rejecting any delay effect as there are estimated time savings in the new accepted proposal. The contractor disputes the same.

Dispute 16 – Trench Heaters: The Contractor submits marble shop drawings which are approved. After tiling starts the Engineer instructs that the trench heaters should be moved away from the façade to prevent heat buildup and possible glass breakage. The Contractor considers the same as a variation. The Engineer argues that it is the Contractor's responsibility to coordinate and ensure the performance of the system.

Dispute 17 – Shop drawings submittal schedule: The Contractor submits a material and shop drawing submittal schedule as requested by the Contract. The Engineer does not approve the same and requests leveling of the number of weekly submittals. The Contractor resubmits the same several times and no agreement is reached. Several notifications of possible delay due to delayed reply of submittals follow during the project lifecycle which the Engineer rejects on the basis that there is no approved submittal schedule.

Dispute 18 – Marble Works: An inspection for the marble cladding around the bathtub is submitted. The Engineer inspects the same several times and requests several modifications which are implemented by the Contractor. The Contractor considers that modifications took place through a trial and error procedure which entitles the Contractor for an extension of time.

Dispute 19 – EDL: The Engineer approves the transformers giving clearance to buy these transformers on condition that EDL approval be obtained for the same. The EDL room file is delayed due to several modifications along with design error in the original EDL file. The Contractor claims his entitlement for an extension of time due to the several modifications to the permit file and the EDL delay in approving the revised file which is an Employer's risk. The Engineer considers the same to be part of the Contractor's responsibility to coordinate. **Dispute 20** – PS for Health Club: The works for this provisional sum are released and then deleted. The Contractor claims for abortive works. The Contractor's entitlement to overhead and profit in the deleted items is disputed.

PROJECT B

Dispute 1 – Rejected NPVs: The Engineer rejects the Contractor's NPV's for additional scope of work on the basis that these are part of his duty to develop construction drawings. The Contractor disputes the same as it is a result of erroneous details in Contract Drawings. **Dispute 2** –Tree Damage: During the execution of the works, an accident occurs which results in permanent disfiguration of a tree. The Engineer informs the Contractor that an amount of money will be deducted from his account for compensation. The contractor argues that compensation in the Contract is allowed for the case of "major damage" considering that to mean "death of tree".

Dispute 3—Optional Works: The Engineer instructs the Contractor to carry out optional works as per the rates submitted by the Contractor during Tender. The Contractor argues that these rates are not applicable anymore as the Engineer did not instruct these works at Tender approval.

Dispute 4 – Cleanouts: The Contractor claims that the Engineer's request to add cleanouts on the Construction Drainage Drawings entitles him for a variation. The Engineer states that the Contractor is under obligation to provide the same as these are required by the National Plumbing Code specified in the specifications.

Dispute 5 – Fire Alarm System: The Engineer issued comments to the Contractor's Shop Drawings for the Fire alarm system that requires adding fire detectors, emergency bulbs, etc. The Contractor considered the same to be a variation. The Engineer disputed that these are part of the Contractor's duty to develop construction drawings that satisfy code requirements. **Dispute 6** – War Delay: After the War the Engineer assesses the extension of time due to the war as 55 days. The Contractor argues that the Engineer ignored the governmental decree that gives an extension of time of 5 months. The Engineer stated that the decree is not applicable for construction projects. The Contractor disputes the same.

Dispute 7–Soft Landscaping: The Engineer informed the Contractor that the Employer wishes to furnish the planting works at a lump sum amount. An agreement is reached and the Contractor submits a notification of possible variation at the agreed value. After a month the Contractor submits a revised amount that the Contractor rejects.

Dispute 8 –Smoke Detectors: The Engineer required access doors next to the motorized fire and smoke dampers as per code requirement. The Contractor disagreed with the Engineer's

reading of the specifications confirming that these access doors are not required by specifications. The Engineer disagrees to the same and reiterated his request.

Dispute 9 – Fire Fighting System: The Engineer requests the Contractor to submit the Sprinkler flow calculations. The Contractor disputed the Engineer's reading of the Contract requirements insisting that the hydraulic calculations are not part of his scope of work. **Dispute 10** – Towel Dryers: The Contractor had priced towel dryers based on a certain model that satisfies the specifications. The Engineer changed the model and assessed the Contractor's entitlement to the market rate. The Contractor disputes that the profit margin in the original BOQ rate should be maintained.

PROJECT C

Dispute 1 – Louvers at roof: The Contractor claims for extra payment for pitched louvers at roof as he has priced to use straight louvers. The Engineer refuses the same as these pitched louvers are necessary for the pitched roof, the straight louvers would not satisfy the technical requirements.

Dispute 2 – Chillers: The Engineer assesses the additional cost for increase of capacity of chillers instructed based on prorated contract unit rate. The Contractor disputes the same and requests actual market rates.

Dispute 3 – Restoration: During demolition of the building unforeseen conditions of the existing structure necessitated the use of new bracing systems for which the Contractor claimed entitlement for additional cost. The Engineer rejects the same as the restoration of the existing building is considered to be the Contractor's responsibility.

Dispute 4 – Glass Balustrade: In answering the request for information regarding the material of the support of the balustrade, the Engineer states that it should be in stainless steel. The Contractor considers the same to be a variation. The Engineer disagrees as the specification for the railing system is clear in that respect.

Dispute 5 – War effect: During the war the Contractor and his staff took the risk of attendance at site offices. The Engineer assessed the extension of time the Contractor is entitled to accordingly. An agreement was reached in this regard. However, the cost implication of the same is disputed.

Dispute 6 – External Works: The Contractor requested missing external works/drainage layout. The Engineer was delayed 3 months in providing the same. The Contractor stated that the layout is still missing.

Dispute 7 – Stone Flooring: The Engineer changed the type of granite specifying the new type. The Contractor submitted the samples. After 60 days the Engineer requested a different type. The Contractor claimed for delay due to the same.

Dispute 8 – Submittal Schedule: The Engineer fails to reply to the Contractor's shop drawing and material submittals in a timely manner. The Engineer argues that these submittals were not submitted in a timely manner as stipulated in the Contract as per the consented programme. For this reason major delay is incurred for long lead items among others.

Dispute 9 –Aluminum Composite Panel: The Engineer requested that the aluminum composite panels be replaced by plaster and paint. However, the instruction was not clear in certain areas which caused disruption delays to the fabrication and execution for which the Contractor claimed.

Dispute 10—Points of Drainage: The Contractor requested the Engineer to issue a complete developed coordinated design showing the points of drainage. The Engineer argued that the Contractor could have detected the drainage location easily had he coordinated the shop drawings.

PROJECT D

Dispute 1 – Access to Roof Area: There has been a series of missing design information requested from the Engineer one of which was the unresolved access to the Roof Area along with direct verbal instructions which caused delays and disruptions to the programme. **Dispute 2** – Raised Flooring: The Engineer requested a quotation for raising the floor at the higher levels of the building after concrete works started. The Contractor stated that the Hourdi Blocks required are not available in the market and that the activity itself will necessitate a longer period. The Engineer then requested the Contractor to proceed instructing him to use a different type of Hourdi Block that is available in the market. The delay effect of this instruction was disputed. **Dispute 3** – Additional Fireplaces: The Engineer instructed making provisions for real fire places. This necessitated coring the reinforced concrete. Late feedback from the Engineer caused disputed delay. The cost implication of the same was also disputed.

Dispute 4 –New Kitchen: The Contractor was requested to execute new kitchen details. The method of payment for these additional works was disputed as the Engineer considered it to be remeasured work while the contractor considered it is a variation order. The Employer interfered to reach a settlement against this matter which dragged as a dispute.

Dispute 5 –Aluminum Works: The Contractor submits a shop drawing which gets rejected by the Engineer. The Contractor argues that this rejection is unjustified and that this rejection was intended to prove a delay by the contractor when the delay in fact is due to a variation added on these shop drawings.

Dispute 6 – Lighting Works: The Contractor requested missing information regarding the audio visual wall outlets. The Engineer issued an A3 sketch that is not sufficient. The Contractor considers this as a failure of the Engineer to provide information of time which is delaying the works.

Dispute 7 – Labor Decrease: The Continuous political instability in the country caused loss of laborers. As such even on days where the log showed high levels this did not give a clear indications as most of the laborers left during the day. Calculation of this delay impact was disputed.

Dispute 8 – EDL Room: The Engineer sent the Contractor of supplier for the transformer room. The Contractor claimed additional cost for the unloading and placing of the transformers insider the EDL room. The Cost of the same was disputed.

Dispute 9 – Fire Rated Door: the Engineer requested 2hr fire resistant system in basement 4 however details of the same were missing. As such the Contractor informed the Engineer that the area was put on hold. Later the Engineer cancelled these items. Both parties disputed the time implication of the same was disputed.

Dispute 10 – Late Issuance of Drawings: Direct communication between the nominated subcontractor and the Engineer to agree on details in specific areas. The lack of detail caused continuous delay for which the Contractor is claiming.

Appendix F: Dispute Factors identified in the Dispute Cases

Tender

1 Design Errors have higher occurrence in project A for the cases of the lift overhead that can not be executed as per design, the trench heater that shows an architectural detail that does not guarantee the functionality of the system and the EDL permit file that is erroneous in calculating the required transformer load. Other cases of design error include several design discrepancies in project B that led to the contractor incurring additional cost, erroneous design for external works and points of drainage in project C and missing access to roof in project D. The cases confirm that these design errors have induced disputes. However, it is difficult to further construe whether the cause of these design errors is related to short design period, negligence by the designer or lack of experience. It is worth noting that these errors were not noticed or raised by the Contractor at tender stage.

2 Contract documents not clear: Two cases of not clear contract document include unclearness in the specification of technical requirements such as the external façade detail in cases 4 and 8 of project A and louvers at roof in case 1 of project B. The second case of unclearness relates to unclear descriptions for the allowed number of submittals per week, the method of calculating the additional cost of new material procured as per late instruction and time validity of optional works included in the contract.

3 Clarification during contract negotiation include cases of clarification of unclear specification requirements or clarification of contractor's obligations that are by common practice understood to be part of the Contractor's obligations but are not clearly stated so in the Specification requirements.

4 Not achievable requirements occurred in the cases of variations that failed to specify what is required in terms of execution.

<u>Risk</u>

5 Clear allocation and highlighting of responsibility/obligation was identified in 11 cases of unclear allocation of responsibility for providing design or even the execution of particular systems. For example in case 4 of Project A there was a clear disagreement to

whether the Contractor should provide the design of the fritting of the external glass or whether it was the Engineer's responsibility to provide this design. Also in case 16 of Project A, there was a clear disagreement to whether it was the Contractor's responsibility to coordinate the trench heater system with the glass and curtain system to ensure proper functionality of the system. The dispute factor of 'clear allocation and highlighting of responsibility' is noticed to be more of a problem in project B where it is identified for 6 out of the 10 disputes. It is worth noting that this dispute factor can be directly correlated with the dispute factor 'evading responsibility by blaming the other party.

6 Contractor failure to satisfy specification requirements appeared as a dispute factor in 6 cases. All cases referred to contract specifications except for the case of lighting fixtures where the contractor failed to meet the specifications transmitted as a variation to the original design and in this case the designer was unable to provide technical details.

7 Price escalation was identified as a dispute factor in four cases namely the cases of 'procurement of new material', 'employee turnover' and 'increase in labor rate' in Project A where the additional cost incurred was augmented due to ongoing price escalation. Also, the case of Optional Works in Project B where the Contractor is requested to execute optional works priced at tender stage. In the case of Chillers in Project C, the Contractor's is claiming for new market rate. Price escalation cases examined included both cases of increase in material rates and increase in wages. The occurrence of this dispute factors is correlated with that of 'Contractor avoiding monetary losses'. It is worth noting that although price escalation was only identified in 4 out of the 50 cases, the occurrence of such price escalation during the construction period of those projects could had an indirect effect on the contractors incurring financial losses and accordingly on their defensive behaviour in some cases to avoid monetary losses. 8 Permits regulations was a dispute factor in 2 of the 50 dispute cases examined. In the case of lift overhead the responsibility of abiding by permit height regulations while developing shop drawings was deemed by the Engineer to be the Contractor's responsibility. Whereas in the case of EDL, the error in permit file submitted that did not

satisfy permit regulations was the Engineer's responsibility.

9 Submittal schedule was raised as a separate dispute case in Projects A and C. It was noticed to be a dispute factor in case of Fire Fighting System in Project B where the Engineer made reference to the Contractor's failure to abide by the submittal schedule requirements. As such in 3 of the 4 projects there was a problem in submitting, approving or abiding by the submittal schedule which is a contract requirement.

10 Failure to notice the technical problem refers to both parties failure to notice the technical problem which occurred in the cases of Lift Overhead and Trench Heaters in Project A. The late perception of the technical problem resulted in delay.

11 Contractor avoiding monetary losses was recorded as a dispute factor in 18 cases, 5 of which were related to price escalation (mentioned earlier). The remaining 13 occurrences related to cases of occurrence of the dispute factors: 'Contract documents unclear' and 'clear allocation and highlighting of responsibility/obligation' except for the case of glass balustrade where the Contract specification is clear but the Contractor insists on claiming the additional cost.

Behaviour

12 Lack of experience was identified in 8 cases 7 of which were in Project A. This could give an indication that both parties on project A had a problem with lack of experience in architectural details. This does not necessarily mean that the parties involved in Project B, C and D were more experienced noting that Project A was adopting some of the state of the art systems that were not yet familiar. This dispute factor can be correlated with the 'trial and error attempts/approach'.

13 Human error/negligence was identified in 4 cases. In the cases of Frit Glass in Project A and Tree Damage in Project B negligence was the Contractors' fault. In the case of Raised Flooring it was the Engineer who was negligent towards the concern raised by the Contractor. In the case of Lift overhead both were equally negligent.

14 Trial and error attempts/approach was as mentioned earlier correlated with lack of experience where the Engineer was unable to make decisive replies to the shop drawing and material submittals before different trial and error attempts were made. All 6 cases identified were on Project A. As mentioned earlier this can be partly explained by the state of the art systems used on Project A.

15 Improper communication channel was only witnessed in the case of Late Issue of Drawings in Project D where the Engineer passed the information directly to the subcontractor.

16 Lack of cooperation refers to cases where attempts to clear the disputed matter could have been resolved earlier and delay to the works would be avoided had the parties sat together in a dedicated technical meeting.

17 Unwillingness to resolve disputes was witnessed in three cases. The first case is the 'Additional shop drawings for agreed late variations' where the agreement regarding the same was reached after contract signature. And where the Contract does not attend to the additional cost incurred, both parties could have agreed to the same and resolved the dispute.

18 Language expressing ill perception of the other party's intentions was found in the written correspondence examined in the cases of 'War Delay' and 'Soft Landscaping' in Project B where the Contractor accused the Engineer by stating: 'it seems that you are willing to take position with respect to the interpretation of the law'. And in the case of 'Submittal Schedule' in Project C where the Contractor accused the Engineer of 'holding back his replies to submittals in anticipation of forthcoming changes/variations or revision of the Works'.

19 Evading responsibility by blaming the other party was witnessed in 17 cases: 6 on Project A, 7 on project B and 4 on project C. These were mostly correlated with cases of 'Contract documents unclear' and 'Clear allocation highlighting of responsibility/obligation'.

20 Engineer firm although contract is grey was noticed in 13 cases: 7 cases on Project A, 4 projects on project B, 1 case on Project C and 1 on Project D. Although the Contract is not clear in most of these cases, yet the Engineer firmly reiterates his position. In the case of the Façade False Ceiling and the case of the New Kitchen in project D, the Employer interferes to accept a variation that is being firmly rejected by the Engineer.

21 Influence by the Employer is marked in 6 cases 4 of which are on project B and are cases of Employer's late reply due to interference in assessments and approvals of variations. These 6 cases refer to cases where the Employer's interference was a dispute factor. There might be further cases where the Employer interfered to resolve the dispute

which are not identified as such. It should be noted that the cases of Employer interference witnessed in these dispute cases are the ones that were identified through correspondence/ documents examined and do not necessarily reflect all cases of Employer interference.

Contract Administration

22 Late issue of missing design/variation is a dispute factor in 13 cases. There are 3 cases in project A and 3 cases in project C. However, this dispute factor is prevalent in Project D where it is witnessed in 7 out of the 10 dispute cases. This dispute factor shows the Engineer's failure to meet his obligations. Also, this dispute factor is correlated with Assessment of Delay.

23 Validity/Assessment of Variation is the dispute factor with the highest occurrence. 29 cases out of the 50 are related to disagreement regarding the validity or assessment of the variation order. In some cases disagreement regarding the validity is related to contract documents being unclear. In other cases it is the valuation of the variation that is being disputed.

24 Assessment of war claim is witnessed in all four projects since all four projects were ongoing during the recent political instability and the war hostilities that followed. Assessment of war emerging as a dispute factor in all four projects signifies that the clause for assessment of contractor's entitlement for extension of time and compensation for this Employer's risk is causing dispute as both parties are disagreeing to its interpretation.

25 Lack of Proper Management/Monitoring by Engineer is prevalent in project A (8 cases of 20) and project D (5 cases of 10). This could signify bad management by the Engineers on these two projects. These two projects are also the most prevalent in the case of the dispute factor 'assessment of delay.'

26 Assessment of delay is the dispute factor with the second highest occurrence. As mentioned above it is more prevalent in projects A and D. It has been correlated with many of the dispute factors mentioned earlier since it is a normal consequence of late attendance to obligations by both parties that will in most cases lead to disputed assignment of responsibility and assessment of those delays.

27 Late approval of submittals by Engineer was noted in 16 cases. Although the Engineer in many of these cases referred to the Contractor's failure to submit/abide by the submittal schedule, this does not relieve the Engineer from his/her responsibility to fulfill the contractual obligation in the best interest of the project.

28 Contractor late/missing submittals is a dispute factor that is prevalent in Project A and all 7 cases of contractor late/missing submittal resulted in a dispute in 'assessment of delay'.

29 Slow attendance to responsibilities these are cases that do not have contractual timelines but even where they are not critical they would be consuming the float and any delay thereafter becomes a critical delay. This is confirmed by the high correlation with the dispute factor 'Assessment of Delay'. It is evident in Projects A, C and D but in none of the cases in Project B.

30 Late intervention of subcontractor is a dispute factor that is mainly caused by the Contractor's late assignment of subcontractor or the poor management of subcontractors. The 3 cases that revealed this dispute factor occur in Project A and are related to the lift overhead case that involves the lift subcontractor and the façade glass and frit glass cases that involve the façade subcontractor.

31 Contractor poor coordination between trades is witnessed in 3 dispute cases: Trench Heaters and EDL in project A and points of drainage in project C. All three occurences have caused delay and the assessment of this delay was disputed.

Appendix G: Dispute Cases (Chronology of Events)







5 May 07 The Contract allowed for a mechanism for issuing late variations. However, since the contractor will incur increase in cost due to these late variations and the contract does not account for inflation it was agreed that the new items resulting from these variations will be assessed based on market rates.




The July war broke in Lebanon

CS no: 5



agreed so that it doesn't affect the contractor's progress of work

The contract agreement stated that the contractor will prepare the corresponding shop drawings.



7 Jan 08 The Contractor replies to the engineer's letter point by point:

1. the alternative has been submitted as an equivalent and not as a substitute. 2.noted

3. the conditional approval based on a commercial offer is not agreed to and is rejected as an approach.

4. the contractor stated that the specifications clearly states that the light should be more intense on certain floors

5. The contractor rejects the statement that 'the specified system achieves lower levels of lighting'.

6. The fact that the vertical and horizontal members are approved as noted until the full lighting of the crown of the building is approved is meaningless as it does not allow the contractor to proceed with the order.

7. the engineer has approved to have the second alternative examined and studied

The contractor reiterated that any cost impact and increase in rates due to delayed approval will be claimed.

13 Dec 07 The contractor comments to the engineer's reply stating that there is a contradiction between the approving the alternative as a valid option, giving the AAN status, and the fact that the final decision will await the crown inspection. As such the contractor can not proceed with ordering the same and the supplier has given notice that the prices will increase. Since the alternative is a valid option the request for the specified system mockup of the crown is not justified. All the additional fixture samples are resulting in additional cost for which the contractor will claim.

13 Dec 07 The engineer replied to the contractor's letter with the following: 1, noted

2. the engineer stated that the frit glass installed is not that of the ring level.

3. the approval of the alternative is conditional to a commercial offer submitted for a requested variation of additional items to be ordered.

4. The engineer stated that the original specification requirements calls for identical vertical and horizontal lighting.

5. The specified system achieves lower levels of lighting. The same should be met by the alternative.

6. The vertical and horizontal members are approved as noted until the full lighting of the building is approved.

7. noted

4 Dec 07 The contractor documented in a letter the discussions held during the inspection and during the technical meeting made

1. The engineer is satisfied with the lighting output of the alternative

2. the engineer requested the glass mockup to be extended to identify the ring effect and confirm that vertical and horizontal light are homogeneous

3. The contractor requested an approval for the alternative to facilitate request of additional samples however the engineer insisted on examining the horizontal mockup first

4. The engineer requested that the horizontal lighting match the vertical lighting. The contractor said this was a variation to the original specification requirements.

5. The engineer expressed his wishes to have the lighting level for the LED lowered not to disturb the tenants. This should be submitted in addition to the specified.

6. The contractor requested that the crown lighting be separated from the vertical and horizontal members and to approve the alternative. The engineer said: "he would look into the matter and give his official reply after consulting all concerned parties"

7. the specialist confirmed that the specified system can not provide the requested effect but the modification will have a cost impact. The engineer requested the specialist to give his recommendation.

1 Dec 07 (after 22 days of the mockup) The engineer replies with ANR commenting as: the mockup should be resubmitted with the approved frit glass.

1 Dec 07 The engineer comments that more lighting fixtures for the test should be provided and approved of – the horizontal lighting shall be provided after the inspection of the mockup – the final decision on the fixture will also depend on the crown – the lighting of the façade is AAN

30 Nov 07 The contractor submitted a field inspection request for the façade lighting system

27 Nov 07 (after 49 days) The engineer replies to the material submittal with ANR requesting: certificates for manufacturing and country of origin, certificate indicating the testing results, certificate of guarantee, mockup sample for the proposed system A and the specified system B (after 49 days).

09 Nov **07** The contractor submitted the proposed layout of LED's on external skin mockup for these light fixtures.

11 Oct 07 The contractor submitted the LED light fixtures



3 Dec 07 The engineer stresses on the fact that the frit glass should be on face 1. He specifies suppliers that are using the same on different projects. The engineer adds that if the current supplier can not provide the same he should be replaced.

1 Dec 07 Engineer replies to inspection requesting fritted glass to be on face 1 as requested in the reply to the material submittal in Jan 07 – the alternative of positioning the frit between the glass panes will be examined on the mockup.

28 Nov 07 The contractor requested the engineer to attend a field inspection request for the external glass mockup.



30 Nov 06 After the war in Lebanon and due to the unstable political situation the contractor claimed that country witnessed immigration of engineers and other human resources on site. This resulted in a slow down to the progress of works. This was raised by the Employer as an Employer's risk

	CS no: 10		Subject: Safety Film			Project: A	Section: Aluminum Façade	
	Yr 1			Yr 2		Yr 3	Yr 4	
y June	Sep.	Jan	April	July	Oct. Jan.	April Jul Ozt.	Jan. April July Oct.	
					10 A notifi the en 14 M safety 7 Marc polycar 28 Feb option o 27 Feb 0 performa 27 Feb 0 performa specifica determin time cons 13 Feb 03	pril 08 The Contrication of possible ngineer to reconsider film ch 08 The contract rbonate system he 08 The Engineer of polycarbonate. 08 Engineer replies ance and available 08 The Contractor ance data is not tions can not be ac e glass specifications suming and subject 8 The Contractor	actor reiterates his position reg e delay regarding this matter ar der. ontractor resubmitted the anti- or replied that with the cancellar will resubmit the 12mm tempered requests the contractor to dist is to material submittal with ANI warranty period of the proposed replies to engineer's letter: poly at available, tempered glass hieved, a test can alternatively b on/performance but this testin t to a VO.	arding the id request vandalism tion of the ed glass regard the R security films rearbonate meeting be made to g will be I with the
					film optio 29 Jan 08 the progres	n and technical att The Engineer rejec s of works	achments of 3 options eted the NPD as this matter is no	t delaying
					23 Jan 08 investigated	The Engineer further and propo	replied that polycarbonate cla sal for PVB should be fully docu	nd should amented
					2 Jan 08 ollows: heat behavior of vell known, round the p xpensive op	The Contractor r toughened glass of combination of te glass supplier is anel to hold the p tion	eplied to the engineer's sugget can not be specified as antivand empered glass with polycarbon proposing relying on bolt an anel in place, PVB can be an a	estions as alism, the ate is not nd sealant alternative
				8.	Jan 08 Cont	ractor issues a noti	fication of possible delay	
				13 ant	Dec 07 (ivandalism r	Contractor submit material submittal	ted shop drawing (4 th subm	ittal) and

19 Nov 07 The Engineer suggested that the Contractor investigates: heat strengthened instead of annealed or glass clad polycarbonate.

17 Nov 07 The Contractor states that following the Engineer's proposal in a technical meeting he has investigated the option of having a film a applied on the glass but the two suppliers contacted were not able to provide full technical backup documents that meet specs.

25 Oct 07 The Engineer replies that the Contractor should investigate possible solutions. The consultant will do the same in parallel.

16 Oct 07 The Contractor comments to the engineer's reply stating that the Engineer is requesting the glass to be anti-vandalism although this glass is to be drilled so it has to be tempered instead of annealed and can not perform as antivandalism.

5 Oct 07 The engineer replied with ANR and AAN

24 Aug 07 The contractor resubmitted the shop drawing

19 May 07 The engineer replied with ANR (after 26 days)

23 April 07 The contractor resubmitted the shop drawings

31 Jan 07 The engineer replies with ANR and NA (41 days later)

¹21Dec 06 The contractor submitted the shop drawings for the spider glass façade

9 Nov 06 Engineer issues a variation order in shop fronts to cancel roller shutters and upgrade the spider glass into antivandalism



17 Aug 07 The contractor submitted the modified shop drawings as per new comments.

21 July 07 the engineer replies that there is no objection to the proposal to modify heights all -10cm inverts should be modified to -14cm.

18 July 07 the contractor states that the minimum clear height of 10cm is required above fibrous gypsum false ceiling panels otherwise it could not be installed.

6 Jun 07 the Engineer (after 22 days) replied with AAN changing the clear height in AC grille area from 19cm to 17cm and modified the cornice head and height in areas of electrical fixtures from 13cm to 12cm. Along with a list of general comments that were not mentioned in previous submittals.

1 June 07 the engineer replied to the RFI commenting to the proposed detail and requesting the contractor to resubmit.

21 May 07 the Contractor submitted a request for information regarding shifting the position of the grille from the lower drop to the upper drop.

15 May 07 the contractor resubmitted the details incorporating the engineer's comments.

After several technical meetings with the engineer regarding the AC grille opening and the electrical fixtures location it was agreed that AC grill opening will have clear height of 19cm and 13cm clear width where electrical fixtures are located.

28 Dec 06 the Engineer replies to the second submittal with AAN changing the height of the cornice where the AC grille is located from 23cm (coordinated with MEP) to 20cm including the gypsum thickness

26 Dec 06 the Engineer replies to the first submittal changing the height of the cornice where the AC grille is located from 22cm to 16cm

22 Dec 06 Contractor submits the general details for the false ceiling (classic and modern details) via 2 submittals











trench heaters intercepting with the stone tile are not yet submitted. The dummy grills are not yet determined and samples are not yet provided.

27 Sep 07 The Contractor issued a notice of possible delay due to the abortive engineering works and time delay in tiling works

04 Aug 07 The Engineer replied to the document transmittal stating that proposed solution is approved and confirmed the 77cm distance between trench heaters and the window. Shop drawings are submitted and the marble tiling proceeds in all areas accordingly.

01 Aug 07 After several joint technical meetings, the Engineer's comment to the tiling shop drawings regarding the trench heater locations, the contractor submitted a typical detail via document transmittal

06 June 07 The Engineer replies with ANR (75 days later) requesting contractor to move the trench heater 35cm from the edge of the aluminum panels and stating that the trench heater width should be confirmed and coordinated with mechanical material and shop drawings.

23 March 07 The Contractor submitted shop drawing for further areas/apartments.

19 March 07 The Engineer replied to the following: The location of the trench heating radiator as shown on the attached installation detail is acceptable thus approved.

15 March 07 The Contractor sends an RFI regarding the detailing of the trench heater showing the trench heater jamming with the aluminum window as per the design drawings.

2 Sept 06 The Contractor proceeds with submitting and resubmitting shop drawings for marble flooring in several areas/apartments that are approved and starts execution of the electrical embedment accordingly.





10 Aug 07 The engineer replied to the FIR with ANR requesting the contractor to change the corner details which caused the contractor to resubmit the shop drawing for typical bathroom detail

9 Aug 07 The contractor requested the engineer for an FIR

20 Feb 07 (after 23 days) the shop drawing is replied as "AAN"

19 Jan 07 The contractor submits typical bathroom tiling shop drawings



Subject: EDL Project: A

Section: Electrical



20 Feb 07 the Engineer issued the revised LV metering configuration

8 Feb 07 the Engineer replies that in case of no reply the contractor is requested to proceed with what is commonly used regarding bus ducts. The final LV metering configuration was promised to be issued soon.

26 Dec 06 the contractor replies that EDL is not responding in providing written confirmation. As for the LV metering configuration, the Contractor questions how it relates to the MV mechanical equipment load. Also, the contractor pointed out discrepancies in the LV metering configuration have been submitted.

9 Dec 06 the Engineer requests a written confirmation regarding the Bus ducts being not permitted. The Engineer also provides the metering configuration but states that since the contractor has not yet submitted the power rating of the final mechanical equipment the metering configuration can not be finalized.

6 Dec 06 the EDL informs the contractor that the permit file is not complete and can not be processed. Thus further documents are necessary. Also, EDL points out that the sizes of the transformer capacities have been changed. The big transformers necessitate having spare transformers.

05 Dec 06 the Contractor visits the EDL to pursue the permit file and is informed that the permit drawings are automatically processed by the EDL engineering department

17 Nov 06 The contractor conducted a visit to EDL to followup on the pending letters that were not responded. The EDL coordinator reiterates that the bus ducts are not permitted, as for the approval to the new permit file it has not yet been approved as the Engineer has been requested to issue the EDL meter configuration to be adopted.

10 Nov 06 the contractor replied that since this VO is issued after 18 months and includes radical modifications that will result in delay in engineering, procurement and execution. Thus giving a notice of delay

1 Nov 06 the Engineer revise the LV distribution to the apartment with revised panel board distribution vide a new Variation Order

19 Oct 06 the Contractor starts coordination with EDL and requests Engineer to issue further drawings for the transformer and metering room

15 Sep 06 the Engineer issued a Variation Order modifying the metering for panels serving service areas.

23 Jun 06 the Contractor requesting EDL to advise in writing about the use of bus ducts inside the EDL room

15 May 06 the Engineer issues revised architectural drawings based on these comments

6 May 06 the Contractor transforms this information in writing to the Engineer

5 May 06 a joint visit to EDL is carried out where the EDL coordinator comments to the modifications on the shop drawings

28 April 06 the Engineer issue EDL shop drawings to be executed by the Contractor to the satisfaction of EDL attaching list of technical modifications to be carried out

25 Feb 06 the Engineer prepares the revised permit file and transmits it to EDL with a copy to the Contractor and asks the contractor to follow up with the coordination.

17 Jan 06 upon the Employer's request the Contractor summarizes the outcome of the follow up of the permit file and confirms that the official request of the Employer for the new configuration of transformers needs to be filled and signed by the Employer.

12 Jan 06 the Contractor transmits the EDL request to the Employer

7 Jan 06 the EDL informs the Contractor that the original substation as set in the contract was approved in principal but the modified substation (resulting from the value engineering done) necessitates submitting a full set of documentation for EDL study and approval before EDL Engineering department can commence drafting the EDL room.



Subject: Cancellation of PS

Project: A

Section:



The Contractor is requested to put the works in the provisional sum on hold

The Contractor prices the BOQ

A variation order is issued modifying the released design of the provisional sum

A Provisional sum is released for a section of the works and the Contractor is requested to price the corresponding BOQ

CS no: 1



The Engineer rejected several Notifications of Possible Variations (NPVs) as they are part of the original Contract Scope of Work and referred, without limitation, to Clauses 7.1.1.A and 12.1.



07 Apr 06 The Contractor replied claiming that the damages are minor and affected only one branch and cannot in any way be considered as "major" or "dead" and noted that he shall supply the Employer with the method of repairing the tree. The Contractor also noted that the tree was still living after the accident before it was transplanted directly by the Employer. The Contractor also rejected the decision to deduct the advised amount from his due monies.

04 Apr 06 The Engineer notified the Contractor that the Employer will be compensated for a total sum of \$8,504.00 calculated in accordance with item 3.5.D.4 of Section 02231 in the Contract Specifications. The Engineer also noted that the above sum will be deducted from any due monies in the next payment.

14 Mar 06 The Engineer noted that after an accident by the Contractor the Delonix tree is permanently disfigured and categorized the damage as "Major Damage" and noted that it will be dealt with in accordance with item 3.5 of Section 02231 of the Contract Specifications.

CS no: 3



22 Mar 06 The Contractor submitted a new price for supplying and installing the lighting masts of \$US 38,740.00 for each item and amounting to \$US 309,920.00 for all lighting masts (8 No.).

24 Mar 06 The Engineer referred the Contractor, without limitation, to drawing EL117 and Section 16510 "Exterior Luminaires" which show all necessary information. The Engineer added that the instructions, are within the Contract requirements and allow for reasonable time limitations to carry out these optional works. The Engineer instructed the Contractor to abide by Contract Price and rejected the Contractor's revised price.

25 Mar 06 The Contractor noted that the lighting masts are part of the Optional works and not part of the Contract price. The Contractor concluded that the decision to award part of the Optional works should have been decided prior to signature of Contract and not within the Contract duration.

28 Feb 06 The Contractor noted that he is unable to perform these works at the same price inserted in the Optional Works summary due to the absence of any clear specification and since the instruction is issued after more than seven months of Site Possession and Commencement date. The Contractor stated that two alternative materials will be submitted along with relevant costs.

11 Feb 06 The Engineer instructed the Contractor to carry out the optional works for lighting masts as called for in the Contract Documents.





Contractor's statement concerning item 3.01 as being "general statement" and stressed it is the Contractor's obligation to confirm and execute the installation of this system in accordance with Section 13852.

27 May 06 The Contractor noted that Specification Clause 3.01 A states: "Install in accordance with requirements of NEC, NFPA70, NFPA72 and applicable codes, as shown on the drawings, and as recommended by the major equipment manufacturer". This statement is a general statement and does not imply any design responsibilities on the Contractor. It relates to installation methods and details such as height and location of devices, coordination with other disciplines or trades, etc... The Contractor added that Construction drawings / Shop Drawings relate to coordination requirements between trades but not to general design issues, such as coverage of areas, number of emergency lights and the like, the responsibilities for which is vested in the Designer and not the Contractor.

10 May 06 The Engineer rejected the Contractor's statement and reminded him that Fire Alarm System shall be installed and executed according to the NEC, NFPA 70 and NFPA 72 requirements. The accuracy of all such information is the responsibility of the Contractor. The Engineer also reminded the Contractor that he shall check the Contract drawings for discrepancies with the above codes that may occur. If any, the Contractor shall notify the Engineer and shall make all the necessary modifications in order to assure a complete system that complies with all applicable codes listed in the Contract Specifications.

04 May 06 Further to the Engineer's comments on the returned Shop Drawings, the Contractor stated that the onus for the adequacy of the design rests with the Engineer.

29 Dec 05 The Engineer reviewed and issued his comments on the submitted Shop Drawings re Fire Alarm System:

a. Detection coverage shall be provided throughout the buildings.

b. Automatic smoke / heat detectors in all areas with smooth ceilings shall be spaced not more than 9m apart and not more than 4.5m from adjacent walls or beams with depth exceeding 30cm.

c. Dual action manual call points shall be installed not more than 1.5 m away from all exiting doors and 1.1 to 1.5 m above finished floor level at center. One or two manual call points shall be sufficient when having multiple exit doors for the same exit. No one should walk for more than 60 m to reach a manual call point. Clear plastic covers with built-in local alarm shall be provided for all manual call points.





y Project: B

Section: General Requirements





Project: B



29 May 07 The Engineer reiterates his position in the Engineer's Decision

20 Mar 07 The Engineer clarified:

The Contractor prepared his own assessment and evaluation to derive the agreed upon amounts. The Engineer did not mislead the Contractor as proclaimed by him since the Engineer is not responsible for preparing and carrying out the Contractor's duties.

The Contractor's submission was agreed upon by the Employer due to the urgency in finalizing this issue as the Contractor informed the Engineer that the supplier intends to travel abroad to purchase these trees and accordingly the submission was finalized <u>as is.</u> After more than a month of reaching the said agreement, the Contractor requested to revisit this issue noting that the Contractor has verbally informed the Engineer that most of the materials are already delivered. The Engineer rejected and expressed reservation re Contractor's statement "this is a kind of fraudulent action which render our agreement null and void unless you reconsider your position and approve the additional amount of US.\$ 8,907.66 that was unlawfully deducted" noting that the Engineer took the Contractor's own assessment <u>as is with no modifications and confirmed the agreement reached with the Contractor in writing after consulting with the Contractor with respect to the terms of the agreement as noted in the above mentioned letter.</u>

07 Mar 07 The Contractor replied as follows:

"We do not accept your answer and your rejection of our NPV-145.B. Our disagreement is based on how you apply the omissions and in particular BOQ item P: 2/13-W. The Employer requires only 19 Citrus Fortunella trees instead of the specified Pyrus Syraca trees whereby the drawings show 19Nr while the Breakdown of the Lump Sum shows 101 Nr. During negotiation the new plantation prices, we requested that you issue new drawings showing the new distribution of trees after the deleting 82 Pyrus Syraca trees and the locations where the 19 Citrus Fortunella are to be planted.

Since on the Original Drawings only 19 trees are shown to be provided, you cannot omit the total quantity of 101 that exists in the Breakdown of the Lump Sum. Your rejected comments proves that you have mislead us into errors by hiding these facts and ignoring the basic principle of the lump sum Contract that BQs are only used for payment and variations purposes and any quantity to be added or deducted has to be ascertained from contract drawings and not from the BQ and this is a kind of fraudulent action which render our agreement null and void unless you reconsider your position and approve the additional amount of US\$ 8,907.66 that was unlawfully deducted. Should you fail to accept, and insist on rejecting our NPV-145.B, then kindly issue an Engineer's Decision pursuant to Clause 67.1 of our contract conditions."
06 Mar 07 The Engineer rejected the Contractor's claim and referred him to Engineer's letter issued which reflects the agreement reached with respect to this subject.

03 Mar 07 The Contractor submitted a revised notification of possible variation requesting an additional amount of US \$25,640.25 instead of the US\$ 16,733.00 agreed upon.

27 Feb 07 The Engineer accepted the variation and noted that the requested amount for varied works has been agreed to via the Engineer's letter.

27 Feb 07 The Contractor submitted a notification of possible variation requesting an additional amount of US\$ 16,733.00.

25 Jan 07 The Employer and the Contractor agreed to provide, furnish and install the revised scope of Planting Works and the irrigation for the terrace surrounding areas as per the Contract based on the following:

1. The Employer will pay the lump sum amount of \$16,733.00

2. The above works do not entail any time implications on the Contract.



Project: B

Section: Mechanical



The Technical Person completely disregarded what was mentioned in point (a) of our letter and his justification after 11 ½ month from the remarks issued on the approved as noted FIR 332 dated November 18, 2006 and hence his new stand is completely non understandable and rejected.

27 Nov 07 The Engineer reiterated his position and referred the Contractor, without limitation, to Clauses 7.1.1.A, 7.1.2, 7.3, 12.1 of the Conditions of Contract, Question No.11 in addendum No.2, Division I and Section 15820 of the Contract Specifications and NFPA 90A.

The Engineer added "the contractor's interpretation of our letter dated 31/07/2006 is not accurate, the requirement of installing access doors is clearly indicated in the contract document. Moreover the contractor contradicts where the contractor was asked clearly in point 13 to submit the access doors for approval. Figure 4-6 from SMACNA attached to the contractor's letter is typically for smoke dampers and not for combination fire smoke dampers where the access door is necessary for cleaning, testing and maintenance. The Contractor's offered justification to avoid compliance with Contract requirements is not valid. The MFSD shall be installed as called for under the Contract Documents."

15 Nov 07 The Contractor noted "we are rejecting your interpretation of Clauses 7.1.1.A and 12.1. Mistakes, errors or omissions in Tender Documents and Tender Drawings are the Employer's responsibility and not ours. MFSD were installed in accordance with SMACNA provisions and standards in accordance with Specification 15820 Clause 3.02A. Figure 4-6 Of SMACNA clearly state that "if the damper operator is located within the duct, an access door must be provided." All dampers operators are located outside the ducts, hence according to SMACNA no access panel is needed. This is in line with the design intent where no panels are shown or detailed on the design drawings."

The Contractor continued

a. The technical person remarks did not ask for any access door in the submitted field inspection request FIR 332 dated November 10, 2006 as a mock-up for MFSD installation.

b. Hundreds of MFSD were inspected during the course of work with out even the slightest remark of requesting an access door, while remarks are made on NCRs for minor issues of works such as paint patching or support leveling (such an important item should have not been overlooked by the technical person to the end of the job). We are being asked for an access door installation after 11 ½ month from the approved as noted FIR332 dated November 18, 2006.

The Contractor also requested to issue an Engineer's Decision

26 Oct 07 The Engineer reminded the contractor to submit the necessary material submittal / data sheet for duct's access doors that are required to be installed next to MFSD, as noted in NCR24 dated on Aug. 27/2007, on urgent basis.

23 Oct 07 The Contractor replied:

The mechanical fire smoke dampers are with external motors and external auxiliary switches, these not requiring access doors as per SMACNA installation standards and manufacturer installation.

27 Aug 07 The Engineer issued a Non Conformance Report noting that "Material Submittal (Technical Data Sheets) for access doors next to motorized fire & smoke dampers is not available. Refer to Specification 15820 section 2.04."

02 Aug 07 The Engineer noted that access doors should be used whenever deemed necessary.

25 Jul 07 The Contractor replied:

Duct access doors are not required as all fire smoke dampers do not require these doors as all FSD components are on the outside of the duct.

12 Jul 07 The Engineer noted that several major and important items are not submitted yet while works are progressing on site including, without limitation, Duct accessories (Canvas, <u>access doors</u>, etc...).

۱

Year 1 Year 2 Year 3 Sep. Oct. Oct. Oct. Jan. Jan. Apr Jul 24 Nov 06 The Contractor notified the Engineer to issue an "Engineer's Decision" pursuant to Clause 67.1. **25 Jul 06** The Contractor noted that the submittals highlighted in his letter ref. 1322 were returned on 07/07/06 i.e. after 62 & 72 days of the receipt date. 22 Jul 06 The Contractor commented on the 22 remarks made by the Technical Person on the fire fighting Drawings and reserved his right to claim for all cost, time and abortive engineering cause by compliance with the "revised fire fighting design". 12 Jul 06 The Engineer noted that the submittals highlighted in the Contractor's letter were returned on 07/07/06. **10 Jul 06** The Engineer replied that: The Contractor failed to comply with the Technical Person's instructions in not submitting credentials for the Professional Fire Protection Engineer whereas the Contractor's responsibility for designing the "Fire Suppression System" is clearly indicated in the design drawings except for the two floor control valves". 05 Jul 06 The Contractor noted that after several requests to expedite the reply on the submitted composite drawings for the basement level in order not to stop the E/M works, he did not receive the relevant submittals. 04 Jul 06 The Contractor stated that there is no need to submit any credentials for a professional Fire Protection Engineer and reminded that he is not responsible for the design of the fire fighting system and the hydraulic calculations. The Contractor also reserved his right on the capacity and performance of the fire pump since the Technical Person carried out the Hydraulic Calculations and determined the required flow of fire pump is 500gpm at 80m head whereas the Contractor has done the same calculation keeping the same pipe sizes unchanged and need a fire pump that is a lot greater than 500gpm. The Contractor reserved his right to claim at least six months extension of time and related cost. 30 Jun 06 In order to mitigate the Contractor's delays and due to the fact that the Contractor also failed to abide by the Technical Person's request to submit credentials for the Professional Fire Protection Engineer prior to resubmitting any further documentation, the Technical Person carried out the Hydraulic Calculations and determined the required fire pump shall be: Flow = 500 gpm and Head = 80m. Accordingly, the Contractor was instructed to submit a fire pump which will meet the above criteria and in accordance with Section 13920 of the Contract Specifications.

Project: B

Section: Mechanical

CS no: 9

Subject: Fire Fighting System

27 Jun 06 The Contractor noted that he is unable to place the order for the fire fighting system although he has approved material submittals yet he doesn't have approved shop drawings. And he requested the Engineer's advice if he should delay the order until the fire fighting system's design is finalized by the Engineer or should he proceed with order based on his shop drawing.

07 Jun 06 The Contractor replied to the technical comments made by the Technical Person's on MAR and noted that the "issue of professional engineer credential is not our concern, as the professional engineer is requested in the quality assurance to implement the designed fire sprinklers system that is already designed on behalf and paid for by the Employer and this is certainly not part of our contract scope."

22 Apr 06 The Engineer replied that:

1) "Upon receiving the first set of your drawings, we issued our letter instructing you to submit Hydraulic Calculations as you failed to comply with the Contract requirements in providing such calculations.

2) The Fire Fighting submittals referred to in your above referenced letter did not even include the basement part which is the only location that is going to be available for implementation of such related works on site. Moreover, the composite drawings (which included Fire Fighting) for the basement were forwarded for the first time in March 2006. Accordingly, the Contractor shall be solely responsible for related delays.

Accordingly, the Contractor is responsible for initiating delays related to submittals.
4) You shall note your failure to abide by your own schedule of submittals which was never respected and subsequently affected the organization and resource allocation of all involved parties in dealing with your unorganized approach in forwarding related submittals.
5) In spite of the fact that many of your submittals lacked the Contractual period as assigned in the Contract, we highlight the fact that several submittals were actioned in a very short period in order not to delay the works because of your failure to submit according to the Contract.
6) After the expiry period of the 126 days allocated in the Contract by an additional period of 150 days, we are still receiving submittals forwarded for the first time which again reflect your failure to comply with the Contract requirements.

12 Apr 06 The Contractor provided the Engineer with a list of submitted construction drawings that were all returned as "revise and resubmit" status after 57-98 days. The Contractor reiterated his stand that although he is a willing to perform the calculation needed once he receives the requested information he holds the Employer responsible for "unreasonably withholding / delaying issuance of your actions on our submitted construction drawings and for all extra quantities that are required but not indicated on the Tender Drawings."

24 Mar 06 The Engineer noted that:

1) "Our letter issued on 23/12/05 clearly instructed you to submit Hydraulic Calculations. Accordingly, our letter issued on 21/02/06 does in fact infer that it is the second time where instructions to submit Hydraulic Calculations are issued to you where up to date you failed to comply.

The Contractor was referred to the sections of the specs that require him to provide the Hydraulic Calculations

Based on the above, the design requirements as expressed in Section 13925 (without limitation) of the Contract Specifications are part of the Contractor's Scope of Contract. Accordingly, the Contractor is solely held responsible for not complying with the Contract requirements as stated here above and all resultant delays (if any) which may have an impact on the Project completion date will be attributed to the Contractor.

In order to mitigate your noncompliance and as expressed in the Management Meeting held on 20/03/06, the Employer decided to provide the Contractor with attached soft and hard copies of the Hydraulic Calculations for guidance and information. The Employer's action in providing the attached documents shall in no way relieve the Contractor of any of his liabilities and obligations under the Contract. "

16 Mar 06 The Contractor informed the Engineer that the preparation of working drawing for the fire fighting system were stopped due to hydraulic calculation issue since management meeting No.6 held on 13 January 2006 and also have an impact on other mechanical works.

23 Feb 06 The Contractor stated that during management meeting No.6, the fire fighting hydraulic calculations issue was discussed and his interpretation of Specifications Section 13925 item 1.04.C "does not mean that we have to submit hydraulic design calculations but to indicate on the Shop Drawings the hydraulic calculations done by the designer at the stage of the design". The Contractor also added the Engineer's quote "again instructed us to submit the sprinkler flow

calculations within 14 days", inferring that this is the 2th time the Contractor is being instructed, is not true.

The Contractor concluded that since instruction were given to redesign the fire suppression system hydraulically he is ready to perform this design within a period of at least 6 weeks at a design fee of US\$15,000.00.

21 Feb 06 The Engineer noted that even though the Contractor stated that "*Hydraulic Calculations are not part of his scope of work*", he is again instructed to submit the sprinkler flow calculations within 14 days.

17 Feb 06 The Contractor noted that hydraulic calculations are not part of his contract and the act of withholding approval based on non submittal of hydraulic calculations is not in order.

10 Feb 06 The Contractor reserved his rights on the delay caused by the approval of Material submittals that are linked to the hydraulic calculations which are "obviously not part of our contract and should be done by others".

13 Jan 06 It was agreed in the Management Meeting No.6 that the Engineer will further review this issue with the Consultant team and Employer, and conclude by next meeting.

30 Dec 05 The Contractor noted that there are no specific requirements or prerequisites for shop drawings or calculation of any kind for submittals. And once the shop drawings are completed pressure drop calculation will be done to check pump head value as requested by the tender documents. The Contractor added that "only fire pump head is to be checked for actual head based on the approved submitted components mentioned above, and to be used in the system, and on layout changes if any, and as such these hydraulic calculations for the pump can not be done if the components are not approved or commented." The Contractor concluded "Based on approval of system components, pressure drop calculation for pump will be done at a later stage and the pump will not be ordered unless head calculations are approved by the Engineer."

23 Dec 05 The Engineer instructed the Contractor to submit Hydraulic Calculations along with comprehensive and detailed shop drawings prior to issuing comments on fire fighting submittals.



Subject: Towel Dryer



Section: Mechanical







8 Mar 06 The Contractor submits a material submittal for louvers at roof



Project: C

Section: Mechanical

CS no: 2

Subject: Chillers

13 Jan 06 The Contractor sends a letter giving notice for an intention to claim extra payment and varied rate or price for the change in character of chillers from reciprocating type to screw type



06 Jul 06 The engineer sends revised structural shop drawings for building W. The facade of building W is to be strengthened and maintained.



Sep 07 RFI is sent by the Contractor to clarify the type of material to be used for the supports of the required glass balustrades.



The Contractor submitted notice for the stoppage of the Works due to Israeli War on the Republic of Lebanon. Although the Ceasefire was declared on Aug 14, 2006, the consequential disruption to Contractor's work continued including loss of productivity and shifting of external activities to the winter season. Due to the outbreak of war, all of the Foreign Labour determined to terminate their employment.





23 Jun 06 The Engineer instructed the Contractor to replace the yellow grey granite stone type Patmas, noted to be the same via letter ref. a with black granite stone upon which the surface finish nor the thickness were specified.







Specification states that the Civil and Architectural shop drawings submittals shall be replied in a reasonable time, and the Electrical and Mechanical shop drawings shall be replied in a period of 2-3 weeks.



16 Dec 05 The Engineer replies to MAR dated 9/12/05 and deleted totally the aluminum composite panels.



etc... of the newly instructed drains locations and then to proceed with the tiling and drainage layouts accordingly. Again the Contractor requested that the Engineer is to issue a complete developed coordinated design-resolution, particularly with the drainage related requirements (such as but not limited to FD locations, types, reservations, routing, etc...) in order not to delay further the execution of the works.

Section: Architectural





Section: Architectural



22 Jun 08 The Contractor presented a claim stating that the Engineer has yet to respond to the rates and prices submitted. The significance of this additional work is not only the lateness of the instruction and the time necessary to execute the varied work but also the disruptive affect it has on completion of work to other Floors. The Raised Flooring requires the transportation of approximately 150m3 of 'wet trade' materials via the 'back of house' access and thereafter vertically by means of Elevator No.8. Clearly the back of house works could not be completed whilst this additional works were under construction. Likewise all finishing works to 27th and 28th Floors were suspended during the Raised Flooring operations as the dust, dirt and other deleterious material so generated would undoubtedly affect such final trade works.

22 Mar 08 In response to the Engineer's allegations that the HOURDI Blocks were readily available on the local market, the Contractor inquired of all local suppliers and found that indeed blocks of the sizes specified were not available locally. Accordingly the Contractor confirmed that the HOURDI Blocks of the size required by the Engineer were not available. The Contractor proposed alternative sizes that were the subject of a further quotation which had been issued in a separate NPV. In order to expedite matters the Contractor proposed proceeding with these works on Site in anticipation of the Employer's acceptance of this quotation.

23 Mar 08 The Contractor forwarded further revised rates and prices for the raised flooring that was required on the 2^{nd} , 27th and 28th Floor levels. These rate and prices including additional mortar bedding and dry mix sand and cement topping.

17 Mar 08 The Engineer subsequently confirmed the requirement for Raised Flooring to 27th and 28th Floors.

16 Mar 08 Although the Engineer's letter referred to both 27th and 28th Floors, there was some confusion as to the actual requirements of the Employer. So the Contractor requested clarification that the 40cm raised flooring was required for the 28th Floor only.

16 Mar 08 The Engineer instructed the Contractor to proceed immediately and in so doing refuted the 20 days delivery period as he was of the opinion the blocks were readily available.

15 Mar 08 The Contractor responded where he provided said breakdown and in so doing made note that mosaic tiling did not form a part of the submitted price. The letter concluded by the Contractor confirming that there would be a delivery period of 20 days for the HOURDI Blocks, this commencing from the date of instruction to proceed.

7 Mar 08 the Engineer requested quotation for varied works for raising the floor by using Hourdi Blocks, the Contractor was requested to provide a detailed price breakdown for the Employer's consideration.



Subject: Additional Fireplaces Project: D Section: Architecture



The letter concluded by requesting from the Engineer further instruction whether or not to proceed with the coring.

8 Mar 08 A variation order was issued for changing from false fireplaces to real/live fireplaces which confirmed that the ID Nominated Subcontractor would require 20 days additional time from the date of approval of their commercial offer and 'shop / working drawings'.

18 Feb 08 This point was confirmed by the Engineer again.

18 Feb 08 The Contractor was informed that the ID Nominated Subcontractor would execute the construction and installation of fireplaces with the hoods and exhaust pipes being provided and installed by the MEP Nominated Subcontractor.



21 Feb 08 The instruction to proceed with the kitchen required various modifications to the previous design as confirmed by the Contractor's letter that pointed out design deficiencies and requested information as to how to proceed with these varied works.





20 Feb 08 The Engineer provided an A3 size sketch upon which there were manuscriannotations indicating the Employer's requirements.



terms of numbers and quality.

Section: General Requirements







1 Nov 07 Upon the Contractor's request, the Engineer issued 5 A3 drawings showing the requirements of 2 hours fire resistant system in basement 4







Appendix H: Analysis of Dispute Cases

	CASEANALYSIS	Project: A
Case No.: 1	Subject: Lift Overhead	Section: Mechanical-Lifts
Synthesis:		General Observations:
The designer specified the exact type installed at the top of the high-rise b building height has been approved b as such a very critical issue: 1. there is no tolerance for exce 2. as a lift in a high-rise builo reaching the last floor, lift i overhead clearance especially above the room that will requ	e of lift machine that will be building given the fact that the by a presidential decree and is eding building height ling with a high speed s known to required an y if the lift machine is to be kired even more machine	
height In spite of the above and the fact the be possibly executed <u>, it was specified</u> <u>machine, its capacity and its speed</u> the building height limit.	t the lift in this case can not <u>I clearly stating the type of</u> which can not be met given	Desígn error
The Contractor upon signing the Co checked the contract documents inc the Contractor should have checked a critical subject. Moreover, <u>the lift s</u>	ntract is deemed to have luding <u>the permit file</u> . As such the lift overhead issue on such <u>ubcontractor should give</u>	Permits regulations
<u>special attention</u> to this lift that is r high rise building. The Contractor did propose an alterr lift. The Contractor should have rais clearly stating that the original <u>spe</u>	eaching the last floor of the native machine above for this ed the concern/warning <u>cifications can not be</u>	Late intervention by subcontractor
<u>executed</u> . The Engineer upon inspecting the R together made a general statement u Contract should adhere to specs requ	FI that dealt with all the lifts at the cover of the RFI that iirement.	Not achievable requirements
 <u>No comment was made</u> to the machinisuch: the Engineer did not invest and accordingly did not not the Engineer did not specula being proposed. The Engineer did not inquirexplanation By providing such a general commentation of the works falls under the speculation. 	time checking details of RFI time checking details of RFI tice the problem ate why an alternative was re the Contractor for an ent as a reply the Engineer did de and vigilance as the ne Contractor's liability.	∓aílure to notice technical problem

•

Contractor proceeds with submitting	g shop drawings 1 year from	
the RFI reply. At this point the Cont	cractor does not raíse concern	
regarding the overhead issue for wh	ich he had previously proposed	
an alternative solution. <u>This 1 year</u>	lag between sending RFI and	
submitting shop drawing is not jus	tified. Coordination on this	
subject was put on hold for a long p	eriod which shows bad internal	Lack of proper
management on the Contractor's side. The shop drawing has		management/monitoring by
mistakes in calculating heights and as such indicated wrong		the Engineer
levels.		
The Engineer approves the same not noticing overhead problem or		Human error/negligence
mistake in calculating levels.	C I	
Contractor resubmits roof drawings	due to modification in	
structure.		
Contractor sends another RFI stating that the building height		
will change.		
From the time the RFI highlighting	the problem is sent 5 months	
are spent till the solution is reached	among the parties. During	
this period each party looks for cont	ractual readings to put the	
blame on other party.	_	
Instead of having a workshop to resu	olve the matter as it was	
finally done different unachievable	/impractical solutions are	
proposed.	· · · · · · · · · · · · · · · · · · ·	Lack of cooperation
The Employer representative accepts	the alternative solution where	Lack of experience
it is the technical Engineer's duty t	o find and advise on such	
decisions. The <u>Contractor submits</u> a	n NPV with the details of the	
additional cost incurred. The Engir	leer rejects the same.	
		Valuation of Variation
	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	
Areas of risk (as categorized by	Zack, 1996) identified in thi	s case analysis:
Impracticality/impossibility	The contract design wan not ac	hievable
Subcontractor or supplier failure	The Subcontract failed to ident	ífy the problem
Delay and Disruption	The lift problem delayed the pou	iring of concrete at roof
untimely responses	The subject problem dragged for	r a long períod
Coordination	The drawings were not well coor	dinated between trades
Permits and Licenses	The building permit did not all	ow for additional building
	height	
Changes	The Contractor considered the n	ew solution reached as a
	variation	
Defective Contract Documents	The original design did not sat	isfy permit regulations
Behavioral Observations:		
- There was a tendency to e	vade responsibility by blamin	ig the other
- Both parties were nealiaer	nt in their behavior	-

	CASEANALYSIS	Projec
Case No.: 2	Subject: Façade False Ceiling	Section: Finishes
Synthesis:		General Observations:
The Brazilian cherry wood was	s used as the basis in the cost	
savinas exercise so it was clear	dy the option that the	
Contractor priced and still the	addendum as drafted by the	
Engineer was not clear in this	respect specifying 3 different	Contract Document unclea
options when it was the cheaper	st one ie "the Brazilian cherry	
wood" that was priced.	0	
The Contractor argued that eve	en where the Contract specifies 3	
options, the Contractor would b	be meeting the requirements by	
providing any one of the 3 opt	ions as long as the material	
submitted is to the Engineer's	approval.	
The Engineer's reading of the	contract was biased toward's	
the Employer's choice of wood a	on the premises that the	
Engineer could choose any of t	the three options and approve it	
accordingly.		
The Engineer rejects the NPV o	n the basis that the Contractor	
is not allowed to any variation	i in this respect and even .	
is not allowed to any variation reiterates his position.	in this respect and even .	
is not allowed to any variation reiterates his position. From the time the NPV is raise	n in this respect and even . ed, 5 months are spent until the	
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resolu-	n in this respect and even . ed, 5 months are spent until the ved through direct contact	
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order	n in this respect and even . ed, 5 months are spent until the ved through direct contact <u>e Employer who accepts</u> to	Influence by the employer
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order	t in this respect and even . ed, 5 months are spent until the ved through direct contact <u>e Employer who accepts</u> to	Influence by the employer
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized	t in this respect and even ad, 5 months are spent until the ved through direct contact <u>Employer who accepts to</u> d by Zack, 1996) identified in	Influence by the employer n this case analysis:
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation	in this respect and even d, 5 months are spent until the ved through direct contact <u>Employer who accepts</u> to d by Zack, 1996) identified in Cost escalation increased the pr	n fluence by the employer n this case analysis: roblem in this case
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol- between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents	in this respect and even d, 5 months are spent until the ved through direct contact <u>Employer who accepts</u> to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should	Influence by the employer n this case analysis: Toblem in this case d have specified the type of
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol- between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents	in this respect and even id, 5 months are spent until the ved through direct contact <u>Employer who accepts</u> to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th	Influence by the employer n this case analysis: oblem in this case d have specified the type of e value engineering
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol- between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of	in this respect and even d, 5 months are spent until the ved through direct contact <u>Employer who accepts to</u> d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 2 options	Influence by the employer n this case analysis: oblem in this case d have specified the type of e value engineering contract drafting the
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes	in this respect and even d, 5 months are spent until the ved through direct contact <u>Employer who accepts to</u> d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contract considered the F	Influence by the employer n this case analysis: Toblem in this case d have specified the type of e value engineering contract drafting the set was disputed
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol- between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes	in this respect and even d, 5 months are spent until the ved through direct contact Employer who accepts to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contractor considered the E variation	Influence by the employer n this case analysis: Toblem in this case d have specified the type of e value engineering contract drafting the set was disputed Engineer's request as a
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes Behavioral Observations:	in this respect and even id, 5 months are spent until the ved through direct contact <u>Employer who accepts</u> to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contractor considered the E variation	Influence by the employer n this case analysis: Toblem in this case d have specified the type of e value engineering contract drafting the set was disputed Engineer's request as a
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes Behavioral Observations: - The Engineer reiterat	in this respect and even id, 5 months are spent until the ved through direct contact <u>Employer who accepts</u> to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contractor considered the E variation	Influence by the employer n this case analysis: Toblem in this case d have specified the type of e value engineering contract drafting the set was disputed Engineer's request as a natter is not clear in the
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol- between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes Behavioral Observations: - The Engineer reiterat Contract.	in this respect and even d, 5 months are spent until the ved through direct contact Employer who accepts to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contractor considered the E variation tes his position although this n	Influence by the employer n this case analysis: Toblem in this case d have specified the type of e value engineering contract drafting the set was disputed Engineer's request as a natter is not clear in the
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol- between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes Behavioral Observations: - The Engineer reiterat Contract. - The Employer accepts	in this respect and even d, 5 months are spent until the ved through direct contact Employer who accepts to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contractor considered the E variation tes his position although this n s what the Engineer has rejecte	Influence by the employer n this case analysis: Toblem in this case d have specified the type of e value engineering contract drafting the set was disputed Engineer's request as a natter is not clear in the d.
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resol between the Contractor and <u>the</u> consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes Behavioral Observations: - The Engineer reiterat Contract. - The Employer accepts	in this respect and even d, 5 months are spent until the ved through direct contact Employer who accepts to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contractor considered the E variation tes his position although this n s what the Engineer has rejecte	Influence by the employer n this case analysis: oblem in this case d have specified the type of e value engineering contract drafting the set was disputed Engineer's request as a natter is not clear in the d.
is not allowed to any variation reiterates his position. From the time the NPV is raise cost implication aspect is resolibetween the Contractor and the consider it as a variation order Areas of risk (as categorized Cost Escalation Defective contract documents Interpretation of requirements Changes Behavioral Observations: - The Engineer reiterat Contract. - The Employer accepts	in this respect and even ad, 5 months are spent until the ved through direct contact Employer who accepts to d by Zack, 1996) identified in Cost escalation increased the pr The Contract documents should wood that was priced during th Notwithstanding the defective interpretations of the 3 options The Contractor considered the E variation tes his position although this n s what the Engineer has rejecte	Influence by the employer n this case analysis: Toblem in this case d have specified the type of e value engineering contract drafting the set was disputed Engineer's request as a natter is not clear in the d.

)

	CASE ANALYSIS	Project:
Case No.: 3	Subject: Procurement of new materíal	Section: General Reqmts
Synthesis:		General Observations:
The agreement reached for the reversations that are to be assesses shows that there was a positive parties. The Contractor accepted to have that it will not result in incurr The Engineer considered that to diligence in proceeding with bu these variation. Late procureme extra cost for which the Engine Contractor delays buying ther <u>didn't explicitly state the Cont</u> allowed at the time original ag assessing new market rates all variation was calculated at the <u>the issuance of the VO</u> . The Con items are difficult to <u>procure in</u> valid reason and should be exa	mechanism of issuing late ed on market rates clearly coordination attitude among the late variations on the basis ing loss. he Contractor should do due kying the material specified in the contractor should do due kying the material specified in the <u>market in incurring</u> er will have no control if the n. However, the Engineer ractor's procurement time reement was reached but in lowed for 1 month. As such the <u>market rate one month after</u> ntractor's claim that some <u>a month</u> time might be a mined by the Engineer.	Príce escalatíon Contract Document unclear Valídíty/Assessment of Variation Submíttal Schedule
Areas of rísk (as categorízed	d by Zack, 1996) ídentífied í	n thís case analysís:
Cost escalation	Again due to the continuous co in exchange Euro rate for most Europe led to increasing the eff	st escalation and the change of the ítems procured from ect.
Changes Defective Contract Documents	There was a large number of va requesting new design materia separately. The Contract Documents didn't	riations because the ID was I for each apartment state the period.
Changes Defective Contract Documents Behavioral Observations:	There was a large number of va requesting new design materia separately. The Contract Documents didn't	riations because the ID was I for each apartment state the period.

	CASEANALYSIS	Project: A
Case No.: 4	Subject: Façade glass	Section: Facade
Synthesis:		General Observations:
Façade details were raised in a <u>workshop meeting held after</u> <u>14months</u> in a 30months contract, this issue given its criticality and that it is not clearly specified in terms of frit design should have been initiated earlier on. Also, both parties did not account for the time period required to receive each sample. The Contractor had clearly underestimated the time needed to provide samples in the programme. This item was described as design build in the BOQ. There was <u>misunderstanding</u> regarding the description as the Contractor considered the internal skin as a design build system to meet specified load whereas the Contractor considered the fritting on the external skin as an aesthetical item to be chosen by the Engineer and not a design build system. It was the Engineer's understanding that both the internal skin and external skin are design-build. The Engineer did not <u>request full range of samples</u> at the first submission to save time. Moreover the notification of possible		Late intervention of subcontractor Clear risk allocation and highlighting Lack of experience Trial and error attempts
delay was <u>rejected although this issue had become critical to</u> the project and was delaying the works and an extension of time was later granted for this delay.		Assessment of delay effect
Areas of risk (as categorized	l by Zack, 1996) ídentífied í	n thís case analysís:
Delay and Disruption Untimely responses Interpretation of requirements	The process of providing the 4 samples took time The Engineer was late in replying to the submittals The responsibility of designing the frit glass pattern was not allocated clearly	
Behavioral Observations:	L	
<ul> <li>The Engineer insister responsibility.</li> <li>Both parties were late</li> </ul>	d on the fact that the delay wi in attending to this subject.	as the Contractor's
	CASE ANALYSIS	Project: A
---------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------------------------------
Case No.: 5	Subject: War Effect	Section: General Regmts.
Synthesis:		General Observations:
The Contractor the timely notification to the Employer's risk event occurrence and submitted the necessary interim substantiation to the ongoing effect thereafter. The Engineer <u>assessed the EOT of time that occurred in July OG and, which</u> had ongoing effects in labor loss till end of 2006, in <u>Nov. 07</u> . <u>The Contractor submitted the cost implications 1 year late</u> with the justification that subcontractor cost implications were delayed. <u>The Engineer made his assessment of</u> the cost implications 2 months after the revised cost implication submittal. However, <u>the Employer took 8 months</u> in reviewing the same before he would grant his approval with comments. At this point the Contractor requested an Engineer's Decision.		Assessment of Delay Engineer Understaffed Late submission of cost implication Assessment of war claim Employer's Interference
Areas of risk (as categorized	t by Zack, 1996) identified in	n this case analysis:
force Mujeur	contract	was a furce majeur ander the
Labor forces	There was a problem in maintainin Delays and disruptions occurred f	ng the required levels of labor rom war and this led to delay
Delays and disruptions	díspute.	
Suspension of work	Suspension of works occurred for	40 aays auring the war. Ining the war
Behavioral Obsenvations.	Site Safety Site Safety was jeoparaized auring the war	
- Ela DÍLA PER WAS LATE ÍLA ASSESSIA PLAT		
- Contractor was late in	- Contractor was late in submitting cost intoligations	
<ul> <li>Employer delayed the assessment of this variation as his approval was contractually required in assessment of claims.</li> </ul>		as hís approval was

	CASEANALYSIS	Project: A
Case No.: 6	Subject: Add. Shop dwgs. for	Section: Preambles
	variations	
Synthesis:		General Observations:
An agreement regarding the milestone for releasing revised design for every apartment was introduced after commencement of the works and after the project programme was submitted and consented. The Contractor <u>claimed for this additional cost 10 months</u> after the modified design was released. As such proper notice was not given. And <u>no consideration was made</u> for the additional effort/cost of preparing shop drawings for the revised design as no addition to the Contract rate was made following this agreement. One possible explanation could be that where the Contractor was examining his running cost, he realized the additional cost resulting from the revised design and claimed for it accordingly. Although there was no contractual agreement regarding the cost, the Employer could still examine the validity of the Contractor's request in good will gesture to maintain a positive relationship.		Validity /Assessment of Variation Contract Document unclear
Areas of rísk (as categorízed	l by Zack, 1996) ídentífied í	n thís case analysís:
Quantity variations Defective contract documents Changes	y variations contract ts The Contract didn't clearly mention the cost impact of having the additional shop drawings There was a clear deviation from the standard design	
Behavioral Observations:		
<ul> <li>The Contractor was avoiding monetary losses</li> <li>The Engineer in this case did not certify any variation as it was not set in the contract.</li> <li>As mentioned above the Employer could reconsider the validity of the request to maintain positive attitude with the Contractor.</li> </ul>		

	CASEANALYSIS	Project: A
Case No.: 7	Subject: Façade Lighting	Section: Electrical
Synthesis:		General Observations:
The Contractor submits the façad	e lighting that needs a lead	Late assignment of
time of 7 months in procurement and 2 months in installation		subcontractor
in Oct 07 ie <u>29 months after commencement</u> of the works and 1		Lack of Proper
month prior to the original contract completion date. The		Management/Monitoring by
Engineer takes 22 days after the	field inspection was requested	Engineer
to state that the mockup should be	submitted with the approved	ALate approval of submittals
frit glass and that more lighting	fixtures are needed. This could	by Engineer
have been replied in 1 day. The Er	gineer requested that the	
horizontal lighting match the ver	tical lighting which the	
Contractor considered as a variati	on as this was not specified in	
the Contract.		
The Contractor requested the appr	oval to the alternative solution	
for two reasons: to put the order an	nd avoid increase in rates, and	
to facilitate/expedite getting more	samples. However the	
Engineer ties his approval to the c	ommercial offer requested by	Employer interference
the Employer for cost savings. Al	so, the Engineer requested the	employer millinger
horizontal lighting be made home	ogenous with the vertical	
lighting and the crown should be		
lighting. The Engineer gives his i	approved as noted status to the	
alternative 2 months after it was submitted. However, this AAN		
status still doesn't allow him to place the order <u>Further</u>		
specification data is required by the Contractor but the Engineer		Lack of experience
insists that it is the Contractor's	responsibility to get the	
required homogeneous effect in co	oordination with the specialist.	
The specialist explains that this ca	an not be achieved technically.	Not addie value reactively entr
		Not achievable requirements
Areas of risk (as categorized b	y Zuck, 1996) identified in t	nis cuse unalysis:
Subcontractor, supplier failure	The façade subcontractor failed to prov	lide requirements
belays and disruptions	complexity.	orneccue, ence approval and
untimely responses	The Engineer takes too much time to r	eply.
interpretation of requirements	Whether it is the Engineer's obligation	to specify or the Contractor's
	responsibility to design the system.	
Change	The Contractor considered the request	for homogeneous lighting as a
Coordination	There was poor coordination	
Defective Contract Documents	The Specifications were not clear in sp	ecifying lighting color
Behavioral Observations:		
- The technical submittal approved is being tied to a commercial offer.		
- The Engineer is requesting modifications without being able to clearly detail		
technical specifications of the variation.		
- The Contractor starts with his submittals late.		

- The Contractor starts with his submittals late.

	CASEANALYSIS	Project: A
Case No.: 8	Subject: Frit Glass	Section: Aluminum Facade
Synthesis:		General Observations:
This matter drags from end	of Nov. 07 to end of March	
08. The Engineer insists tha	t the requested product is	
available in the market and	lísts supplíer contacts that	
would provide the same. The	Contractor through the same	
contacts proves that the prod	uct is not yet in the market	
with the requested 10 yrs gu	arantee. If the Contractor's	
statement is true, then the E	ngineer has meant to	
specify a system with a 10 g	eover his not specificing	Lack of experience
clearly that it is on face 1. th	ne Contractor interpretation	. ,
that the specification calls fo	or face 2 that can meet the	· Contract Document unclear
10 yrs warranty is valid. T	he Contractor failed to notice	
the comments made by the E	ngineer in the first	Human error/negligence
submittal that requested frit	ting on Face 1. This was	Contractor tace the submittents
further confirmed after <u>sever</u>	al mockups were inspected.	<ul> <li>Tríal and error attempts</li> </ul>
The valuation of variation is critical since it is difficult		
to determine whether the Contractor is entitled to such a		
Variation and <u>obtaining the market of such variation is</u>		Valuation of Variation
also altricult because limited specialists price such item.		
Areas of rísk (as categorízed	by Zack, 1996) ídentífied ív	r thís case analysís:
Delays and disruptions	Delay due to unclear requirement	nts
Untimely responses Defective contract documents	Not clear specifications	
Interpretation of	Interpretation of the fritting set	: In the specifications
requirements		
Behavioral Observations:		
- Each party was putting the blame on the other party to avoid monetary losses.		
- The Contractor out of negligence lost 8 months in initiating procurement of		
requested fritting.		
- Engineer firm although contract is grey.		
- There is lack of management/monitoring of Engineer to the delay in		eer to the delay in
initiating procuremen	rt.	

	CASE ANALYSIS	Project: A
Case No.: 9	Subject: Employee Turnover	Section: General Regmts
Synthesis:		General Observations:
The Contractor is attributing t	he raise in salaries to the	
political situation in the count	ry that is encouraging	Accessionent of war claim
Engineers to leave. The Engine	er from his point considered	Contract Document unclear
that the move of Engineers tow	ards neighboring Arab	
countries is due to the high den	hand in the construction field.	
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied í	n thís case analysís:
Force Majeur Labor forces Delays and disruptions Productivity	The war is the influencing factor identified in this claim. The ongoing political situation in the country affected the availability of human resources. The loss of human resources in the country and the employee turnover led to disruption of the work. Due to Employer turnover, learning causes loss of productivity.	
Benavioral Observations:		
The Engineer was firm in considering the Employee turnover as not related to the		
internal political situation.		
The Contractor claims to compensate his losses from market inflation by attributing		
it to the war (Employer's rísh	و)	

	CASEANALYSIS	Project: A
Case No.: 10	Subject: Safety Film	Section: Aluminum Façade
Synthesis:		General Observations:
The Engineer issued a variation order canceling the roller shutters; however, the variation order requested that the glass in the shop fronts become antivandalism and the Contractor was requested to provide different options that could satisfy this requirement. The Engineer referred to the options such <u>as polycarbonate clad or PVB</u> <u>options</u> along with the safety film system proposed by the Contractor. The Engineer clearly <u>was not technically</u> <u>knowledgeable</u> in the area and was relying on the Contractor's investigation. The Contractor preserved his contractual right and sent an NPD, although this issue was not delaying the whole project completion. The variation was raised and debated beyond <u>the original completion period</u> of the project and it <u>spanned for 4</u> <u>months</u> .		Not achievable requirements Trial and error attempts Lack of experience Late issuance of missing design/variation Slow attendance to responsibilities
Avence of victo (ac acteo ovized	huzach 1000) identified in	+ this pase an alusis
Areas of risk (us cutegorized	by Zuck, 1996) wentified w	this case analysis:
Untimely responses Changes Interpretation of requirement	The response rate was slow The Change requested was to remove the roller shutters The Contractor's obligation to provide options for systems instructed by the Engineer were debated.	
Behavioral Observations:	· · · · · · · · · · · · · · · · · · ·	
The Engineer was not knowledgeable enough. He was requesting systems that were not achievable. Both were evading responsibility by blaming the other party.		

	CASE ANALYSIS	Project: A
Case No.: 11	Subject: False Ceiling	Section: Finishes
Synthesis:		General Observations:
The issue is raised in December 2006, however the Contractor reissued the drawings in May 2007 (ie <u>after</u> <u>5 months</u> ). The gypsum detail was submitted and changed 4 times. 40 days after the second submittal reply the contractor raises the issue of the 10cm min clear height. After the 3rd submittal the Engineer requests to add C-channels. 2 months in discussion is spent until the Engineer is <u>convinced of the need</u> for the 10mm plaster. The Engineer had requested a negative variation in previously removing this 10mm plaster. The whole process takes <u>10 months of discussions which</u>		Contractor late/missing submittals Lack of experience Assessment of Delay
<u>delays the works bec</u> ause fal predecessor to many interna	se ceiling was a critical L finishes	
	hu zach cooc) (dautí (ind í	
Areus of risk (us cucegorized	The process spanned for 10 months	until a final decision was
Delays and disruptionsThe process spanned for 10 months reachedUntimely responsesThe responses by both parties are lat The responses by both parties are lat The Engineer modifies the detail ser Work QualityWork QualityThe aesthetics of the work quality is		ite everal times is raised and discussed
Behavioral Observations:		
<ul> <li>The Engineer was having a series of trial and error attempts to reach the required results</li> <li>The Contractor was failing to notify of missing details in a timely manner</li> <li>The process dragged for long due to late submittals and responses to submittals</li> </ul>		

	CASEANALYSIS	Project: A
Case No.: 12	Subject: Lighting Fixtures	Section: Electrical
Synthesis:		General Observations:
The Engineer replies in $\neq 2$ days to the first submittal to request samples. Upon the request of 2 samples as per <u>specification requirements</u> the Contractor claims for the resulting additional cost as the Contractor considers that there is no obligation to submit 2 samples in all submittals. The Engineer replies after one month to the second submittal requesting a sample of the specified model. It takes the Contractor 5 months to provide the specified model although the specification requires the Contractor to provide samples of both the specified model and the alternative. The Engineer rejects the third submittal after one month stating that the alternative is not equivalent to the specified model. The Contractor objects to the same stating that if the rejection is due to the light beam angle this can be modified in the alternative. The Engineer <u>sends an NPD accordingly</u> . The Engineer reiterates his position and rejects the NPD on the basis that it's the Contractor's responsibility to provide an equivalent alternative. This issue drags for 13 months.		Contractor failure to satisfy requirements Assessment of delay
Areas of risk (as categorized	by Z.ack. 1996) ídentífied í	n this case analysis:
Delays and dísruptions Untimely responses Interpretation of requirements	The issue drags for 13 months Both parties are late in providing response There is a disagreement regarding that classifies as an alternative	
Behavíoral Observations:		
<ul> <li>The Engineer could have raised his comments to submittals in a shorter period</li> <li>The issue dragged for long due to late response attendance by both parties.</li> <li>Engineer is firm although the contract is grey regarding the equivalent alternative.</li> </ul>		

	CASEANALYSIS	Project: A
Case No.: 13	Subject: Increase Labor Rate	Section: Regrmts
Synthesis:		General Observations:
The Contractor submits the <u>í</u> trades claímina that it resul	ncrease in labor rate for all	Clear rísk allocatíon and hiahliahtina
Engineer rejects the same ar	guing that there has been a	
global fluctuation of rates in	the whole region and it is	Príce escalation
not due to the war in Leouno	n. Only increase in Labor	Assessment of war claim
Areas of rísk (as categorízea	by Zack, 1996) ídentífied í	n thís case analysís:
Labor forces	There was a problem of scarcity of	labor forces after the war
Cost Escalation	Rates had increased due to the War growth	r sítuation and regional economic
Productivity Force Majeur	The scarcity caused loss of productivity The war effects fall under Force Majeur	
Behavioral Observations:		
<ul> <li>The Contractor was claiming for labor increase on all trades under the pretense that it was an ongoing Employer's risk event to avoid monetary losses.</li> </ul>		

	CASEANALYSIS	Project: A
Case No.: 14	Subject: Façade Alum. Color	Section: Façade
Synthesis:		General Observations:
It takes the Engineer 40 day	js to request that the	
Contractor to relocate the sample. <u>The Contractor sent a</u> <u>notification of possible delay</u> although the Engineer stated that there was a parallel delay in the tests being conducted. In 17 Sep 07 the Contractor submits the material submittals, (after 21 days) and requests the Contractor to indicate the material reference numbers on		Assessment of delay
Contractor to indicate the material reference humbers on the submittals. The Contractor provides the same the second day. The Engineer replies that the panel labeled 3 is approved. The Engineer could have inspected the material, requested a clarification on the reference during the inspection and approved it. This would have saved one month in the process. Moreover, <u>it took the Engineer more than 4 months</u> to approve a material that was submitted from the start of the process. The Engineer requested to inspect different colors before the final decision was made.		Late approval of submittals by Engineer
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied í	n this case analysis:
Delays and disruptions Untimely responses Interpretation of requirements	The late approval of the inspection of the material delayed the work The Engineer was late in giving response The requirements regarding the Contractor's duties in submittal and details of the submittal requested is disputed.	
Behavíoral Observatíons:		
<ul> <li>The Engineer was having a series of trial and error to reach the aesthetic result</li> <li>The Engineer could have been more proactive and reduced the reply period.</li> </ul>		

	CASE ANALYSIS	Project: A
Case No.: 15	Subject: Steel Structure Design	Section: steel structure
Synthesis:		General Observations:
The material submittal is sent in Nov 06 the Engineer replies to the same in Feb 07 as ANR. The Contractor resubmits in 25 days. The Contractor replies to the second submittal in April 07 (2 months later) again as		
ANR. 2 months later the Contractor submits the requested details of the option for hot dip galvanizing. However, 2 weeks after the Contractor submits the alternative paint option that could reduce the delivery period. The Engineer accepts the same on condition that		-Late approval of submittals by Engíneer
the <u>cost savings should</u> inure to the Employer's benefit and based on the <u>time savings</u> and actual progress of the works the NPD is rejected.		Assessment of delay Validity/assessment of Variation
		•
Areas of risk (as categorized	l by Zack, 1996) identified in	n this case analysis:
Delays and disruptions Delay occurred due to late submitt by the introduced alternative Both sides were late in submitting An alternative was proposed by th Means and methods of construction		al and approval but was reduced and approving respectively e Contractor
Behavioral Observations:	<u>1</u>	
- The Contractor should - The Engineer could h	d have submítted alternative e ave been more proactive in pro	arlíer. víding tímely replíes

	CASEANALYSIS	Project: A
Case No.: 16	Subject: Trench Heaters	Section: Mechanical/Finishes
Synthesis:	L	General Ohcen/ations.
The shop drawings for tiling w	as submitted by the Contractor	General Costi vacionas.
and approved by the Engineer as per the Contract Design for		
several areas. Both parties failed to notice the problem of		desígn error
having the trench heater jamm	ing with the gluminum	
facade. The Engineer notices th	e problem on shop drawings 9	
months after first shop drawin	a approval and requests the	
Contractor to move the trench h	eater away from the facade	
requesting the Contractor to con	afirm and coordinate the same	
with the mechanical sustem de	tails At this point the problem	
is raised but there is no decision	reached reagrating the final	
acceptable distance since	tractica regurating the linat	
1 The Englineer records the mo	tter as a porting time issue	
1. The chighteer regards the had	Le autrin considicte é a part	Contractor poor coordination
of the Constraint and the const	-e curtain specialisis i.e. parc	between trades
of the contractor's scope	the loss of Costinue of the	
2. The Contractor considers this	s to be a <u>monification of the</u>	
design issue to be determined b	y the Engineer.	<ul> <li>Valuation of Variation</li> </ul>
The Contractor coordinates with	the specialists who	
recommend a range to the mini	mum clear aistance required	
but fail to provide technical sub	stantiation. The Engineer is	
the one who takes the final decision based on the data		
available 11 months after the first submittal of tiling		
drawings made. However, the <u>allocation of responsibility</u> in		Contract Document unclear
the delay incurred is disputed This matter would have had a		
bigger impact had they proceeded with the original design		
and breakage of the façade glass occurred. Although it was		
technically resolved in 11 months the dispute regarding		
allocation of responsibility con	tinues for another 11 months.	
Areas of rísk (as categorízed	by Zack, 1996) ídentífied í	n thís case analysís:
Delays and disruptions	The delay occurs because execution	of work as per approved shop
	drawing is stopped	
Untimely responses The issue dragged for 11 months		
Coordination There is a coordination problem bety		WEEN TRUES
Defective contract accuments I he accument is not clear there is a cluster pretation of requirements Each party is all posting responsibility		lifty to the other party herauce
	this issue is not clearly specified.	
Changes	The Contractor considers the Er	igineer's request to relocate the
trench heater as a variation		
Behavioral Observations:		
- Both parties failed to notice the problem and each was allocating it to the other		
party to avoid responsibility for the delay incurred.		
- Also, each party wanted to evade responsibility for the executed system.		
		-

	CASEANALYSIS	Project: A
Case No.: 17	Subject: Submittal schedule	Section: General Regmts.
Synthesis:		General Observations:
The Contractor submitted the s	ubmíttal schedule as per the	
Contract requirements. Howeve	r, the process <u>of submitting</u>	
and commenting the schedule	dragged for more than 18	Suppoittal Schedule
months and a submittal sched	ile was never approved.	Submitter Sonthand
Accordingly an alternative for	marking the critical ones in	
the weekly updates (and not al	l the submíttals that have	
exceeded the review period) was	proposed to go around the	
problem. However, this did not i	resolve problem as the	
Contractor díd not waive his rig	ght to claim for extension of	
time where the <u>delay in review c</u>	of submittals is delaying the	Late approval of submittals by
progress of works. Also, the End	gineer did not have an	Engineer
approved submittal to allocate s	taff for review accordingly or	5
to monitor the proper progress o	f submíttals. Both parties were	
late in submitting and reviewi	ng these schedules. Also, the	
parties did not agree to the nun	iber of submittals allowed per	
week. This could have been reso	lved if the weekly figure had it	Contract Document unclear
been included in the Contract.	Also, it is evident from the	
disagreement on the number of	submittals allowed per week	
that the Engineer could not car	ry on the big number of	
submittals which could be due	to the fact that he/she is	
understaffed or due to the fact	that the Contractor did not	
proceed with submitting in a le	veled timely manner as he is	
expected to. There was also a pro	blem in the high number of	
resubmittals as well which the	contractor attributed to the	
Engineer not including all con	iments on the first submittal.	
The Engineer on the other hand	claimed that the contractor is	Contractor late/missing
not coordinating the submittal	s well and including all	submittals
necessary actails which is lead	ing to un increased number of	
resubmittais.	hu Zach 1996) identified in	+ this case an alusis.
Areas of risk (as cacegorized		and and a first and a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a first a
Untimely responses	Both purches were late in submitter	his submittels annous different
Coordination	trades to avoid delay	this submittents undrig utgetene
befective contract documents The Contract document should set some levels in this respect		some levels in this respect
Behavioral Observations:		
- Both parties could not reach an agreement regarding the submittal schedule.		
- Each party was blaming the other party for not having an approved schedule		an approved schedule
- Lack of proper management/monitoring where the Engineer failed to impose a		
reasonable weekly no. of submittals.		

	CASE ANALYSIS	Project: A	
Case No.: 18	Subject: Marble Works	Section: Finishes	
Synthesis:		General Observations:	
The Engineer during the field inspection request decides to			
change the corner detail which was approved in a shop		validity/assessment of	
drawing, 6 months ago request	ing the Engineer to resubmit	Variation	
the related shop drawing. The C	ontractor resubmitted the	Late issue of missing design	
general detail in 12 days. The	Engineer replies to the same		
AAN in 12 days as well. The C	ontractor sends an NPV due to		
the requested change. The Cont	ractor submits the first FIR 1		
month after the shop drawing i	s approved. <u>4 FIRS are</u>		
submitted for this item. At the	first submittal a review of the	Late approval of submittals by	
shop drawings is required. The	Engineer replies in the second	Engineer	
and third FIR as follows:			
grout not installed, tile repair	not finished, touch up		
required is not executed			
- grout is missing, eage change	d again to be allowed		
bathtub to be treated. Walls and	a thoor to be cleaned.		
The comments show that the Co	eeric remarks raised in the		
fust EID However the English	ter s remarks rused in the		
edge chamfering and interface	to be treated which were not		
made at the first EIR and not a	then ded to The Contractor		
claimed for an extension of tim	the for the delay stating that the		
modifications took the form of	tríal and error attempts	-Assessment of delay	
during the FIR 3 months span between the submittal of the			
first FIR and the approval of the fourth one.			
Areas of risk (as categorized	l by Zack, 1996) identified in	n thís case analysís:	
Untimely responses The responses from both sides wer		late	
Changes Engineer changes corner detail			
Coordination	Coordination is needed with sanite	ary fixture	
Interpretation of requirements Specs requirements for grout on s		ides is argued	
Delay and Disruption	rks in the wet areas		
Behavioral Observations:			
- The Engineer was having trial and errors to reach the requested aesthetical			
model.			
- The Engineer was adding requirements through the FIR comments			
- The Contractor was not satisfying the comments raised in the FIRs in			
dílígence			
- Each was evading responsibility by blaming the other party			

	CASE ANALYSIS	Project: A
Case No.: 19	Subject: EDL	Section: Electrical
	<u>_</u>	
Synthesis:		General Observations:
The original design was approved in principal but missing		
due to the value engineering th	at only stated ideas without	aesign error
details. The EDL (Electricity of Lebanon) requested that a		
revised file be submitted with an official reavest from the		
Employer for the bigger substat	tion. The Engineer provides the	
revised permit file however the p	rocess takes several months	Dermit regulations
since the EDL requests further	details during the follow up	
visits which are to be provided b	y the Engineer in a specific	Luck of proper
format. Also the value engineer	ing requests larger	Engineer
transformers (nonstandard on	es) which necessitate an	
undertaking by the Employer	for the spare transformer. EDL	
does not provide any confirmat	ions or replies in writing; The	
Contractor representative who is	responsible for coordination	
with EDL to execute the substat	ion room is late in following	lack of experience
up on the EDL file between Mar	of and Dec of beyond which	
further details are still requeste	d by EDL. And the substation	contractor poor coordinacton
room becomes a critical activiti	delaying the project	Decween crudes
completion date as it is a predec	essor to the testing and	
commissioning of electrical an	d mechanical systems. The	
Engineer blames the Contractor	for not performing due	
diligence in following up the file with the EDL. The Contractor		
states that this delay is due to the value engineering that was		
initiated by the Engineer after contract award and which		
necessitated a revised permit file and several additional		
official paperwork that were not required by the original		
design. The Contractor requests	s an extension of time for the	Accession and of delay
same.		Assessment of heldy
Areas of rísk (as categorízed by Zack, 1996) identified in this case analysis:		
Delays and disruptions The activity becomes critical and		delays the work
Defective Contract Documents	Defective Contract Documents The first file is submitted by Eng	
Coordination The Contractor does not follow up		promptly
Interpretation of requirements	Duties of EDL coordination followi	ing the $VE$ were not clear at
contract signature		
Permits and licenses The value engineering required a		revised permit file
Behavioral Observations:		
- The Engineer made a	nger process that the	
Contractor had not ac		
- The Contractor díd na	rely manner.	

CASE NO:: 20       Subject: cancellation of PS       Section:: general Requires         Synthesis:       General Observations:         The Engineer makes an assessment of the Contractor's entitlement following the cancellation of the provisional sum works. The Engineer sends the same to the Employer for his approval as per the Contract requirements since all variations assessed by the Engineer are to be approved by the Employer. The Engineer are to be approve the same which results in a case of conflict. Also, this leads to the assessment is to be made by the Engineer who is supposed to be impartial to both parties. Where the Employer does not approve of the same the Contract doesn't state how to proceed. The Engineer can not certify such variation since the Employer didn't approve it and if it is influenced by Employer's assessment then it would be in contradiction with the spirit of the FIDIC.       Areas of risk (as categorized by Zack, 1996) identified in this case analysis:         untimely responses       The Engineer was late in assessing this variation because he cot receive an approval from the Employer         Changes       The Engineer was late in assessing this variation because he cot receive an approval from the Employer         Changes       The Engineer was late in assessing this variation because he cot receive an approval from the Employer         Changes       The Engineer through his authority to give approval was delaying the assessment.		CASEANALYSIS	Project: A
Synthesis:       General Observations:         The Engineer makes an assessment of the Contractor's entitlement following the cancellation of the provisional sum works. The Engineer sends the same to the Employer for his approval as per the Contract requirements since all variations assessed by the Engineer are to be approved by the Employer. The Employer representative does not approve the same the Contract deservation since the assessment is to be made by the Engineer who is supposed to be impartial to both parties. where the Employer does not approve of the same the Contract doesn't state how to proceed. The Engineer can not certify such variation since the Employer didn't approve it and if it is influenced by Employer's assessment then it would be in contradiction with the spirit of the FIDIC.       Influence is analysis:         Areas of risk (as categorized by Zack, 1996) identified in this case analysis: Intimely responses       The Engineer was late in assessing this variation because he contrective an approval from the Employer         Changes       The Engineer was late in assessing this variation of this variation         Behavioral Observations:       The Engineer through his authority to give approval was delaying the assessment.	Case No.: 20	Subject: cancellation of PS	Section: General Regmnts.
The Engineer makes an assessment of the Contractor's entitlement following the cancellation of the provisional sum works. The Engineer sends the same to the Employer for his approval as per the Contract requirements since all variations assessed by the Engineer are to be approved by the Employer. The Employer representative does not approve the same which results in a case of conflict. Also, this leads to the <u>assessment</u> being delayed until the Employer approves the same. The assessment is to be made by the Engineer who is supposed to be impartial to both parties. Where the Employer does not approve of the same the Contract doesn't state how to proceed. The Engineer can not certify such variation since the Employer didn't approve it and if it is influenced by Employer's assessment then it would be in contradiction with the spirit of the FIDIC. Areas of risk (as categorized by Zack, 1996) identified in this case analysis: untimely responses The Engineer was late in assessing this variation because he c not receive an approval from the Employer The cancellation led to a dispute in valuation of this variation Behavioral Observations: - The Employer through his authority to give approval was delaying the assessment.	Synthesis:		General Observations:
Areas of risk (as categorized by Zack, 1996) identified in this case analysis:         untimely responses         The Engineer was late in assessing this variation because he c         changes         The cancellation led to a dispute in valuation of this variation         Behavioral Observations:         - The Employer through his authority to give approval was delaying the assessment.	The Engineer makes an assess entitlement following the cance works. The Engineer sends the approval as per the Contract req assessed by the Engineer are to The <u>Employer representative dow</u> results in a case of conflict. Als <u>being delayed</u> until the Employ assessment is to be made by the be impartial to both parties. Wh approve of the same the Contrac The Engineer can not certify su Employer didn't approve it and Employer's assessment then it the spirit of the FIDIC.	ment of the Contractor's ellation of the provisional sum same to the Employer for his quirements since all variations be approved by the Employer. es not approve the same which so, this leads to the <u>assessment</u> yer approves the same. The e Engineer who is supposed to ere the Employer does not t doesn't state how to proceed. uch variation since the if it is influenced by would be in contradiction with	influence by the employer Validity/Assessment of Variation
untimely responses       The Engineer was late in assessing this variation because he constructive an approval from the Employer         Changes       The cancellation led to a dispute in valuation of this variation         Behavioral Observations:       -         -       The Employer through his authority to give approval was delaying the assessment.	Areas of rísk (as categorízed	by Zack, 1996) ídentífied í	n thís case analysís:
<ul> <li>Behavioral Observations:</li> <li>The Employer through his authority to give approval was delaying the assessment.</li> </ul>	untimely responses Changes	The Engineer was late in assessin not receive an approval from the En The cancellation led to a dispute in	g this variation because he could mployer n valuation of this variation
assessment.	Behavioral Observations: - The Employer through	n his authority to give approve	al was delaying the
- Slow attendance to responsibilities by both parties is witnessed.	assessment. - Slow attendance to re	sponsíbílítíes by both parties	ís wítnessed.

	CASEANALYSIS	Project: B
Case No.: 1	Subject: Rejected NPVs	Section: General Requir.
Synthesis:		General Observations:
The Contractor held the Engineer liable for <u>issuing incomplete</u>		design error
<u>design</u> that necessitated these <u>c</u>	hanges resulting in cost	validity/accessment of
<u>ímplicatíons</u> . He sent NPV acco	ordingly to cover these	variation
modifications. Both parties are	making reference to	
contractual clauses to defend th	neir stand. The Engineer made	
it clear that the Contractor has	a <u>Contractual obligation to</u>	Clear rísk allocatíon and
<u>develop and prepare</u> Construction	n drawings which are valid	highlighting
for execution. Including this st	atement in the contract	
Tender Documents needed furt	per development The	
Contractor is disputing the san	he since this development of	Contract documents not clear
construction drawinas is causi	ng a running cost that	
exceeds reasonable level account	ed for by the Contractor. The	
Engineer is clear about his cont	ractual position in this	
regard. The Engineer was awar	e of the fact that the Tender	
Documents need further develop	oment but whether this was	
clarified at the Tender stage is	not clear.	
Avenc of vich (or apter ovized	hu Zoob 1000) identified in	+ this appendix structure
Areus of risk (us cucegorized	by Zuck, 1996) thence the	tional madifications as a
Changes	variation	econal monifications as a
untimely responses	The development of construction dr	rawings and their correction
<u> </u>	thereto was taking considerable tin	ne
Defective contract documents	The Tender documents needed furt	her development and although the
contract included a statement to o		evelop and prepare drawings the
extent of further design needed wa		s not clear.
design.		sponstonerg we neversponerie of
Behavioral Observations:		
- The Contractor had not accounted for the levels of design development required		
at the tendering stage and as such is trying to avoid monetary losses.		
- The Engineer having	- The Engineer having included a statement regarding the design development	
in the contract considered that this risk was clearly allocated to the contra		allocated to the Contractor
Whether the level of missing design was clearly explained/clarified to the		lained/clarified to the
Contractor at the tender stage is not known.		
- Both parties are evadi	ng responsibility by blamina	the other
Local particles are containing responsibilities by our addition of the bear		

	CASEANALYSIS	Project: E
Case No.: 2	Subject: Síte Works	Section: Division 2
Synthesis:		General Observations:
The issue disputed is related to because the <u>main branch was b</u> is due to the meaning <u>of the w</u> Engineer considers major dam the main branch, the Contracto equivalent to death of tree. <u>The</u> <u>raised to arbitration is \$8,504</u> to the contract value. The Contr regarding the tree damage and deducted is clearly not in the 'n Engineer is firm about his stan contract wording. By changin that the tree is disfigured but to the expected transplantation of that he was trying to escape an	a tree that was disfigured <u>roken</u> . The major disagreement <u>ord 'major damage'</u> . Where the age to be the case of breaking or considers main damage to be <u>value of this dispute that was</u> which is very small compared factor by reiterating his stand I the amount that will be avoiding dispute' spirit. The nd and his reading of the g his argument and he admits hat it was inevitable during the tree the Contractor proves y liability in this regard and	Human error/negligence Contract Document unclear Validity/assessment of variation
thus corresponding penalty.		
thus corresponding penalty. Areas of risk (as categorized	l by Zack, 1996) ídentífied í	n thís case analysís:
thus corresponding penalty. Areas of risk (as categorized Interpretation of requirements	by Zack, 1996) identified in The transplantation process of Contract but it is not clear whet involved cutting of certain brai	n this case analysis: the tree is requested in the ther this transplantation nches.
thus corresponding penalty. Areas of risk (as categorized Interpretation of requirements Defective Work	by Zack, 1996) identified in The transplantation process of Contract but it is not clear whet involved cutting of certain bran The Engineer is applying penal in carrying out his duties.	n this case analysis: the tree is requested in the ther this transplantation nches. ty or the Contractor's default
thus corresponding penalty. Areas of risk (as categorized Interpretation of requirements Defective Work Behavioral Observations:	by Zack, 1996) identified in The transplantation process of Contract but it is not clear when involved cutting of certain bran The Engineer is applying penal in carrying out his duties.	n thís case analysís: the tree is requested in the ther this transplantation nches. ty or the Contractor's default

	CASE ANALYSIS	Project: B
Case No.: 3	Subject: Optional Works	Section: General
		Requirements
Synthesis:		General Observations:
These optional works were included part of the contract BOQ but there was <u>no clear reference</u> to them in the Contract Conditions. The optional works that were instructed by the Engineer were <u>repriced by the Contractor as a variation on</u> the basis that these optional works are no more valid as these should have been triggered before contract signature. However, the Engineer made it clear that the tender documents were different than the contract documents signed and as such these works were intentionally left in the Contract because the Engineer is given the option of instructing them later on during the execution of the works. The Contractor stated that these optional works were among other works that were		Contract Document unclear Validity/assessment of Variation
<u>cleared and documented at con</u> invited the Contractor to list ex been removed twice. However, th cost implications resulting fro do so.	s. <u>These should have been</u> <u>tract award</u> . The Engineer amples of such work that have the Contractor who was avoiding m <u>fluctuation of rates</u> failed to	Clear allocation and highlighting of responsibility Price escalation
Areas of rísk (as categorízed	l by Zack, 1996) ídentífied í	n thís case analysís:
Defective contract documents Interpretation of requirements Cost escalation	Whether these optional works were deemed to be triggered at award on anytime during the execution of the works is not clear from the wording. Cost implications are augmented by the cost escalation. The interpretation of Contract regarding the optional works was disputed.	
Changes	hanges Contractor considered the optional work released as a variation.	
Behavioral Observations:		
<ul> <li>The Contractor was relieving himself from the obligation to carry out the works at the set rate to avoid monetary losses.</li> </ul>		
<ul> <li>The Contractor was trying to evade responsibility but was unable to prove the arguments raised</li> </ul>		but was unable to prove the
- The Engineer was firm in his interpretation of those works		

ĺ

	CASE ANALYSIS	Project: B
Case No.: 4	Subject: cleanouts	Section: Mechanical
Case No.: 4 Synthesis: The <u>additional cleanquts dispur</u> set in the specifications. The Co code is of interest to the designed does not justify the inclusion of as the Contractor is the party <u>ep</u> <u>specifications.</u> The Engineer rei Engineer's Decision that Contro of additional cleanouts.	Subject: cleanouts	Section: Mechanical General Observations: Clear risk allocation and highlighting Contractor failure to satisfy requirements validity/assessment of variation
Areas of risk (as categorized	l by Zack, 1996) ídentífied í	n thís case analysís:
Interpretation of requirements Untimely response Changes	The main Engineer and the Contro regarding the reading of the speci- The issue dragged for more than 5 The Contractor considered the requ variation.	actor are in disagreement Fications Tmonths Lest for additional cleanouts as a
Behavioral Observations:	L	
<ul> <li>The Contractor had not accounted for the level for design development required at the tendering stage.</li> <li>The Contractor was relieving himself from the obligation to carry out the additional works to avoid monetary losses.</li> <li>The Engineer was firm in his interpretation of those requirements.</li> </ul>		

	CASEANALYSIS	Project: B	
Case No.: 5	Subject: Fire Alarm System	Section: Electromechanical	
Synthesis:		General Observations:	
In reviewing the fire alarm system shop drawing the Engineer made modifications requesting additional detector. The Contractor replied to these modifications on the shop drawings stating that 'the design onus remains with the Engineer'. The Contractor seems to be concerned with the adequacy of the fire alarm system. The Engineer states it is the Contractor's responsibility to check the adequacy of design with the specified code. The Engineer stated that it is <u>part of the Contractor's responsibility of preparing</u> <u>construction drawings</u> . The Contractor even considered the additional detectors and other items added by the Engineer on the shop drawings to be a <u>variation</u> . The Engineer rejected the same since it is the <u>Contractor's responsibility to provide</u> <u>the additional items requested by code</u> . The Engineer reiterates his position in the Engineer's Decision. The process spanned for over 12 months due to delay of more than 2 months at each reply.		Clear risk allocation and highlighting Validity/Assessment of variation Contractor failure to satisfy requirements	
Areas of risk (as categorized	d by Zack, 1996) ídentífied í	n thís case analysís:	
untímely responses	Both parties were as evident in the attending to the matter	chronology of events late in	
Interpretation of requirements Obsurges Design.		esponsibility in development of itional requirements as a	
variation.			
Behavioral Observations:			
- The Contractor had n	- The Contractor had not accounted for the level of design development required		
at the tendering stag	at the tendering stage.		
- The Contractor was r	- The Contractor was relieving himself from the obligation to carry out the		
additional works to avoid monetary losses.			
- The Engineer was firm in his interpretation of those works			
- Each party was allocating responsibility to the second party.			

	CASE ANALYSIS	Project: B	
Case No.: 6	Subject: War Delay	Section: General Requirts.	
Synthesis:	<u> </u>	General Observations:	
The Contractor gave timely notice of <u>the war effect</u> delay. The Contractor disagreed with the assessment made by the Engineer based on the substantiation		Assessment of war impact	
made by the Engineer based on the substantiation submitted. The Contractor <u>considered this extension to be</u> <u>unjust</u> and considered that the Engineer 'deliberately ignored' the extension of time granted by Law. The Engineer replied that the law the contractor is referring to is not applicable to this type of contract. The Contractor further disagreed and considered that the Engineer is 'taking position with respect to the interpretation of the law'. The Engineer reiterated his position in the Engineer's Decision.		Assessment of delay	
Areas of rísk (as categorízed	d by Zack, 1996) ídentífied í	n thís case analysís:	
Force Majeur	The war that occurred in Lebanon contract	was a Force Majeur under the	
Labor forces	There was a problem in maintaini Delays and disruptions occurred f	ng the required levels of labor from war and this led to delay	
elays and disruptions suspension of work productivity dispute. Suspension of works occurred for During war period productivity is		40 days during the war. affected	
Reliance Charles Stinner	Suc sutery is jeopuruizen		
DEMANDIAL DUSERVALIONS:	LOOID OLD THE HEAD OF COLORS	of the Englisher and it is	
The contractor was in alsag	The Contractor was in disagreement with the assessment of the Engineer and it is		
clear from the use of terms s	uch us accountery ignored a	ina carring position that the	
Contractor was accusing the	congineer of being purchul.		

	CASEANALYSIS	Project: B	
Case No.: 7	Subject: soft Landscaping	Section: soft Landscaping	
Synthesis:		General Observations:	
A revised scope of planting wor	ks was agreed and the		
Contractor was requested to pri	ce the same. The Contractor		
made his own assessment and	the rate was agreed		
accordingly. Few days after th	e Engineer's acceptance of the		
variation cost which was based	on the agreement mentioned	validity/assessment of	
above between the Employer and	a the Contractor, the Contractor	Variation	
Contractor tries to explain the	sincer rejects it. Although the		
made, the Engineer considers t	he contractor lighte for the rate		
he submitted at the beginning.	At this point the Engineer		
could have reconsidered the revi	sed rate and check the same		
with the Employer but instead	reiterated his position. The		
amount of disagreed value is o	nly \$8,900. However, the		
Contractor accuses the Enginee	er to be 'Fraudulent' and the		
amount to be unlawfully dedu	cted'. The Engineer reiterates		
his position in the engineers u	ssessment.		
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied ín	n thís case analysís:	
Changes Defective contract documents	The Contractor did not allow a me	orks. chanism for assessment of	
	variations		
Interpretation of requirements	The interpretation of agreement rea	ached was dísputed.	
Behavioral Observations:			
- was it clear to the Engineer at the time the Contractor submitted the first price that there was a			
mistake in the method of calculation and did the Engineer accept the total figure in spite of			
- The value of this dispute is \$8.900 and it is clear from the dispute that took lengthy			
correspondence that both parties are not showing any willingness to compromise and resolve			
this matter.			
- Although the Contract does not clearly specify the method of valuating and although the			
Engineer accepted the Contractor's offer, the price could be reconsidered on the basis of an erro			
in calculation. The constructor used terms such as "Fraudulent" and "unlowfully deducted" converses the			
- The contractor used terms such as "Franklient" and uniawfully deducted" accusing the Engineer of being partial.			
- The Engineer refrains from such language in his replies.			

	CASEANALYSIS	Project: B
Case No.: 8	Subject: Fire Smoke Dampers	Section: Mechanical
Synthesis:		General Observations:
The Engineer is requesting the Contractor to provide duct access doors and requesting technical data sheets for the same. The Contractor insists that the system provided meets SMACNA standards. There is <u>a clear</u> <u>disagreement whether a damper located within the duct</u> requires an access door. The Engineer reiterates his position that it is the <u>Contractor's duty to meet the</u> <u>contractual obligations</u> and the Contractor has failed to do so. The Contractor on the other hand is disputed the Engineer's unjustified rejected of variation is not <u>approving the access doors as a variation</u> . Also, the Engineer's technical assistant who was approving submittals and inspecting the works on site failed to raise the problem of the missing access doors during the past 11 ¹ / ₂ months.		Contract Document unclear Clear allocation/highlighting or responsibility Contractor failure to satisfy requirements Validity/Assessment of variation
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied í	n this case analysis:
Archis of rusic (us outcogorized og Zucic, 1) pe) thereeffect to this outse unity sol.Delays and disruptionsThe ongoing disagreement regarding the system requirements are delaying the works that had been ongoing for the past 11½ months.Interpretation of requirementsThere is a clear disagreement regarding the Contract requirements regarding the Code specified in the Specs.Defective workThe Engineer is considering the work as submitted by the Contractor as defective.Untimely response ChangesThis issue drags for 6 months The Contractor submits it as a variation and the Engineer rejects the same		
Behavioral Observations:		
<ul> <li>The Contractor had n at the tendering stag</li> <li>The Contractor was n additional works to a</li> <li>The Engineer was fir</li> <li>Each party was evadi</li> </ul>	ot accounted for the level for d e. elíeving himself from the oblig void monetary losses. m in his interpretation of thos .ng responsibility by allocativ	esign development required gation to carry out the se works ng it to the second party.

	CASE ANALYSIS	Project: B		
Case No.: 9	Subject: Fire Fighting System	Section: Mechanical		
Synthesis:		General Observations:		
The Engineer requested the c	Contractor to províde detaíls			
of the hydraulic calculations	s along with comprehensive			
and detailed shop drawings.	The Contractor insisted that			
this was <u>not part of his respo</u>	nsibility to provide the	Contract Document unclear		
<u>same</u> . The Contractor also res	served his rights for the			
resulting delay. The Contrac	tor stated that he is willing			
to do the design at a fixed fe	e. The Engineer then decided	Validity/Assessment of		
to design the hydraulic calci	ulations in an attempt to	Variation		
mitigate the <u>contractor's nor</u>	icompliance with the	Contractor failure to satisfy		
contract requirements. The o	lesign responsibility was	requirements		
unciear in the contract docu	ments and was the main			
del que of alspite in this case.	Also, it is noticed that the			
Contractor The Engineer Suppre	Lied that the Contractor	Late approval of submittal by		
Contractor. The Engineer replied that the Contractor		che engineer		
function to notice og the preserve				
Areas of vish (as asteonyized by Zaph 1005) identified in this asse an plusis.				
Areas of risk (as categorized by Zack, 1996) identified in this case analysis:				
Delays and disruptions The ongoing disagreement regarding the system requirements a				
Defective contract documents	The process took more than 7 moni	the pust 11-12 months.		
Interpretation of requirements There is a clear disagreement regarding the Contract requirements				
relating to the Code specified in the Specs.				
Changes The Contractor considered the request for hydraulic calculation as				
Behavioral Observations:				
- The Contractor had not accounted for the level of design development required				
at the tendering stage.				
- The Contractor was relieving himself from the obligation to carry out the				
additional works to avoid monetary losses.				
- The Engineer was firm in his interpretation of those works				
- Each party was allocating responsibility to the second party.				

	CASE ANALYSIS	Project: B
Case No.: 10	Subject: Towel Dryer	Section: Mechanical
Synthesis:		General Observations: Late approval of submittals by
hold until the final decision	is issued in a variation	Engineer
order. An agreement is discu	issed between the Employer	Validity/Assessment of
and the Contractor which the	Contractor later on rejects	Fluctuation
sínce ít cuts down on the Cou	ntractor's overhead and	
profit. As such the agreemen	t allows for the fluctuation	
in material rate only.		
Areas of rísk (as categorízeo	l by Zack, 1996) identified in	n thís case analysís:
untimely response	The several modifications made lec The Engineer was changing the sp about the requested model.	d to a delay in the procurement. ecifications and was not clear
Behavioral Observations:		
- The Employer is interfering in an assessment of a variation		

	CASE ANALYSIS	Project: C	
Case No.: 1	Subject: Louvers at Roof	Section: Architectural Finishes	
Synthesis:	u	General Observations:	
The Contractor failed to subr	nit the requested louvers for	Contractor late/missing	
<u>a full year</u> . The Employer in	terfered to ask the Engineer	submíttal	
to instruct the Contractor on	how to proceed. The		
Engineer didn't take this in	itiative himself before being	Lack of proper management	
so instructed by the Employ	er although ít ís hís		
obligation under the Contrac	t to monitor the Contractor's		
progress and proper submitte	al of shop drawings and		
materials and as such give p	proper instruction where		
necessary. The Contractor no	ot only <u>failed to provide the</u>	Contract Document unclear	
specified louver, he also sent	a notification of possible	Lack of experience	
variation for the instructed l	ouvers. The Engineer rejects	Validity/assessment of	
the same on the basis that th	e louvers instructed are	variation	
those requested by the specifi	cations. Contractor is		
requesting a variation Orde	r to cover the additional cost		
of <u>the special pitched louvers</u> .		Contractor failure to satisfy	
		requírements	
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied í	n thís case analysís:	
Delays and disruptions	The failure to provide the requir works.	ed louvers was delaying the	
Defective contract documents	The Contract document díd not	: clearly detail those louvers.	
interpretation of	Due to the unavailability of clea	ar louver specs, the	
requírements	interpretation was disputed.		
Changes The Contractor considered the special pitched louvers a		e special pitched louvers as a	
variation			
Behavioral Observations:			
- The Contractor had not accounted for the special pitched roof.			
- The Contractor was relieving himself from the obligation to carry out the			
additional works to avoid monetary losses.			
- The Engineer was firm in his interpretation of those works			
- The Engineer having included a statement regarding the design development			
in the contract considered that this risk was clearly allocated to the Contractor.			
whether the level of missing design was clearly explained/clarified to the			

	CASEANALYSIS	Project: C
Case No.: 2	Subject: chillers	Section: Mechanical
Synthesis:	L	General Observations:
The type of the Chillers was changed and as such the Contractor sent a notification of a possible variation. The Engineer disagrees with the Contractor's reading of the specs and doesn't agree to the change between the reciprocating type and the screw type to be a variation. A further change is made to the chiller sizes. The Contractor submits the <u>new market rates of these</u> <u>chillers</u> . The Engineer goes <u>for prorated rates based on</u> the Contract rates. If the Contractor has placed an order based on the previous rates and needs to make a new order for the new sizes, then the Contractor will be incurring a loss due to this variation.		Príce escalatíon Validíty/assessment of Variation
Areas of risk (as categorized	by Zack, 1996) identified in	n thís case analysís:
Changes Defective contract documents Interpretation of requirements Cost escalation	The size of chillers is changed The Contract doesn't clearly spe Due to the fact that the type is n reached regarding the requirem The market rate has increased b	ecify these chiller types not clear a disagreement is ents. Deyond contract rates.
Behavioral Observations:		
<ul> <li>The Contractor is trying to avoid monetary losses due to prorated rates in the variation since the Contract does not allow for fluctuation.</li> <li>The Engineer uses prorated rates although this is not a clear statement in the Contract.</li> <li>There is a slow attendance to responsibilities by both parties</li> </ul>		due to prorated rates in the cuation. ot a clear statement in the h parties

Case No.: 3       Subject: Restoration       Section: Site works         Synthesis:       General Observations:         The Contractor is deemed to have satisfied himself with the site conditions. At pricing the demolition work as an experienced contractor he should have accounted for the associated risk of cracks. However, the Engineer did not explicitly include details of strengthening and maintaining the façade in the construction drawings or in the BOQ for the Contractor to account for the same and price accordingly. The Contractor gives a notification by sending a request for approval of variation       Contract documents uncle clear risk allocation pricir validity/assessment of variation         Areas of risk (as categorized by Zack, 1996) identified in this case analysis:       Catent site conditions       Cracks appeared later on the contractor is considering the strengthening as a new requirement. The Engineer is considering the strengthening as a new requirement. The Engineer is considering as a new requirement. The Engineer is considering as a new requirement. The Contractor should have accounted for the same.         Change       The Contractor had not accounted for the additional cost         The Contractor had not accounted for the obligation to carry out the this is a requirement.		CASEANALYSIS	Project: C
Synthesis:       General Observations:         The Contractor is deemed to have satisfied himself with       start observations:         the site conditions. At pricing the demolition work as an experienced contractor he should have accounted for the associated risk of oracks. However, the Engineer did not explicitly include details of strengthening and maintaining the façade in the construction drawings or in the BOQ for the Contractor to account for the same and price accordingly. The Contractor gives a notification by sending a request for approval of variation       Contract documents uncle Clear risk allocation priciv a notification by sending a request for approval of variation         Areas of risk (as categorized by Zack, 1996) identified in this case analysis:       Latent site conditions         Latent site conditions       Cracks appeared later on The Contractor should have accounted for the same.         Interpretation of requirements       Cracks appeared later on The Contractor should have accounted for the same.         Change       The Contractor considering the strengthening as a variation.         Behavioral Observations:       The Contractor had not accounted for the additional cost         The Contractor had not accounted for the additional cost       The Contractor was relieving himself from the obligation to carry out the ordivene box of is more the subscience is considering the strengthening as a variation.	Case No.: 3	Subject: Restoration	Section: Site Works
The Contractor is deemed to have satisfied himself with the site conditions. At pricing the demolition work as an experienced contractor he <u>should have accounted for the</u> <u>associated risk of cracks</u> . However, the Engineer <u>did not</u> explicitly include details of strengthening and <u>maintaining</u> the façade in the construction drawings or in the BOQ for the Contractor to account for the same and price accordingly. The Contractor gives a notification by sending a request for <u>approval of</u> variation order. The Engineer rejects the same. Interpretation of Interpretation of requirements Cracks appeared later on Interpretation of requirements Change Change The Contractor had not accounted for the additional cost The Contractor had not accounted for the additional cost The Contractor was relieving himself from the obligation to carry out the The Contractor was relieving himself from the obligation to carry out the	Synthesis:		General Observations:
Areas of risk (as categorized by Zack, 1996) identified in this case analysis:Latent site conditionsCracks appeared later onInterpretation ofThe Contractor is considering the strengthening as arequirementsnew requirement. The Engineer is considering that thChangeThe Contractor considering the strengthening as avariation.Variation.Behavioral Observations:The Contractor had not accounted for the additional cost-The Contractor was relieving himself from the obligation to carry out the	The Contractor is deemed to have satisfied himself with the site conditions. At pricing the demolition work as an experienced contractor he <u>should have accounted for the</u> <u>associated risk of cracks</u> . However, the Engineer <u>did not</u> <u>explicitly include details of strengthening and</u> <u>maintaining</u> the façade in the construction drawings or in the BOQ for the Contractor to account for the same and price accordingly. The Contractor gives a notification by sending a request for <u>approval of</u> <u>variation order</u> . The Engineer rejects the same.		Slow attendance to responsibilities Contract documents unclear Clear risk allocation pricing Validity/assessment of Variation
Latent site conditionsCracks appeared later onInterpretation ofThe Contractor is considering the strengthening as arequirementsnew requirement. The Engineer is considering that thChangeContractor should have accounted for the same.ChangeThe Contractor considering the strengthening as a variation.Behavioral Observations:The Contractor had not accounted for the additional cost-The Contractor was relieving himself from the obligation to carry out the	Areas of rísk (as categorízed	by Zack, 1996) ídentífied í	n thís case analysís:
Interpretation of requirements The Contractor is considering the strengthening as a new requirement. The Engineer is considering that th Contractor should have accounted for the same. The Contractor considering the strengthening as a variation. Behavioral Observations: - The Contractor had not accounted for the additional cost - The Contractor was relieving himself from the obligation to carry out the	Latent site conditions	Cracks appeared later on	_
<ul> <li>Behavioral Observations:</li> <li>The Contractor had not accounted for the additional cost</li> <li>The Contractor was relieving himself from the obligation to carry out the</li> </ul>	Interpretation of requirements Change	The Contractor is considering the strengthening as a new requirement. The Engineer is considering that the Contractor should have accounted for the same. The Contractor considering the strengthening as a variation.	
<ul> <li>The Contractor had not accounted for the additional cost</li> <li>The Contractor was relieving himself from the obligation to carry out the</li> </ul>	Behavioral Observations:		
- The Contractor was relieving himself from the obligation to carry out the	- The Contractor had not accounted for the additional cost		
	- The Contractor was relieving himself from the obligation to carry out the		gation to carry out the
additional works to avoid monetary losses.	additional works to avoid monetary losses.		
- Each party was allocating responsibility to the second party.	ond party.		

	CASEANALYSIS	Project: C
Case No.: 4	Subject: Glass Balustrade	Section: Facade
Synthesis:		General Observations:
The Contractor sends a requi specification of balustrade. T referred the Contractor to the that specifies the railing sys Contractor sends a request of but when rejected the Contract defense.	est of information for the The Engineer should have section of the specification stem as stainless steel. The f <u>approval of variation order</u> ctor has little argument in	validity/assessment of variation
Areas of risk (as categorized	l by Zack, 1996) ídentífied ín	n thís case analysís:
Interpretation of	The Contractor is disputing	the Contract requirements
requirements	regarding the specifications	of the railing system.
Changes	The Contractor was consideri	ng the stainless steel
	requírement as a variation.	
Behavioral Observations:		
- The Contractor had not accounted for the stainless steel requirement.		
- The Contractor was relieving himself from the obligation to carry out the additional works to avoid monetary losses.		

Case No.: 5	Subject: War Effect Delay	Section: General
		Requirements
Synthesis:		General Observations:
The Contractor submitted th	ne notice on time for the	
ongoing war effect. The Cov	itractor made hís best	
endeavors to attend the site	inspite of the risk taken in	
the absence of insurance con	/er. The time impact was	
resolved because both parties	; agreed to the delay analysis	
technique to be used. Howev	er, <u>the cost implications were</u>	Assessment of war claim
<u>disputed</u> because the Employ	jer decided to pay the	
Contractor in parts inspite a	of the extra effort that was	
made by the Contractor dur	ing the war period.	
	•	
Areas of risk (as categorize	d by Zack, 1996) ídentífied í	n thís case analysís:
Force Majeur	The war that occurred in Lebanon	was a Force Majeur under the
	contract	5
Labor forces	There was a problem in maintaining	ng the required levels of labor
	Delays and disruptions occurred f	rom war and this led to delay
Delays and disruptions	alspute.	40 days during the war
Suspension of work Productivity	During war deried droductivity is	affected
Síte Safety	Site safety is jeopardized	
Behavioral Observations:		
Had the Contractor decided	to abide by his contractual rig	ht not to attend while there
is rísk to the safety of hís ei	nployees, the Contractor would	have been automatically
entitled for more extension of time and as such the Project would have incurred more		
delay and the Employer woi	uld have paid higher premiums	. The Contractor's positive
attitude was not rewarded h	y the Employer.	,
THE CONTRACTOR TAKEN A ADD TO TAKEN A CONTRACT		

	CASEANALYSIS	Project: C
Case No.: G	Subject: External Works	Section: Site Works
Synthesis:		General Observations:
The Engineer was late in providing the Contractor with the <u>missing drawings that were necessary for him to</u> <u>carry out the works</u> . The <u>Engineer then provided</u> <u>drainage drawings</u> that were again missing and not well coordinated as they differed from the contract floor levels on the contract drawings. The Contractor stated that he has not received a reply to the <u>RFI in 55 days</u> and this was <u>delaying the works</u> .		Design error Late issue of missing design Late approval of submittals by Engineer Assessment of delay
Areas of rísk (as categorízea	l by Zack, 1996) ídentífied í	n thís case analysís:
Delays and disruptions Untimely responses Defective contract documents	The missing drainage design was delaying the works. The RFI was replied to 55 days. The Contract documents were missing drainage details.	
Behavioral Observations:		
- The Engineer was late in providing the missing details.		
- The Contractor was g	iving timely notice to raise th	e críticalíty of these details

	CASE ANALYSIS	Project: C
Case No.: 7	Subject: Stone Flooring	Section: Architectural
		Finishes
Synthesis:		General Observations:
The Engineer was late in providing the Contractor with details of the stone finishing. The type of stone was <u>specified in 2 months.</u> And then two months after that the type was changed again. ie <u>4 months later</u> . Engineer's issuance of incomplete design information, finalization of the design during the Progress of the Works and the continuous issuance of design changes was <u>causing delay</u> .		Late issue of missing design Assessment of delay Validity/assessment of variation effect
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied í	n thís case analysís:
Delays and disruptions	The missing drainage design	n was delaying the works.
untimely responses	The MAR ref. e was responde	d in 1 month.
Defective contract documents	The Contract documents wer	e míssíng.
Behavioral Observations:		
- The Engineer was late in providing the missing details.		
- The Contractor was giving timely notice to raise the criticality of these details		

ł

	CASE ANALYSIS	Project: C	
Case No.: 8	Subject: Submittal Schedule	Section: General	
		Requirements	
Synthesis:		General Observations:	
The Engineer was not attending to the Contractor's <u>submittal in a timely manner</u> as specified and agreed to in the Contract. Each of the two parties was blaming the other party for not meeting specification requirements and the schedule of submittal in the programme. The Contractor even <u>accused the Engineer of intentionally delaying the replies</u> to allow time to include 'forthcoming changes/variations' in these submittal replies. The Contractor prepared a claim of the all the delay that occurred due to the <u>delayed reply to submittals</u> .		Late approval of submittals by Engineer Assessment of delay Submittal Schedule	
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied ír	r thís case analysís:	
untimely responses	The Engineer was unable to p	neet the specification	
Delay and Disruptions Defective Contract Documents	sruptions The lat review of submittals was causing delay tract The Contract should have specified an achievable/realistic schedule.		
Behavíoral Observatíons:			
<ul> <li>The Contractor was accusing the Engineer of intentionally delaying these replies.</li> <li>The Contractor is using language accusing the Engineer of including 'forthcoming changes/variations."</li> <li>Each party was evading responsibility by blaming the other party.</li> </ul>			

	CASEANALYSIS	Project: C
Case No.: 9	Subject: Alum. Composite Panel	Section: Façade
Synthesis:		General Observations:
This issue dragged for more Contractor was <u>still requesti</u> <u>confirmation</u> to the exact det composite panel. From the tin his material approval reques changed his decision in <u>Apr</u> <u>in August 06</u> .	than a year where the <u>ng clarification</u> tails of the aluminum ne the Contractor submitted t in Dec 05, the Engineer <u>il 06, then in May 06 then</u>	Slow attendance to responsibilities Late approval of submittals Late issue of variation Assessment of delay
Areas of risk (as categorized	l by Zack, 1996) ídentífied í	n thís case analysís:
Delays and dísruptíons Untímely responses	The ongoing changes caused The engineer replied to the fi Contractor after 4 months r aluminum	d delay Irst MAR in 119 days, the equested details of
Behavioral Observations:		
<ul> <li>The Engineer was lat delay.</li> <li>The Contractor raised</li> </ul>	e in issuing modifications th I a claim regarding the missi	nat were causing further ng details requested.

	CASE ANALYSIS	Project: C	
Case No.: 10	Subject: Points of Drainage	Section: Mechanical	
Synthesis:		General Observations:	
The Engineer was rejecting r	esponsibility for the		
discrepancy of drain around	the construction joint	Desígn error	
considering it to be part of th	ne <u>Contractor's</u>	Contractor failure to satisfy	
responsibility in coordinatio	n among trades. The	requirements	
Contractor on the other hand	was again rejecting this	Clear allocation of responsibility	
responsibility <u>considering it</u>	to be the Engineer's	Contract documents unclear	
responsibility to clear such a	iscrepancies raised by the		
Contractor through RFI or ev	en to <u>nave a better</u>	Contractor poor coordination	
coordinated design arawing.	<u>s</u> .	between trades	
		Valídíty/Assessment of	
		variation	
		Assessment of delay	
- - 			
Areas of risk (as categorized	l by Zack, 1996) identified in	n thís case analysís:	
Delays and disruptions	Delay was incurred due to the c responsibility	ngoing dispute regarding	
untimely response	The matter was disputed without providing the necessary		
	developed design	_	
Changes	The Contractor considered the missing details as a variation		
Defective Contract Documents The Contract document did not		: account for well coordinated	
	points of arainage		
Behavioral Observations:			
- Each party is blaming the	otner party		
	CASE ANALYSIS	Project: D	
---------------------------------------------------------------------------	--------------------------------------	--------------------------	--
Case No.: 1	Subject: Access to Roof Area	Section: Architectural	
Synthesis:		General Observations:	
The Design is defective as it	does not <u>allow for proper</u>		
access to the roof from the 28	sth floor. The Contractor sent	design error	
an RFI in this regard. The re	ply to the RFI by the	Design Distripting	
Engineer stated that the Eng	gineer had solved the roof		
area access issue to his reply	to a shop drawing		
submittal at the 28th floor. $ au$	he corridors introduced at	Late issue of missing	
this floor created a problem w	vith the panel boards. This	design/variation	
necessitated an instruction	to relocate the panel boards.		
This was followed by <u>a modi</u>	fication to the block wall	. Validity/Assessment of	
perimeter at the 28th floor, ae	ietion of marble tuing and	Variation	
the deletion of the waterproof	culted in a discuption to the		
works	Succes in a <u>disruption to the</u>	• Assessment of delay	
Morta:			
Areas of rísk (as categorízed	l by Zack, 1996) ídentífied ír	n this case analysis:	
Delays and disruptions	The discrepancy in the roof access	was delaying these works	
Changes	The defective design necessitated t	his change	
Defective contract documents	The Contract drawings did not sho	ow any access.	
Behavioral Observations:			
- The Engineer could have provided all the information in a timely manner			
- The Contractor could have noticed this problem earlier.			

	CASE ANALYSIS	Project: D	
Case No.: 2	Subject: Raised Flooring	Section: Architectural	
Synthesis:		General Observations:	
The Engineer had <i>instructed</i>	raising the floor at the two	Late iccure of miccina	
highest floor which necessita	ted moving wet trade	design/variation	
material to those floors throu	gh elevators. However, the	Ŭ	
instruction given was not cl	ear and where the Contractor		
stated that the Hourdí blocks	s requested are not readily	Lack of proper	
available in the market, the t	Engineer refuted the same	Management/Monitoring by	
without checking the validit	y of the Contractor's	Engineer	
statement. The Contractor pr	esented an alternative for		
available Houral sizes along	with their unit rates. The	Late approval of submíttals	
Engineer was late in attendi	ng to the same although	Slow attendance to	
finishing works at four of	The Engineer was also late	responsionnes	
in assessing the variation		Validity /accessivent of	
the ussessing <u>the variation</u> .		Variation	
		Assessment of Delay	
Areas of risk (as categorized	hu Zack 1996) identified in	this case analysis	
Delaus and disruptions	Because the Hourdí specífied was n	ot readily available this led to	
	delay		
Untimely responses	The Engineer was late in providing	) feedback on the crítical activity	
Defective contract documents	The Engineer was late in his reply		
Change	The Engineer issued different char	n original design haes that led to the dispute	
5	5		
Behavioral Observations:			
- The Engineer disregarded the Contractor's comment regarding the unavailability of			
the Hourdí Blocks instead of taking a proactive approach of finding an alternative.			

	CASEANALYSIS	Project: D	
Case No.: 3	se No.: 3 Subject: Additional Fireplaces		
Synthesis:		General Observations:	
The Engineer made the decision several months after the start of the works <u>to change the false fireplaces</u> to real/live ones. 2 weeks after issuing the variation, the Engineer confirmed that Contractor should proceed with coring (350mm) to allow for proper insulation. After another week the Engineer changed the diameter and type of exhaust stacks. This also led to the <u>additional cost of surplus material</u> since the originally specified stainless steel exhausts were already specified black steel exhaust pipes. <u>The Engineer did not reply to the same in a month</u> period which delayed the finishing in the area of the fireplaces and thus caused a suspension of the works.		Late issue of missing design variation Lack of proper Management/Monitoring by Engineer?? Valuation of Variation Assessment of Delay Slow attendance to responsibilities	
Areas of risk (as categorized by Zack, 1996) identified in this case analysis:Delay and Disruption Changes Interpretation of requirementsThe variation was late It was not specified clearly The Contractor needed more details to be able to execute			
<ul> <li>Behavioral Observations:</li> <li>The Engineer issued the variation late and missing. It was further changed several times.</li> <li>The Engineer's feedback was late</li> </ul>			

	CASE ANALYSIS	Project: D	
Case No.: 4	Subject: New Kitchen	Section: Architectural	
Synthesis: The Engineer had <u>modified the kitchen design</u> and expected the variation to be assessed as remeasured works. The Contractor sent an NPV regarding the same and sent the quotation for carrying out these works received from the MEP subcontractor. By negotiating the price the <u>Employer was accepting</u> the fact that the <u>additional work was a variation</u> and that the Contract rates and prices would not apply. There was on average a one week reply period in the correspondence carried out regarding this variation which proves that there was no		General Observations: Late issue of missing design/variation Influence by the Employer Validity/assessment of Variation Assessment of delay	
negligence/delay by any of	the parties to reply on time.		
Areas of rísk (as categorízea	by Zack, 1996) ídentífied í	n thís case analysís:	
Changes	Engineer issued modified ki	tchen design	
being und disruption Interpretation of	Modified design caused dela	y to the works	
requirements	There was a disagreement regarding the method of irements assessment of this variation which was not clear from the contract		
Behavíoral Observations:			
- The Engineer rejected	an NPV that the Employer ac	cepted.	
- The Engineer was un	able to make a fair assessmen	t.	
- Engíneer firm althou	gh contract ís grey		

Subject: Aluminum Works on related to the façade in accuses the Contractor of	Section: Facade General Observations:		
<u>on related</u> to the façade in <u>accuses the Contractor</u> of	General Observations:		
on related to the façade in <u>accuses the Contractor</u> of			
	Late issue of missing design		
being late in executing the façade works. At this point the Contractor lists all modifications related to the façade that have been requested. The Contractor then			
nodífications related to the <u>n giving the approval</u> to	, ate approval of submittals by		
e works which has become s claim expressed his intent this aspect of the Works	Engineer		
and so pressed the Nominated Subcontractor for Aluminum works to give his best possible dates for			
th the requested written	Slow attendance to		
	responsibilities Lack of proper		
by Zack, 1996) ídentífied í	n thís case analysís:		
A late variation is issued wh	ich delays the work		
The Variation affects the façi The Engineer did not attend	to the matter in a timely		
manner			
Behavioral Observations:			
- The Contractor proved due diligence in proceeding with the works in delayed			
areas to minimize the effect.			
<ul> <li>Although matter is cleared in a meeting the Engineer is hesitant to give written approval.</li> </ul>			
<ul> <li>Language expressing ill perception of the other party is expressed where the Contractor accuses the Engineer of intentionally delayed approval of shop drawings to allow time for contemplated modifications.</li> </ul>			
	accuses the Contractor of cade works. At this point cations related to the ed. The Contractor then codifications related to the <u>n giving the approval</u> to e works which has become claim expressed his intent this aspect of the Works I Subcontractor for best possible dates for to resolve this issue but the th the requested written by Zack, 1996) identified in A late variation is issued wh The variation affects the faço The Engineer did not attend manner due diligence in proceeding v effect. ared in a meeting the Engin ill perception of the other part Engineer of intentionally d e for contemplated modificat		

	CASEANALYSIS	Project: D	
Case No.: 6	Subject: lighting works	Section: Electrical	
Synthesis:		General Observations:	
The Engineer issued the Employer's requirement regarding <u>the revised design for lighting works</u> . The contractor sends a notice for the variation and for the delay. The Engineer does not reply to both notifications. The Contractor sends details of the additional cost as received from the Nominated Subcontractor and requested the Engineer's approval of the same to proceed with the works. The Engineer was late in assessing the <u>value of the variation</u> and confirming his approval for the works to proceed.		<ul> <li>Late issue of revised design</li> <li>Lack of cooperation</li> <li>Slow attendance to</li> <li>responsibilities</li> <li>validity/assessment of</li> <li>Variation</li> <li>Assessment of Delay</li> </ul>	
Areas of rísk (as categorízeo	l by Zack, 1996) identified in	n thís case analysís:	
Delays and disruptions Changes The revised design for lighting works delayed the works The revised design in lighting works results in a notice of variation by the Contractor			
Behavioral Observations:			
<ul> <li>The Engineer does not reply to the Contractor's notices and is late in assessing the variations</li> <li>The Engineer is late in conforming his approval of the works</li> </ul>			

	CASEANALYSIS	Project: D
Case No.: 7	Subject: Labor Decrease	Section: General
		Requirements
Synthesis:		General Observations:
The situation in the country	following the unstable	
political situation led to <u>loss</u>	of labor and fights that	
took place on site among lab	orers. The Contractor díd	Assessment of war claim
due diligence to minimize th	ne effect of the loss of labor	
and made an effort to recruí	t Indian Labor which was	
prohibited by the Lebanese la	bor law. The <u>Contract was</u>	
not clear in this regard. The	assessment of this Force	
Majeur is disputed.		Contract Document unclear
5		
Areas of risk (as categorized	by Zack, 1996) ídentífied ír	r thís case analysís:
Force Majeur	The unstable political situation wa	is an Employer's rísk
Labor forces	The shortage of labor forces	
beings und disruptions	The coss of cabor was causing aelay	
Behavioral Observations:		
- The Contractor made	efforts to minimize the effects	of the Employer's risk
- The governmental aut	chorities did not help in resolvi	ing the labor problem.

	CASEANALYSIS	Project: D
Case No.: 8	o.: 8 Subject: EDL room	
Synthesis:		General Observations:
The Engineer requested the c	Contractor to proceed with the	
works as per an attached off	er by supplier. <u>The scope of</u>	
activity of each of the Contro	actor and the supplier in	
carrying out these works wa	<u>s not clear</u> . The Engineer	Contract Document unclear
clarified the same. However,	two activities were not	
accounted for in the definition	on of the scope and were	
considered by the Contractor	to be a variation:	► Validity/assessment of
1. The unioualing and plu the EDL room	icing of the EDC cells inside	variation
2 The cost of required m	edium voltage between EDI	
room and the transfor	mer room.	
The Engineer was late in app	proving the additional scope.	
5	5	
Areas of risk (as categorized	by Zack, 1996) identified iv	r this case analysis:
Delays and disruptions	The Engineer's delay in reply was	delaying procurement
Coordination	Coordination is required between the	ne three parties to meet the full
	scope of works	
Behavioral Obcenvations.		

	CASE ANALYSIS	Project: D
Case No.: 9	Subject: Fire Rated Doors	Section: Architectural
Synthesis:		General Observations:
The contract design was missing the fire resistant system in basement 4, the Engineer provides details of the same. The Contractor had sent the <u>subcontractor's</u> offer and was waiting for the Engineer's <u>approval</u> to proceed. Meanwhile the Contractor stated that he was putting this area on hold. <u>79 days after the issuing of</u> the drawings the Engineer states that these items were <u>no more required</u> and requested the Contractor to proceed with the Contract design. The issued drawings included other areas in basement 3 which upon the Contractor's questioning, the Engineer confirmed that fire rated items in basement 3 should proceed as per design. This process delayed the progress of the works.		Slow attendance to responsibilities Late issue of missing design/variation Validity/assessment of Variation Assessment of Delay
Areas of risk (as categorized	by Zack, 1996) ídentífied í	n thís case analysís:
Defective contract documents Changes	The original design was missing the fire resistance system. The Engineer provides missing design which he deletes from the scope later.	
Behavioral Observations:		
<ul> <li>The Engineer was hes</li> <li>The Contractor stated react promptly.</li> </ul>	ítant ín hís requírements that the area was put on hold	, but the Engineer did not

	CASEANALYSIS	Project: D	
Case No.: 10	Subject: Late Issue of Drawings	Section: Architectural	
Synthesis:		General Observations:	
The Contractor requested <u>the finishing details between</u> <u>the ceiling and the wall cladding</u> which was preventing him from proceeding with critical activities. This led to putting the gypsum ceiling on hold. The Engineer replied to the Contractor's request stating that the required details had been passed directly to the ID subcontractor. The Contractor resented that this has occurred several times on the project. As such the Engineer had not respected the <u>proper communication</u> <u>channels</u> . Moreover, the referred to details were still missing. The Engineer provided the required details <u>34</u> <u>days after the Contractor initially requested them. This</u> delayed the progress of the works.		Contract Document unclear Improper communication channel Slow attendance to responsibilities Assessment of Delay	
Areas of rísk (as categorízeo	l by Zack, 1996) ídentífied ír	this case analysis:	
Delays and disruptionsThe Contractor was late in providing the detailsDefective contract documentsThe Contract documents were missing the detailsUntimely responseThe Engineer was late in providing the details			
Behavioral Observations:			
<ul> <li>The Engineer was unable to meet his obligation and provide the details on time.</li> <li>The Engineer was not respecting the proper communication channel.</li> </ul>			

l

### Appendix I: Expert Comments to Recommendation

#### Section 1: Conditions of Contract

The amendments to the Conditions of Contract in the format suggested is likely to result in a reduction of disputes: Expert A: Yes – Quite Significant

Expert B: Yes – Quite Significant Expert C: No Expert D: Yes Expert E: Yes – Very Significant

### Have you experienced disputes on projects related to one or more of the clauses addressed in this section? If yes please specify?

**Expert A**: These issues are typical heads of claim in most construction contracts and I have experienced many of them. The introduction of the proposed amendments and mechanisms will have an impact of the resolution or even avoidance of claims. However, the proposals will only work if both parties to the contract buy into this philosophy and act professionally throughout. This is from inception to completion. The Employers team must recommend the correct procurement route to provide for greater certainty of risk. The Employer should empower the Engineer to act impartially where necessary. The Contractor must work with the Employer to achieve their joint goal of completing the project. The Employer must recognize the Contractors aim to make money and the Contractor must recognize the Employers aim to achieve value for money. Such a mature approach will help in promoting the perceived success of a project.

**Expert B**: Yes. Extensions of Time, Valuation of Variations, Price Adjustment, on many occasions.

**Expert D**: I have experienced disputes related to all of the clauses during my tie of working as a Commercial Manager, Contract Administrator/ claims Consultant & Project Manager on a number of major projects in the Middle East including Bahrain International Airport, Regency Intercontinental Hotel Bahrain, Equestrian Club Riyadh, Qatar Cultural Village, Al Ain University and Churchill Towers Dubai.

1.1. Further detailing of clauses related to time assessment, cost assessment and Employer's risk are prone to reduce disputes on projects:
Expert A: Yes
Expert B: Yes
Expert C: None
Expert D: Yes

Expert E: Yes

### 1.1.1. An Extension of Time clause in the format described would assist in reducing disputes:

**Expert A**: Agree- Again, the success of such amendments will require a proper understanding of risk apportionment prior to the appointment of any contractor. This may involve the re-education of an experienced Employer who may expect certain rights. The Contractor must understand the resource required for proper implementation of this Clause.

Certain time limits to the mechanism of such a clause would be required.

**Expert B**: Agree - Provided the Employer accepts the clause as detailed at pre-contract negotiation stage, and provided the Engineer is genuinely acting not only impartially, but independently, without any undue influence from the Client, then some of these provisions would certainly serve to reduce the scope for dispute in the event of elongation. Particularly, if the level of detail for substantiation was set, this may obviate the tactic of continual referral for 'further and better particulars' sometimes employed by an Engineer under encouragement from a Client.

**Expert C**: Disagree - More clauses mean more obligations, responsibilities and cost and therefore more disputes.

**Expert D:** Strongly disagree - I consider that introducing detailed prescriptive requirements into Clause 14 would only increase the risk of further disputes over the Programme as the parties may engage 'experts' to support their positions.

I have experienced, on a significant number of projects, disputes arising at the initial stages of a contract over acceptance/ approval of the Master Programme which has resulted in contracts starting poorly and distorting relationships. I consider that, in simplistic terms, there are lies, statistics and PROGRAMMES' and the introduction of the proposed amendments would encourage disputes over extremely detailed procedures rather than focus on the project team working together.

Expert E: Agree

### **1.1.2.** A Valuation of Variation clause in the format described would assist in reducing disputes:

**Expert A**: Agree- *If greater certainty and clarity can be introduced at any point in a Contract, especially regarding cost, then as long as both parties fully understand the risk apportionment, this should be a positive step. However, it should be remembered that any contract requires balance and the drafter must consider a fair apportionment of risk* **Expert B**: Agree - The Valuation of Variations under the UK JCT 1998 Standard Form of Building Contract is set out under clause 13.5, and prioritizes the methods of valuation in strict order of preference. The general underlying premise is that "no Variation shall vitiate a Contract" and the RICS Definition of Prime Cost of Daywork is utilized in the case of a dissimilar character to those in the Priced Document. Any substitution of the words "appropriate or 'reasonable' for something with pre-defined parameters from FIDIC 1999, would remove any room for interpretation and certainly

assist in reducing disputes.

#### Expert C: Agree

**Expert D**: Agree - I consider that a clearly worded and detailed VoV Clause should provide transparency in establishing and agreeing rates between the parties and assist in reducing disputes.

I would be concerned over the amendment proposed to 1999 Red Book Clause 12.3 'should the measured quantity of the item be changed by more than 10%' – as this would change and increase the contractual significance of the quantities of items in the BoQ's at a time when various parties wish to minimize the detailed measurement of numerous items which have little cost significance.

**Expert E**: Strongly Agree - Whenever the valuation of variation clause is detailed for each item of work, and if a related change in quantity is by more than x %, an assessment is done based on market rates for additional quantities and work executed if a dispute arise the parties are inclined to resolve it.

### 1.1.3. Adding details for Contractor's entitlement of compensation for Employer's Risk as proposed would assist in reducing disputes:

Expert A: Agree- As noted under section 1.1.2 above.

**Expert B**: Agree - Employer's Risk in areas or times of Political uncertainty or instability in this Lebanese example would firstly have to be defined or classified in terms of whether Acts of War, Civil Commotion, Riot or other such Civil unrest would constitute Force Majeure or not.

Provided that at tender stage there was clear prior knowledge of where the risk of such events might lie, in order that it may be insured against, or even better, accurately priced in advance as a Contingency / Vacation of Contract (and area) plan, then this area of potential dispute could be virtually eradicated.

#### Expert C: Agree

**Expert D**: Strongly Agree - I consider that a clearly worded Clause detailing the Contractor's entitlement to compensation for Employer's Risk would significantly assist in reducing disputes.

My view has developed from my experience of working on major projects in Bahrain during the Iraq invasion of Kuwait when the uncertainty arising over the Contractor's entitlement to payment resulted in a major contractual dispute which impacted negatively on the project when it recommenced.

**Expert E**: Strongly Agree - Adding details for contractors' entitlement of compensation, as a consequence the Lebanese 2006 July War a lot of disputes arose, a specified list of the cost for the contractors' entitlement of compensation is very useful.

#### What further clarifications to these clauses would you propose?

**Expert A:** As stated above, greater certainty under any contract should be considered a good thing by either party. If both parties can agree detailed rules to address, measure and value most typical eventualities under a construction contract, this can only be a positive step.

**Expert B:** Definitions of exactly what Employer's Risks and relevant events would trigger this Contractor's Entitlement' clause. The expansion of sphere of any works to areas such as Afghanistan, Iraq and Kurdistan etc. would no doubt merit some specialist advice from experience in these areas.

**Expert C:** All the 'damage' is done by the time the contracts are set up. The employer wants a building and the contractor wants money. The procurement process (especially in the Middle East) is set up so that the cheapest tender wins – the contractor that submits the lowest price wins. How the contractor gets to be the lowest is by cutting corners, profit, overheads, the potential effect of risks, etc. When the project is delayed, it costs the contractor money and he will try to recover his losses through whatever means because his minimal margins are being or have been eroded away.

The amended forms of contract used in the Middle East are so heavily in favour of the Employer that the Contractor is carrying an unreasonable and dispropionate amount of risk (in my view) and the contractor that wins the tender will invariably be the one that either is incompetent in assessing the risks or, is hoping all goes well regardless and that he can just muddle through.

Adding in the additional clauses as suggested will increase the burden on the contractor and in a competitive tender, corners will be cut to reduce prices. I would therefore suggest less clauses, not more with possibly an independent body (paid for by the Employer – on the basis that having disputes resolved quickly will save him money and get his building quicker) carrying out the duties you describe in the additional clauses. Otherwise, incorporating a 'Lebanese version' of the SCL Protocol into the contract could work for projects over a certain value.

Expert D - Proposed Clause is reasonable.

1.2. Care is advised to be taken in amending particular conditions listed in section,

this should allow for minimizing disputes:

Expert A: Yes Expert B: Yes Expert C: Yes Expert D: Yes Expert E: Yes

1.2.1. An 'Employer's Prior Approval' clause in the format proposed would assist in reducing disputes:

**Expert A**: Agree - It is the intention of the 1987 FIDIC Contract that the Engineer should act impartially in administering the Contract. However, under the 1999 form, he generally acts on behalf of the Employer. However under Clause 3.5 he is required to provide "fair determination". The Employers right to dispute the Engineers decision will always provide for a level of uncertainty. This level of uncertainty may be reduced but not removed by the Employer's right to agree variations in principle. The proposed mechanism of the variation budget similarly provides the Employer with the right of approval and thus would not provide the Contractor with any greater cost certainty until such approval is received. However, I do recognize the importance of "agreement in principle".

**Expert B**: Neither agree nor disagree- An interesting option employed in In the UK under certain tripartite financing arrangements, which can be incorporated in certain forms of Contract, is provision for a "Bank Monitoring Surveyor" typically in Development Projects, usually an RICS Chartered Quantity Surveyor directly employed by, and acting specifically to protect the interests of the Funding Bank. The Engineer or Architect will issue a Payment Certificate to the Contractor, Employer / Developer and Funder, but this will not be paid by the Bank until the "Monitoring Surveyor" approves it. He will attend all Site Meetings, and get involved with the Contractor / PM and Engineer in order to attempt to significantly reduce any risk to the Contract and the Bank of disputes between the Parties. He has power to overturn decisions of the Engineer or Architect.

Expert C: Disagree - Delays in approval from the Employer would cause disputes.Expert D: Agree - I agree in principle that such a Clause, if implemented in accordance with its intent, may reduce disputes.

I consider however that the proposal to set a budget limit for variations which the Engineer may certify would be of limited practical value as the Engineer should be implementing the procedures proposed as part of his standard 'good practice' reporting of costs to the Employer.

Expert E: Agree

### 1.2.2. Recommendation to replace the 'Engineer's Decision' in the format described would assist in reducing disputes:

**Expert A**: Strongly Agree - The use of any form of alternative dispute resolution (ADR) should be encouraged. DAB's provide such a form of ADR and are considerably cheaper and faster than arbitration or litigation. The introduction of Adjudication in the UK has been successful and is now the principle means of dispute resolution in the Construction Industry.

Expert B: Disagree - Engineers decisions are always likely to be the same as the Engineers
Determinations, especially if there is Client influence behind them. Mandatory use of the
Dispute Arbitration Board or 'DAB' referred should ideally be incorporated into balanced
Contracts, the underlying purpose is to provide a vehicle for dispute resolution whilst the work
proceeds, much as the Adjudication Procedure which became Mandatory in the UK under the
1996 Construction Act. Anything which avoids protracted Litigation is to be recommended.
Expert C: Agree - The Employer is normally not a technical person/body and may fail to
understand the significance of what he is doing and why. Delays would cost the
Employer money and give the Contractor reason to delay further – hence more disputes.
Expert D: Strongly Agree - I have never had the experience of an Engineer reversing a
previous decision under this Clause. I therefore would support the use of a DAB.
Expert E: Agree - Especially in the Middle East, I witnessed the cancellation of the DAB
clause, the number of disputes in these projects increased, the impact of the cancellation of the DAB
clause has been aggravated after the financial crises where many construction

## 1.2.3. Recommendation regarding 'Amendments for Lump Sum Contracts' as described would assist in reducing disputes:

**Expert A**: Agree - It must be noted that the choice of the correct procurement route and contractual mechanism is a major factor in the success or otherwise of a project. I agree that the form of Contract should be correct for the works in hand, however, it may be that the Lump Sum form is suitable where deign is incomplete with correct use of provisional sums.

**Expert B**: Strongly Agree - Care in selection of the correct Form of Contract suitable for the procurement of the proposed form and nature of Work is always highly recommended. **Expert C**: Agree

**Expert D**: Strongly disagree - I consider that Lump Sum Contracts have been and may continue to be used successfully for the delivery of large complex projects provided that competent consultants are appointed and the time is made available to define the brief and produce a full set of documents on which the lump sum price can be accurately established. There are also adequate provisions under a lump sum contract to adjust the lump sum price to take into consideration the various changes in conditions that may be encountered during the execution of the works.

For an Employer, one of whose critical criteria is 'certainty of price', the use of a lump sum contract should offer him confidence that such certainty is achievable.

I consider that a Design-Build Contract for a large complex project may increase the risk to the Employer of him not obtaining the quality of project he expects should the Employer's Requirements not be fully developed or implemented by a D&B Contractor who may be more focused on buildability & delivery of the project.

**Expert E** - A lot of disputes arise in Lump Sum Contact and notably claims for delays and disruptions due to error in design, a literature on patent and latent error in design is submitted to evidence.

## 1.2.4 A 'Price Escalation' clause in the format described would assist in reducing disputes:

**Expert A**: Agree - Price adjustment formula in a fluid or volatile market place can be seen as vital to the success of any project as if a Contractor is loosing money due to market conditions beyond their control, then this will encourage claims – not necessarily correctly. However, the Employer must fully understand how this risk is apportioned under the contract.

**Expert B**: Disagree- I assume the Theorie d'imprevision contained in the Civil Code is an underlying principle of balancing or redressing the unfair allocation of risk. In the case of escalation / compensation / market fluctuations, the British Cost Information Service and Royal Institution of Chartered Surveyors are years ahead of foreign markets in the fair and transparent adjustment of fluctuations via NEDO (National Economic Development Office) published formulae. These simple weightings, published on a monthly basis in times of base price turbulence, were universally accepted by Client / QS / Contractor alike, and should be introduced into FIDIC / international markets to reduce such areas of dispute.

Expert C: Agree

**Expert D:** Disagree - I fully agree with the use of a price adjustment clause on projects of a medium to long term duration, particularly during periods of high volatility in costs/ prices.

I can certainly appreciate the apparent advantages of using a formula method to calculate fluctuations with its advantages of ease and speed of use with minimal expenditure of resources.

However I do not have confidence in the use of general formula for the calculation of such fluctuations as they may produce anomalies which do not reflect actual market prices and may therefore give rise to disputes.

I consider that implementation of Clause 70.1, fully detailed and resourced and verified may produce a result which will not result in disputes and will be acceptable to both parties. Such a system may be implemented cost effectively through utilization of appropriate cost data bases etc.

**Expert E:** Strongly Agree - Especially the increase in the rates of materials witnessed last year, the Arab countries witnessed it in the increased number of disputes in this relation.

Expert A*	Expert B	Expert C	Expert D	Expert E
Amendments for	Correct	Price	Engineer's	Price
Lump Sum	Definition and	Fluctuation/	Decision	Fluctuation/
Contracts	application of	Adjustment		Adjustment
	Lump Sum			
	Contracts			
Employer's	Engineer's	Engineer's	Employer's	Employer's
approval to	Decision	Decision	approval to	approval to
Engineer's			Engineer's	Engineer's
determination			determination	determination
Engineer's	Employer's	Amendments for	Amendments	Engineer's
Decision	approval to	Lump Sum	for Lump Sum	Decision
	Engineer's	Contracts	Contracts	
	determination			
Price Fluctuation/	Price Fluctuation/	Employer's	Price	Amendments
Adjustment	Adjustment	approval to	Fluctuation/	for Lump
		Engineer's	Adjustment	Sum
		determination		Contracts

#### Please rank them in the order of their impact on dispute minimization

* The ranking is, in some way, subjective and may be different depending on the individual circumstances surrounding a project, including economic and political climate

#### Kindly express any views/comments/additions you may have regarding this section

**Expert A**: As previously stated a Contract must be balanced in terms of risk apportionment. Any move to provide greater certainty for both parties must be welcomed but may require some education and must be fair.

Expert B: As detailed above

**Expert C:** Good idea – provided that the <u>independent</u> FIDIC Engineer still gets the final word.

**Expert D:** I consider that amendments to standard forms of contract should be discouraged as the amending party is frequently motivated to obtain a contractual or commercial advantage which may be unfair or unreasonable to the other party. Such amendments may give rise to suspicion or the feeling of unfairness from the start of a contract which may produce an atmosphere in which disputes may be encouraged.

#### Section 2: Procurement Practices

The inclusion of Procurement Practices in the format suggested is likely to result in a reduction of disputes:

Expert A: Yes - Very Significant

Expert B: Yes – Of reasonable significance

Expert C: No

Expert D: Yes - Quite significant

Expert E: Yes - Greatly significant

2.1. A pre-bid meeting in the format described would assist in reducing disputes:

Expert A: Strongly Agree Expert B: Neither agree or disagree Expert C: Disagree Expert D: Agree

Expert E: Strongly Agree

#### 2.2. A pre-award conference would assist in reducing disputes:

Expert A: Strongly Agree Expert B: Agree Expert C: Disagree Expert D: Agree Expert E: Strongly Agree

## Have you participated in pre-bid meetings/pre-award conferences similar to those described above:

**Expert A**: Yes. The use of these processes can greatly reduce the incidences of claims through misinterpretation of the tender and Employers requirements. The meeting should have a detailed agenda and be properly recorded to be effective.

**Expert B**: I have never attended a 'pre-bid' meeting in the manner described, but have met many potential Clients in the UK in order to assess mutual suitability at a pre- tender interview. Once qualified, and once a successful competitive bid is are submitted, we normally schedule a post-tender or pre-contract meeting to review such matters. **Expert C**: No

**Expert D:** I have been involved in a number of public sector projects which have been conducted on a 2 Stage Competitive Tender basis where there have been meetings/ briefings conducted with all tenderers at 'time of issue' and 'mid-tender'.

I consider that these meetings/ briefings were successful in providing tenderers with a fuller understanding of the project, clarified ambiguities and tenderers queries and resulted in more complaint & comprehensive tenders being submitted.

## Would you propose/add any other point to be discussed during those pre-bid and pre-award meetings?

**Expert A**: The combined result of both meetings should be that the Employer is made fully aware of what he is buying for the proposed Contract price and the Contractor is fully aware of what the Employer is expecting to get for this price.

**Expert B**: The Term 'Best Practice' is the subject matter of this issue, many volumes an guides about which have been written, and Industry Professionals in the UK normally follow such Practices.

## 2.3. The MOU of higher management proposed would help maintain a cooperative spirit

Expert A: Agree

Expert B: Strongly Agree Expert C: Disagree Expert D: Disagree Expert E: Strongly Agree

#### Kindly express any views/comments/additions you may have regarding this section

Expert A: These practices should also be used by the Contractor with their suppliers. Expert B: Best Practice Protocol is always recommended, many standard procedure manuals are available.

**Expert D:** I consider that procurement procedures that provide tenderers with the opportunity to engage more fully in the tendering process are most beneficial to the success of a project in that they encourage the tenderers to devote the resources required to understanding more fully the project for which they are tendering. I consider that an MOU between the parties is a positive step and may be reinforced through the introduction of a clause, such as that included in the NEC suite of contracts which states that the parties shall act in a spirit of mutual trust and co-operation.

#### Section 3: Regulating the Industry

### The implementation of regulation to the industry in the format suggested is likely to result in a reduction of disputes:

**Expert A**: Yes - Quite Significant - Depending on how the industry is regulated it may result is quicker settlement of those claims that do arise.

Expert B: Yes - Quite Significant

Expert C: Yes - Very Significant

Expert D: Yes - Quite Significant

Expert E: Yes - Very Significant

#### The Government should contribute in cultivating a healthy construction industry:

Expert A: Strongly Agree Expert B: Agree Expert C: Agree Expert D: Strongly Agree Expert E: Strongly Agree

# **3.6.1. Enforcing regulations/restrictions to Drafting of Construction Contracts would reduce disputes:**

Expert A: Agree Expert B: Agree Expert C: Agree Expert D: Agree Expert E: Agree

### 3.6.2. Enforcing regulations to Procurement Practices would reduce disputes:

- Expert A: Strongly Agree
- Expert B: Agree
- Expert C: Agree
- Expert D: Strongly Agree
- Expert E: Agree

### 3.6.3. Enforcing regulations to Professional Skill Requirement would reduce

disputes:

- Expert A: Agree
- Expert B: Agree
- Expert C: Agree
- Expert D: Agree
- Expert E: Strongly Agree

### 3.6.4. Regulations to safety code should be better enforced:

Expert A: Strongly Agree Expert B: Agree Expert C: Agree Expert D: Strongly Agree Expert E: Strongly Agree

### Kindly express any views/comments/additions regarding your agreement/disagreement to the points raised above

**Expert A**: Regulation of the industry in the UK has a mostly positive effect particularly in terms of dispute resolution and health and safety and should be encouraged. Contractors and Employers should also have joint representation to encourage discussion between both parties and the regulators.

Expert B: Most of the statements and assertions made are fairly obvious – Best Practice implemented by experienced and Qualified Industry Professionals, imposing balanced and reasonable steering criteria upon a developing market will always be preferable.
Expert E: regulations to safety code is a recommended to be enforced especially in the Arab countries where there are not adequate laws or no laws at all

#### <u>General</u>

Which of the three sections/recommendations above you rate as the most significant in achieving the goal of minimizing disputes? Why? Which of the three sections/recommendations above you consider as the least significant in achieving the goal of minimizing disputes? Why?

**Expert A:** I believe that all three sections are important and combined correctly will reflect the increasing maturity of the Construction Industry.

As I stated under the first section, all the proposals within the report abstract must be accepted by both parties to a construction contract in order for them to be effective. An industry with greater certainty in terms of cost, risk apportionment, fairness and safety should be welcomed by all.

**Expert B:** The Construction Process is a holistic entity, and the isolation or ranking of one individual area above the other in this context is probably not conducive, as they are not mutually exclusive.

The section on Conditions of Contract is most relevant in Dispute Resolution, and perhaps the other two, Procurement Practices and Regulation could both be construed as an equally important 'follow on'' from the basis of Contract.

**Expert C:** Most significant is Section 3. Fixing something before it breaks completely is always preferable to waiting until everything collapses around you and then doing something!!

Least Significant is Section 2. It is too late to make meaningful changes by this stage. **Expert D**: Most significant is Section 2 –Procurement Practices - because implementation of 'best practice' in the procurement of a contractor should result in him fully understanding and allowing for the requirements of the contract, including risks, in his contract price. The Contractor will therefore not be under pressure to attempt to maximize the opportunities of recovering costs not fully identified and allowed for and which are frequently the sources of disputes.

The Contractor's senior management should also have made a formal commitment to deliver the project in a cooperative and non-adversarial approach.

Least significant is Section 1 – Regulating the Industry – It would be extremely difficult to minimize disputes within the construction industry by regulation (except possibly in a command economy) due to this being a long term process requiring a significant change in culture; while there are so many parties involved in each project most of whom have short term goals – many of which are conflicting. Also from my experience it is frequently the case that regulatory authorities develop into self serving bureaucracies who have little interest in developing solutions but become part of the problem through their strict adherence to the process or overly bureaucratic procedures.

Expert E: Most significant are Sections 1, 2, 3